The History and Theory of Rhetoric

By tracing the traditional progression of rhetoric from the Greek Sophists to contemporary theorists, this textbook gives students a conceptual framework for evaluating and practicing persuasive writing and speaking in a wide range of settings and in both written and visual media.

The book's expansive historical purview illustrates how persuasive public discourse performs essential social functions and shapes our daily worlds, drawing on the ideas of some of history's greatest thinkers and theorists. The seventh edition includes greater attention to non-Western rhetorics, feminist rhetorics, the rhetoric of science, and European and American critical theory. Known for its clear writing style and contemporary examples throughout, *The History and Theory of Rhetoric* emphasizes the relevance of rhetoric to today's students.

This revised edition serves as a core textbook for rhetoric courses in both English and communication programs covering both the historical tradition of rhetoric and contemporary rhetoric studies.

This edition includes an instructor's manual and practice quizzes for students at www.routledge.com/cw/herrick.

James A. Herrick is Emeritus Professor of Communication at Hope College, USA. His publications include *Argumentation: Understanding and Shaping Arguments*, 6th edition (2017) and Visions of Technological Transcendence: Human Enhancement and the Rhetoric of the Future (2017).

Praise for the Previous Edition

"The History and Theory of Rhetoric balances illuminating and concise reviews of canonical thinkers with critical discussions of the rhetorical tradition's blind spots, internal conflicts, and periodic course corrections. Its accessible prose and lively pace help students readily apprehend potentially intimidating material and recognize how the insights of the past reverberate through contemporary culture."

-Brett Ingram, Boston College

"Offering a foundation that encompasses over 2000 years, this text is digestible and meticulous rather than reductive. The author synthesizes history and theory such that readers can easily reference one part of the book while reading another and can utilize its term lists long after the first read. It is both a critical sourcebook and required reading for anyone interested in how rhetoric evolved."

- Zosha Stuckey, Towson University

"In my opinion, *The History and Theory of Rhetoric* is the best rhetorical history textbook on the market. Students at all levels appreciate its engaging, straightforward approach. Even so, serious philosophical and theoretical questions that arise from rhetoric's history receive the attention and care they deserve. The book strikes a perfect balance between accessibility and rigor."

-Jackie Wheeler, Arizona State University

"James Herrick invites students 'further up and further in' to the contours and depths of the art of rhetoric. He offers serious engagements with historical texts that ground the art and wades into the currents of contemporary theories that simultaneously interrogate and rely upon our rhetorical heritage."

-Calvin L. Troup, Geneva College

"The History and Theory of Rhetoric is not simply a book but a fascinating journey to explore the magnificent world of mind and speech interconnection throughout centuries, and its influence on various social developments of humanity. It is diverse and inclusive for all times and civilizations. The book is easy to understand and never simple. It provides valuable knowledge, useful skills, and competitive tools. The book is a must for all those who want to explore the world of rhetoric!"

-Anahit Parzyan, Nanjing University

7th Edition

The History and Theory of Rhetoric

An Introduction

James A. Herrick



Seventh edition published 2021 by Routledge

52 Vanderbilt Avenue, New York, NY 10017

and by Routledge

2 Park Square, Milton Park, Abingdon, Oxon, OX14 4RN

Routledge is an imprint of the Taylor & Francis Group, an informa business

© 2021 Taylor & Francis

The right of James A. Herrick to be identified as author of this work has been asserted by him in accordance with sections 77 and 78 of the Copyright, Designs and Patents Act 1988.

All rights reserved. No part of this book may be reprinted or reproduced or utilised in any form or by any electronic, mechanical, or other means, now known or hereafter invented, including photocopying and recording, or in any information storage or retrieval system, without permission in writing from the publishers.

Trademark notice: Product or corporate names may be trademarks or registered trademarks, and are used only for identification and explanation without intent to infringe.

First edition published by Pearson Education, Inc. 2005 Sixth edition published by Routledge 2017

Library of Congress Cataloging-in-Publication Data

Names: Herrick, James A, author.

Title: The history and theory of rhetoric : an introduction / James A Herrick.

Description: 7th edition. | New York, NY: Routledge, 2021. |

Includes bibliographical references and index.

Identifiers: LCCN 2020031218 (print) | LCCN 2020031219 (ebook) |

ISBN 9780367429294 (hardback) | ISBN 9780367427344 (paperback) |

ISBN 9781003000198 (ebook)

Subjects: LCSH: Rhetoric-History. | Rhetoric.

Classification: LCC PN183 .H47 2021 (print) | LCC PN183 (ebook) |

DDC 808.009-dc23

LC record available at https://lccn.loc.gov/2020031218

LC ebook record available at https://lccn.loc.gov/2020031219

ISBN: 978-0-367-42929-4 (hbk) ISBN: 978-0-367-42734-4 (pbk) ISBN: 978-1-003-00019-8 (ebk)

Typeset in Times New Roman and Futura by River Editorial Ltd, Devon, UK

Visit the companion website: www.routledge.com/cw/herrick

Contents

Preface

1 An Overview of Rhetoric

Re-evaluating Rhetoric 3 Rhetoric and Persuasion 4

	Defining Rhetoric 6 Rhetoric and Symbol Systems 6 Effective Symbolic Expression 8	
	Rhetorical Discourse 9 Rhetoric Is Planned 9 Rhetoric Is Adapted to an Audience 10 Rhetoric Reveals Human Motives 11 Rhetoric Is Responsive 12 Rhetoric Seeks Persuasion 12 Rhetoric Addresses Contingent Issues 15	
	Social Functions of the Art of Rhetoric 16 Rhetoric Tests Ideas 17 Rhetoric Assists Advocacy 17 Rhetoric Distributes Power 18 Rhetoric Discovers Facts 20 Rhetoric Shapes Knowledge 20 Rhetoric Builds Community 21	
	Conclusion 22 Recurrent Themes 22	
	Questions for Review 24	
	Questions for Discussion 24	
	Terms 28	
2	The Origins and Early History of Rhetoric The Rise of Rhetoric 33 Athenian Democratic Reforms 33	32

xii

1

	vi		
Contents		Education in Athens 35 Courts and Assemblies in Athens 37	
		The Sophists 38 The Flourishing of Athens 39 The Sophists' Reputation 39 What the Sophists Taught 40 How the Sophists Taught 41 Why the Sophists Were Controversial 42	
		Two Influential Sophists 45 Gorgias 45 Gorgias' Encomium and Rhetorical Devices 46 Protagoras 48	
		Isocrates: A Master of Rhetoric 49 Isocrates the Teacher 50 Isocrates the Political Theorist 51	
		Women Writers of Ancient Greece 52 Sayings of the Spartan Women 52 Aspasia 53 Sappho 54	
		Conclusion 55	
		Questions for Review 56	
		Questions for Discussion 56	
		Terms 57	
	3	3 Plato versus the Sophists: Rhetoric on Trial	62
		Plato's Gorgias: Rhetoric on Trial 63 The Debate with Gorgias: Rhetoric's Nature and Uses 64 Socrates versus Polus: Rhetoric as Power 66 Socrates versus Callicles: The Strong Survive 69 The Outcome of the Gorgias 70 Is Plato Fair to Rhetoric and the Sophists? 70	
		Rhetoric in Plato's Phaedrus: A True Art? 72 The Complexity of the Soul 73 The Myth of the Charioteer 73 Components of a Techne of Rhetoric 74 Rhetoric as Soulcraft 75	
		Conclusion 76	
		Questions for Review 77	
		Questions for Discussion 77	
		Terms 78	
	4	4 Aristotle on Rhetoric	81

Aristotle on Rhetoric

Defining Rhetoric 82

Rhetoric and Dialectic 82

Dialectic and Rhetoric: Similarities and Differences 83

Similarities 83

Contents

Differences 83
Rhetoric as a Techne of Discovery 84

The Enthymeme 86

An Argument Constructed with the Audience 86 A Democratic Argument 87

Three Rhetorical Settings 87

Deliberative Oratory 88 Epideictic Oratory 89 Forensic Oratory 90

The Artistic Proofs 90

Logos: The Logic of Sound Arguments 91 Pathos: The Psychology of Emotion 91 Ethos: The Sociology of Good Character 92

Topoi: Lines of Argument 93

Special Topics 93 Common Topics 94 Some Common Fallacies 95

Aristotle on Style 95

Metaphors and Other Devices 96

Conclusion 96

Questions for Review 97

Questions for Discussion 97

Terms 97

5 Rhetoric at Rome

101

Roman Society and the Place of Rhetoric 102
Rhetoric and Political Power 102
Rhetoric and Roman Education 104
The Rhetorical Theory of Cicero 106
De Inventione and Rhetorica ad Herennium 106
Studying Invention and Memory 108
Cicero's De Oratore 111

Quintilian 115

The Toga 116

Rhetoric and the Good Citizen 117

The End of Cicero's Life 115

Educating the Citizen–Orator 117

Longinus: On the Sublime 119

The Emotive Power of Language 119

Rhetoric in the Later Roman Empire 123

The Second Sophistic 123

Conclusion 124

Questions for Review 125

Questions for Discussion 125

Terms 126

\sim		
(on	tonte	

6 Rhetoric in Christian Europe and North Africa

131

Rhetoric, Tension, and Fragmentation 131

Rhetoric in the Early Middle Ages 133

St. Augustine 133

Augustine's Rhetorical Theory 134

De Doctrina Christiana 135

Augustine and Rhetorical Education 136

Martianus Capella 137

Boethius 138

Rhetoric and Medieval Education 138

Disputatio and Sententiae 139

Women and Education 139

Rhetorical Continuity 140

Rhetorical Arts in the High Middle Ages 140

The Art of Preaching 141

The Art of Letter Writing 142

Letter-Writing Skill 146

Women and Letter Writing 148

The Art of Poetry 150

Marie de France 152

Conclusion 153

Questions for Review 154

Questions for Discussion 155

Terms 155

7 Rhetoric in the Renaissance

160

Women Rhetoricians of the Renaissance 161 Christine de Pisan 162

_

Italian Humanism 163

Marks of the Umanista 163 Notaries 164

Retrieving the Classical Tradition 165

Petrarch 166

Pico della Mirandola 168

Lorenzo Valla 169

Juan Luis Vives 171

Rhetoric and Renaissance Education 172

Rhetoric and Renaissance Art 174

The Turn toward Dialectic 174

Agricola 174

Peter Ramus 175

Renaissance Rhetoric in Britain 176

Teaching Rhetoric 177

Magic, Science, and Style 177

The Late Renaissance and Conversational Rhetoric 178

Madame Madeleine de Scudéry 178

Contents

	Conclusion 179 Questions for Review 179 Questions for Discussion 180 Terms 180	
8	Enlightenment Rhetoric Rhetoric's Changing Roles 186 Margaret Cavendish 186 Writing Career 186 Vico and the Rhetoric of Human Thought 187 Vico vs. Descartes 187 Rhetoric and Imagination 188 Rhetoric and the Evolution of Human Thought 189 Vico on Myth: The Civilization-Shaping Force of Narrative 190 Influence 191 British Rhetoric in the Eighteenth Century 191 Rhetoric in British Education 191 The Elocutionary Movement 192	185
	The Scottish School 193 The Belletristic Movement 194 Lord Kames 195 Hugh Blair 196 George Campbell 197 Richard Whately Revives Classical Rhetoric 199 Maria Edgeworth 201 Conclusion 202 Questions for Review 203 Questions for Discussion 203 Terms 204	
9	Contemporary Rhetoric I: Arguments, Audiences, and Science Argumentation and Rational Discourse 209 Perelman and Olbrechts-Tyteca: A New Rhetoric 209 The Centrality of Audience 210 Jürgen Habermas on a Rational Society 212 The Rhetoric of Science 215 Precursors 216 A Rhetorical Approach to Science 218 The Power of Unexamined Science 218 Advocacy in the Sciences 219 Science and Communities 219 Deirdre McCloskey and the Rhetoric of Economics 220 Rhetoric in Anthropology 220 John Campbell on the Rhetoric of Charles Darwin 221 Conclusion 224	208

\sim	
(on	tonte

Questions for Review 225 Questions for Discussion 225 Terms 225

10 Contemporary Rhetoric II: Narrative, Display, and Objects

230

258

Rhetoric in Context 230

Kenneth Burke and Rhetoric as Symbolic Action 231 Lloyd Bitzer and Rhetoric as Situational 237

Rhetoric and Narration 239

Changes 239

Mikhail Bakhtin and the Polyphonic Novel 240

Wayne Booth and the Rhetoric of Fiction 242

Authors as Advocates 242

Reflecting on Narrative 243

Walter Fisher on the Rhetoric of Narration 244

Ernest Bormann on Stories and Communities 245

The Rhetoric of Display 246

The Rhetoric of Things 247

Digital Rhetorics 248

A Contest for Attention 249

A New Literacy 250

Conclusion 251

Questions for Review 252

Questions for Discussion 252

Terms 253

11 Contemporary Rhetoric III: Texts, Power, and Alternatives

Postmodern Rhetoric 259

Questioning the "Taken for Granted" 259

Jean-François Lyotard and Postmodernism 259

Michel Foucault 260

Queer Theory 264

Jacques Derrida 265

Feminism and Rhetoric 268

The Challenge of Male Language 269

The Loss of a Woman's Voice 269

Invention and Theory 271

Constructing Gender Rhetorically 272

Solutions: Reconceptualizing Rhetoric 272

Rejecting Rhetoric as Conquest 273

Rhetoric as Invitation 274

"Works," "Texts," and the Work of Reading 274

Feminism and the Ancient Tradition 275

Comparative Rhetoric 276

Argument in Ancient Indian Rhetoric 276

African Rhetorical Forms 277

Contents

George Kennedy on Non-European Rhetorics 278
Aztec and Egyptian Rhetorics 278
Rhetoric in Ancient China 279
Chinese Sophists and the Intrigues of the Warring States 279
Jian, Shui, Pien, and the Traveling Persuaders 281
Averroes and the Aristotelian Tradition: Arabic Rhetoric in the Twelfth Century 281
Re-visioning the Greek Tradition 282

Conclusion 283

Questions for Review 284 Questions for Discussion 284 Terms 285

Glossary	292
Bibliography	301
Index	315

My goal in the seventh edition of *The History and Theory of Rhetoric*, as in previous editions, is to provide students with an engaging and accessible survey of the history of rhetoric, primarily as it has developed in the Western world. This text also equips students with a conceptual framework for evaluating and practicing persuasive writing and speaking in a wide range of settings and in various written and visual media.

Each chapter introduces readers to influential theories of rhetoric advanced by some of history's greatest thinkers. Through encountering the rhetorical tradition, students are better prepared to understand and participate in the wide array of persuasive symbolic practices that mark our social world, and develop, recognize, and evaluate rhetorical qualities in a wide range of texts, including the growing world of digital rhetoric.

NEW TO THIS EDITION

I have sought to make the writing in this edition clear and economical, and the content readily comprehensible. I have added material to several of the historical chapters in an effort to bring into clearer focus the cultural contexts of the rhetorical theories discussed. Additional biographical details have been added to the description of several particularly important theorists and the coverage of comparative and feminist rhetorics has been expanded in this edition. I have also further developed the discussion of materialist approaches to rhetoric, and the complex networks of influence presented in Actor-Network Theory. These recent theoretical perspectives expand the traditional notion of rhetorical agency, even as they challenge the notion of exclusively human agents.

OVERVIEW OF CHAPTERS

Chapter 1 presents several defining characteristics of rhetoric as a type of discourse, as well as discussing the social functions of the art of rhetoric. I have also sought to remain mindful of the tensions and ambiguities that always attend efforts to define rhetoric. The question of definition is rendered more nuanced with the recognition of

visual rhetoric as a major component of our corporate symbolic activity. With this foundation in place, the balance of the text is organized historically.

Chapter 2 considers the Sophists as early teachers, practitioners, and theorists of rhetoric. The chapter examines the ways in which these experimental and controversial rhetoricians shaped not just the practice of rhetoric, but also our understanding of the symbolic nature of human existence. While I do address the controversy the Greek Sophists generated, their role as theorists who achieved considerable insight into rhetoric's nature and power is also emphasized. I have tried to demonstrate that the Sophists do not all belong to a single school of thought regarding rhetoric but vary in their approaches to the art.

Chapter 2 also addresses the issue of women's voices in ancient Greek rhetoric. The chapter introduces the poet Sappho, considers women as empowered voices in the Spartan marketplace, and discusses the rhetorical innovator Aspasia.

Chapter 3 takes up Plato's famous, and controversial, criticism of the Sophists in the dialogue *Gorgias*. The centrality of Plato's insistence on an account—a *logos*—from the Sophists concerning the nature of their art is recognized. The chapter also addresses the great philosopher's musings about a true art of rhetoric in *Phaedrus*.

Chapter 4 explores Aristotle's highly influential rhetorical theory advanced in his *Rhetoric*. This chapter attends to Aristotle's assessment of rhetoric as a *techne* or true art. The concepts of the enthymeme, artistic proofs, and *topoi* of argument are all reviewed.

Chapter 5 considers Roman adaptations of Greek rhetoric to a new cultural context. The striking differences between the Roman and Greek conceptions of the citizen are emphasized. Recent scholarship provides a clearer view of rhetorical theory and practice in Rome. Key components in Cicero's theory, including his famous canons of rhetoric and his concern for the preparation of the orator-leader, remain focal points. Roman attention to judicial oratory and argument is also stressed. The contributions of Quintilian and Longinus to rhetorical thought are explored, as is the close relationship between rhetoric and citizenship in ancient Rome.

Chapter 6 explores the theories and uses of rhetoric that characterized the medieval period, first in North Africa and later in Europe. St. Augustine's adaptation of the Roman rhetoric he knew so well to the needs of the Christian church is discussed. Other early medieval rhetoricians such as Martianus Capella and Boethius are also covered. In addition, the chapter examines the rhetorical arts that developed in the later Middle Ages, including preaching, letter writing, and poetry. A new section has been added on the letter-writing activity of Alcuin, an early practitioner of the art. Other important figures representing the medieval rhetorical arts are also introduced. The emergence of a female reading public is addressed, as is the corresponding rise of female writers such as Marie de France.

Renaissance rhetorical theory and the Italian Humanists remain the focus of Chapter 7. The Humanists' intense interest in classical texts and languages, the arrival in fifteenth-century Italy of large numbers of Greek texts from Byzantium, and the period's fascination with Cicero, led to the extraordinary impact of rhetoric on European education and civic life.

The entry of women into rhetorical roles in a substantial fashion during the Renaissance is also outlined, as are women's increased access to education and fora of communication. The tension between contemplative and active lives is discussed, as is rhetoric's role in the rise of commercial cities, and its encouragement of

Renaissance preoccupation with magic. The late-Renaissance emergence of the conversational model of rhetoric is noted, as is rhetoric's struggle with dialectic and a scientific style of writing.

Chapter 8 focuses on the period from 1650 through 1850. The chapter discusses the intriguing Italian theorist Giambattista Vico, who saw myth as foundational to civilization and assigned rhetoric a central role in the evolution of human thought. The chapter also considers writers of the Scottish Enlightenment, including George Campbell, Lord Kames, and Hugh Blair. English rhetorical theorists, including the elocutionist Thomas Sheridan and traditionalist Richard Whately, are also introduced. The chapter opens and closes with women writers—Margaret Cavendish and Maria Edgeworth—whose rhetorical efforts stood largely outside, and as a challenge to, the mainstream of the Enlightenment rhetorical tradition.

Interest in rhetoric during the second half of the twentieth and early twenty-first centuries is explored in the book's concluding three chapters. Chapter 9 focuses on rhetorical theories developing around perennial rhetorical constituents of argument, audience, and public discourse. Chaim Perelman's and L. Olbrechts-Tyteca's audience-and argument-based "new rhetoric" is discussed. Jürgen Habermas' theory of how to create a more rational society is also taken up, with attention to his prescriptions for a rational public arena. The chapter also explores the various ways in which science can be understood as rhetorical.

Chapter 10 explores rhetoric as preparation for living effectively in the always evolving world of symbols. A discussion of Kenneth Burke explores his leading ideas and his central theory of dramatism. The chapter examines Burke's famed pentad and associated discussion of the worldviews. Lloyd Bitzer's situational theory is also introduced here. Mikhail Bakhtin's and Wayne Booth's influential theories of the rhetoric of narration receive careful treatment as well, as do the narrative theories of Walter Fisher and Ernest Bormann.

An expanded section in Chapter 10 introduces the rhetoric of material objects, concepts such as circulation and generalized symmetry, and Actor-Network Theory (ANT). The crucial issue of rhetorical agency is addressed. Visual rhetoric or the rhetoric of display is introduced as adding an important and often missing dimension to rhetorical analysis. Finally, this chapter considers how digital media affect the rhetorical experience, and may call for new and broader conceptions of rhetoric.

Chapter 11 considers several theories of symbol use and public discourse shaped by Continental criticism and postmodern thought. The rise of postmodernism in Europe is addressed, setting several major figures in their intellectual context. Michel Foucault's insights into the close connections among discourse, power, and knowledge are discussed, as is Jacques Derrida's critique of the instability of language itself. Queer Theory, closely related to Foucault's theoretical insights, is introduced.

An expanded section considers various strains of feminist thinking about rhetoric, moving the discussion of feminism beyond efforts to retrieve a distinctly feminine rhetorical voice. This chapter also features a discussion of several non-Western conceptions of rhetoric, with attention to how the competitive nature of ancient Greek culture dramatically affected Western rhetoric. The contributions of Muslim scholars to rhetorical studies during the Middle Ages are explored, focusing on the work of Averroes.

The seventh edition of *The History and Theory of Rhetoric* seeks to integrate material within and among chapters, and to describe the intellectual and historical

contexts for each theorist and school of thought. Each chapter includes a list of key terms as well as questions for review and for discussion. A complete glossary of terms should also be useful for review of important concepts, and the bibliography can be of assistance to students who wish to do additional reading on a particular topic or theorist. A detailed instructor's manual can be found on the companion website, available at www.routledge.com/cw/herrick.

The centrality of symbolic activity to our public and private lives propels our interest in symbols and their strategic use; the record of this interest constitutes the history of rhetoric. Our reliance on rhetorical interaction for the development and maintenance of cooperative social arrangements makes the history and theory of rhetoric a crucial study for all informed people. Given the pluralistic nature of contemporary society and the resulting necessity of discovering equitable compromises through discourse, the study of rhetoric is more relevant today than it ever has been.

The rational flexibility demanded by our increasing reliance on digital forms of communication provides further justification for a concentrated focus on insights from the history of rhetoric. I hope that this new edition of *The History and Theory of Rhetoric* captures the vitality and adaptability of rhetoric as the art of the purposeful and productive management of all types of symbols.

I would like to thank Mrs. Linda Koetje for her help in preparing the manuscript. I am also indebted to several reviewers for their insightful comments that prepared the way for this new edition. These respondents include Robert W. Barnett, University of Michigan-Flint; Ferald J. Bryan, Northern Illinois University; Catherine A. Dobris, Indiana University–Purdue University Indianapolis; Belle A. Edson, Arizona State University; Nichole Kathol, University of Kansas; and Kathleen Torrens, University of Rhode Island.

James A. Herrick



My first problem lies of course in the very word "rhetoric."

—Wayne Booth, The Vocation of a Teacher

,

By rhetorical, I refer to something's ability to induce change in thought, feeling, and action; organize and maintain collective formation; exert power, etc.; as it enters into relation with other things (human or nonhuman).

-Laurie Gries

HIS chapter explores the history, theories, and practices of rhetoric. But, as literary critic Wayne Booth (1921–2005) suggested in the quotation above, the term *rhetoric* poses some problems at the outset because of the various meanings it has acquired. For some people, rhetoric is synonymous with empty talk or even deception. We hear clichés like, "That's mere rhetoric" or "That's just empty rhetoric," which are used to undermine or dismiss a comment or opinion.

Meanwhile, rhetoric has once again emerged as an important topic of study, and its significance to public discussion of political, social, religious, and scientific issues is now widely recognized. Scholars and teachers express great interest in the subject; colleges and universities offer courses in rhetoric; and dozens of books are published every year with rhetoric in their titles. Clearly, rhetoric arouses mixed feelings—it is a term of derision and yet a widely studied discipline, employed as an insult and still recommended to students as a practical subject of study. What is going on here? Why all the confusion and ambiguity surrounding the term *rhetoric*?

Negative attitudes toward rhetoric are not of recent origin. In fact, one of the earliest and most influential critical discussions of rhetoric occurs in Plato's dialogue *Gorgias*, a work written in the opening decades of the fourth century BCE when rhetoric was popular—though also highly controversial—in the Greek city-state of Athens. The great philosopher, as his dialogue makes clear, takes a dim view of rhetoric, at least as practiced by some teachers of the day called Sophists. The character Socrates, apparently representing Plato's own perspective, argues that the type of rhetoric being taught in Athens was simply a means by which "naturally clever" people "flatter" their unsuspecting listeners into agreeing with them and doing their bidding. Plato condemns rhetoric as

"foul" and "ugly," a judgment that has haunted the discipline ever since. We will discuss his specific criticisms of rhetoric in Chapter 3, note that Plato was involved in an ongoing debate about the topic, and consider that he apparently changed his perspective on rhetoric later in his life.

Ever since Plato's *Gorgias* first appeared, rhetoric has struggled to redeem its tarnished public image. Rhetoric bashing continues in an almost unbroken tradition from ancient times to the present. In 1690 another respected philosopher, John Locke (1632–1704), advanced a view of rhetoric not unlike, and likely influenced by, Plato's. The following quotation represents Locke's writing in his highly influential book, *An Essay Concerning Human Understanding*:

If we speak of things as they are, we must allow that all the art of rhetoric, besides order and clearness; all the artificial and figurative application of words eloquence hath invented, are for nothing else but to insinuate wrong ideas, move the passions, and thereby mislead the judgment; and so indeed are perfect cheats²

Locke does acknowledge that one aspect of rhetoric, what he calls "order and clearness," is useful. However, he rejects the study of "artificial and figurative" language as deceptive. As we will see in Chapter 7, Locke was immersed in a debate about figurative language when he expressed this opinion—so he was hardly a neutral witness. He was also aware that the greatest English language master of rhetoric—William Shakespeare (1564–1616)—lived just a few decades earlier.

The nineteenth-century German philosopher and classicist Friedrich Nietzsche (1844–1900)—who made a serious study of rhetoric—wrote, "We call an author, a book, or a style 'rhetorical' when we observe a conscious application of artistic means of speaking; it always implies a gentle reproof." A "gentle reproof" certainly reflects a more measured assessment than Locke's "perfect cheats."

But, Nietzsche was aware of something else, something deeper and more fundamental, lurking in the realm of the rhetorical:

[I]t is not difficult to prove that what is called "rhetorical," as a means of conscious art, had been active as a means of unconscious art in language and its development, indeed, that the *rhetorical is a further development*, guided by the clear light of the understanding, of *the artistic means which are already found in language*.

What does Nietzsche mean by the curious phrase, "the artistic means already found in language"? Is he, perhaps, suggesting that language itself possesses an irreducible artistic or aesthetic quality that rhetoric merely draws out? He continues:

There is obviously no unrhetorical "naturalness" of language to which one could appeal; language itself is the result of purely rhetorical arts. The power to discover and to make operative that which works and impresses, with respect to each thing, a power which Aristotle calls rhetoric, is, at the same time, the essence of language 3

If Nietzsche is correct that nothing in the realm of language is purely "natural" and unmarked by "rhetorical arts," that rhetoric is "the essence of language," then rhetoric

is certainly a matter that deserves our attention. Few disciplines can make such a comprehensive claim regarding their consequence for both public and private life.

An Overview of Rhetoric

RE-EVALUATING RHETORIC

Opinion about rhetoric has always been dramatically divided. In recent decades a number of prominent writers have re-evaluated rhetoric, sometimes arriving at surprising—and potentially paradigm-shifting—conclusions.

Wayne Booth, whom we have already encountered, was one of the twentieth century's leading literary critics. Booth affirmed that rhetoric held "entire dominion over all verbal pursuits. Logic, dialectic, grammar, philosophy, history, poetry, all are rhetoric." Entire dominion? *All* verbal pursuits are rhetoric? What could Booth have had in mind in making such sweeping assertions regarding rhetoric?

Nevertheless, Booth is not alone in maintaining such a stunning view of rhetoric. Another important twentieth-century literary scholar, Richard McKeon (1900–1985), expressed virtually the same opinion. For McKeon, rhetoric was best understood as "a universal and architectonic art." Rhetoric is universal, that is, present everywhere we turn. But what about *architectonic*? McKeon meant that rhetoric organizes and gives structure to *all* the other arts and disciplines, that it is a kind of master discipline that orders and lends form to other undertakings. This is because rhetoric is, among other things, the study of how we organize and employ language effectively. Thus, it becomes the study of how we organize our thinking on a wide range of subjects.

In apparent agreement with Booth and McKeon, Richard Lanham (b. 1936) of the University of California has called for a return to rhetorical studies as a way of preparing us to understand the impact of digitization on how we read and write. Rather than developing a completely new theory of literacy for the digital age, Lanham argues that "we need to go back to the original Western thinking about reading and writing—the rhetorical *paideia* [educational program] that provided the backbone of Western education for two thousand years." For Lanham, the study that originally taught the Western world its approach to education and communication—rhetoric—can still teach us new things, like how to adapt to the emerging world of digital communication.

Professor Andrea Lunsford (b. 1942), Director of Stanford University's Program in Writing and Rhetoric, is among a growing number of scholars who, like Lanham, have returned to rhetoric as providing guidance in understanding how the digital revolution is shaping our reading and writing habits. After analyzing thousands of students writing samples—including blogs, tweets, and classroom assignments—Lunsford and her colleagues concluded that students today expect their writing to change the world they live in. For today's students "good writing changes something. It doesn't just sit on the page. It gets up, walks off the page and changes something."

Rhetoric scholar Laurie Gries brings a rather different—and highly consequential—perspective to rhetoric, writing: "By *rhetorical*, I refer to something's ability to induce change in thought, feeling, and action; organize and maintain collective formation; exert power, etc.; as it enters into relation with other things (human or nonhuman)." Notice that Gries refers to "something's" rhetorical capacity, broadening rhetorical agency beyond human beings and thus beyond language.

Booth, McKeon, Lanham, Lunsford, and Gries find much to commend in the study that Plato condemned as "foul and ugly," and ask us to reconsider those elements of eloquence that Locke referred to as "perfect cheats." It appears that we are at a point in our cultural history where rhetoric is re-establishing itself as an important study with insights to offer about a surprisingly broad spectrum of human—and even non-human—communication activities.

At the same time, we must acknowledge that the practice of rhetoric maintains its Jekyll and Hyde quality, shifting without notice from helpful and constructive to deceptive and manipulative. Why does this study of the effective uses of language and other symbols prove so difficult to evaluate, eliciting as it does such sharply opposed judgments? A complete answer to this question requires some knowledge of rhetoric's long history, which is the subject of this book. But almost certainly, rhetoric's mixed reviews have a lot to do with its association with persuasion, that most suspect but essential human activity. A brief digression to explore this connection between rhetoric and persuasion will be worth our while.

RHETORIC AND PERSUASION

Though there is more to the study of rhetoric than persuasion alone, rhetoric traditionally has been closely concerned with linguistic techniques for gaining compliance. This long-standing association with persuasion has been at the heart of the conflict over whether rhetoric is a neutral tool for bringing about agreements, or a dubious activity that ends in manipulation.

Rhetoric's intimate connection with persuasion has prompted both intense suspicion and broad interest. After all, we all are leery of persuasion. Who has not had a bad experience as the object of someone else's persuasive efforts? Think of the last time you knew you were being persuaded by a high-pressure sales technique, a religious advocate, a politician, a professor, or simply by a friend or family member. Something in you may have resisted the persuasive effort, and you may even have felt some self-protective irritation. But you may also have felt you were being drawn in by the appeal, that you were, in fact, being persuaded. If the person doing the persuading was employing rhetorical techniques, you might conclude that you had some reason to distrust both rhetoric and the people who practice it. So, most of us have developed a healthy suspicion of persuasion, and perhaps a corresponding mistrust of rhetoric.

At the same time, a moment's thought suggests that all of us seek to persuade others on a regular basis. Many professions, in fact, require a certain understating of and capacity to persuade. Persuasion can even be understood as an important part of economics and the world of work. Economist Deirdre McCloskey (b. 1942) has argued that "persuasion has become astonishingly important" to the economy. She estimated, for example, that one quarter of the work force depended on skill with words to do their work. What has she concluded? "I gradually realized that the *economy* ... is rhetorical. An economy is continuously negotiated *with words*." McCloskey adds, "an economy is a conversation." She explains: "The point is that the economy is very largely about persuasion, because it is negotiated and innovative and above all because it is about a future to which we are vulnerable." 10

But, what about in our private lives? It seems we remain perpetual persuaders in our personal relationships. Who does not make arguments, advance opinions, and

seek compliance from friends? Moreover, we typically engage in these persuasive activities without thinking we are doing anything wrong. In fact, it is difficult *not* to persuade; we participate in the practice on an almost daily basis in our interactions with friends, colleagues at work, or members of our family. We may attempt to influence friends or family members to adopt our political views; we will happily argue the merits of a movie we like; we *are* that salesperson, religious advocate, or politician. It is difficult to imagine a relationship in which persuasion has no role, or an organization that does not depend, to some degree, on efforts to change other people's thoughts and thus to influence their actions.

Let us consider some additional examples of how universal persuasion can be. We usually think of sports as a domain of physical competition, and not of verbal battles. Yet, even sports involve disagreements about such things as the interpretation of rules, a referee's call, or which play to execute. And, these disagreements often are settled by arguments and appeals of various kinds, that is, by persuasion. British psychologist Michael Billig (b. 1947) notes that many of the rules governing a sport result from rhetorical interactions about such issues as how much violence to allow on the field of play. He writes, "The rules of rugby and soccer were formulated in order to transform informal agreements, which had permitted all manner of aggressive play, into defined codes that restricted violence." Rhetoric, especially its argumentative aspect, was crucial to the creation of these rules of play. "Above all, the rules were formulated against a background of argument." Even the rules by which athletes compete, it appears, came into being through rhetoric.

Is the same true in a technical field like medicine? If medicine is a science, should not argument and persuasion be non-existent? In fact, medical decisions are often made after a doctor advances a convincing case for or against a particular procedure in a rhetorical exchange with other doctors. And, the decision-making exchange often is not limited to technical issues, such as the interpretation of medical data like the results of a blood test. To be sure, the arguments advanced will involve medical principles, but they are arguments nonetheless, they are intended to be persuasive and they range beyond strict medical guidelines.

In medical dialogue we are likely to hear ethical concerns raised, the wishes of a family considered, and even questions of cost evaluated. Moreover, the patient often has to be persuaded to take a particular medicine, follow a specified diet, or allow doctors to perform a surgical procedure. As physicians argue, rival medical theories may be in conflict and rival egos may clash. Who should perform a needed corneal transplant on a famous politician? We might think that such an important decision would be based on medical criteria alone. Yet, even a question like this may be resolved on the basis of arguments between two well-known physicians at competing hospitals. Clearly, the science of medicine has its rhetorical side.

Bringing the focus down to a more personal level, does romance involve persuasion? When I seek the attention of someone in whom I am romantically interested, I start to develop a case—though perhaps not an explicit and public one—about my own good qualities. When in the vicinity of the individual concerned, I may attempt to appear humorous, intelligent, and considerate. My words and actions take on a rhetorical quality as I build the case for my own attractiveness. I might be convincing, or may fail to convince, but in either event I have made choices about how to develop my appeal, so to speak. Once begun, romantic relationships go forward (or backward)

on the basis of persuasive interactions on topics ranging from how serious the relationship should be to whether to attend a particular concert.

Other activities also bring us into the realm of rhetoric. Business transactions, from marketing strategies to contract negotiations, involve persuasive efforts. As McCloskey has pointed out, many people make their living by means of their abilities as persuasive speakers or writers. Nor is education immune from rhetorical influence. You often are aware that a professor is advocating a point of view in a lecture that ostensibly presents simple "information," or that classmates argue with one another hoping to persuade others to their point of view. As a matter of fact, you have been reading an extended persuasive case for the importance of studying rhetoric. Textbooks, it should come as little surprise, often have a persuasive agenda embedded within them.

Efforts at persuasion mark many, perhaps all, of our interpersonal activities. In fact, we even persuade *ourselves*. The internal rhetoric of "arguing with yourself" accompanies most of life's decisions, big or small. Though our experiences may leave us leery of persuasion, it is also an important component of our occupational, social, and private lives.¹²

Now, back to rhetoric. If rhetoric is, in part, the systematic study of persuasion, recognizing how crucial persuasion is to daily life may suggest that this art deserves our attention. To acknowledge what we might call "the pervasiveness of persuasiveness" is not to condemn persuasion or rhetoric. Rather, it is to begin to appreciate the centrality of this activity to much of life, and to recognize that human beings are rhetorical beings. At this point, it will be important to develop a more precise definition of rhetoric.

DEFINING RHETORIC

Scholars have advanced a variety of insightful definitions of our topic. Rhetoric scholar James J. Murphy has suggested "advice to others about future language use" as one way of defining rhetoric. Murphy's definition implies prior study of the topic rhetoric that has resulted in a set of tested theories and reliable techniques. Classicist George Kennedy defines rhetoric more broadly as "the energy inherent in emotion and thought, transmitted through a system of signs, including language, to others to influence their decisions or actions." This definition suggests that rhetoric is simply part of who we are as human beings: Every time we express emotions and thoughts to others with the goal of influence, we express a kind of energy that Kennedy calls rhetoric.

Rhetoric and Symbol Systems

Note that for Kennedy rhetoric involves "signs, including language." I would like to focus attention on this important point for a moment, and suggest that rhetoric develops in the realm of symbols of one type or another. So, what are symbols? An individual word such as *boat* is an example of a symbol. It is a general term referring to any mark, sign, sound, or gesture that communicates meaning based on social agreement. Individual symbols usually are part of a larger symbolic system, such as a language.

Language is the symbol system on which most of us rely for communicating with others on a daily basis. However, many arts and other activities also provide symbolic

resources for communicating. In fact, social life depends on our ability to use a wide range of symbol systems to communicate meanings to one another, and a rhetorical dimension can be detected in many of these.

An Overview of Rhetoric

Music

Musical notation and performance constitute a symbol system, one that uses notes, key, melody, harmony, sound, and rhythm to communicate meanings. Movie sound-tracks provide convenient examples of how the symbol system of music can communicate meaning. For instance, musical techniques were used to enhance audience tension in John Williams's famous score from the movie *Jaws*, as well as Bernard Herrmann's screeching violins in the frightening shower scene in Alfred Hitchcock's *Psycho*. The stirring music of Tchaikovsky's famous *1812 Overture* set the right triumphal note for the opening and closing scenes of the 2006 film *V for Vendetta*. More recently, British composer Steven Price intentionally broke with movie sound-track conventions—for instance, he avoided the use of percussion—in creating his Academy-Award-winning soundtrack for the movie *Gravity* (2013). Price's score perfectly conveys the risks and emptiness of space. Perhaps the rhetoric of music is so well established that we readily understand what it is "saying" to us.

Dance and Acting

Many of the movements in dance are also symbolic because they express meaning on the basis of agreements among dancers, choreographers, and audience members. For instance, three dancers in a row performing the same robotic movement may symbolize the tedium and regimentation of modern life. Similarly, gestures, postures, and facial expressions allow mime artists and actors to communicate with audiences symbolically but without employing the symbols of spoken language. There is no actual connection between pondering a question and scratching your head, and yet a theatrical scratch of the scalp means "I don't know" or "I'm thinking about it" by a kind of unstated social agreement.

Actors and impressionists such as Jim Carrey and Kristen Wiig have mastered a range of physical symbols—gestures, postures, and facial expressions—that allow them to communicate instantly with audiences, often without speaking a word.

Painting

In painting, form, line, color, and arrangement can be symbolic. A stark line of dark clouds may symbolize impending disaster, even though clouds do not typically accompany actual disasters. But, because storms and calamity are sometimes associated, and because we often fear storms, we understand the artist's intent. Norwegian painter Edvard Munch used such a technique in his 1893 painting *Shrik (Scream)*, where a brilliant orange–red sky symbolizes terror. But, then, what does Mona Lisa's slight grin "mean." No doubt Leonardo da Vinci had something in mind in crafting that half smile, but scholars and the public alike have never come to an agreement as to his intentions.

Architecture

The lines, shapes, and materials used in architecture can also be employed symbolically to communicate meaning. The protests by veterans' groups that greeted the unveiling of the Vietnam Memorial in Washington, D.C. were responses to what some observers took

to be the *meaning* of the monument, a meaning with which they did not agree.¹⁵ Much of the monument is below ground, perhaps suggesting invisibility or even death. Is it significant that the memorial cannot be seen from Capitol Hill? The principal material used in the monument is black granite rather than the more traditional and triumphal white marble. The polished surface is covered with the names of the fifty thousand Americans who died in the war rather than with carved scenes of battle and victory. What does the Vietnam Memorial mean? One would be hard-pressed to find its meaning to be, "A united America triumphs again in a foreign war." Nevertheless, each symbolic component prompts one to ask deep and troubling questions about a long and tragic war.

Sports

Perhaps the symbols employed in music, dance, acting, painting, or architecture can be readily understood as rhetorical, as they carry a meaning that can be intentionally selected and refined. However, can an athletic event carry rhetorical significance? Long-distance swimmer Diana Nyad requested permission from the Cuban government to swim the 103-mile distance between Cuba and Florida. The Cuban government reluctantly granted her permission for the swim. "The Cubans don't like the implication of somebody walking out on one of their beaches and swimming away," Nyad said in a 2010 *Los Angeles Times* report. ¹⁶ The symbolism of swimming *away* from Cuba apparently was felt to reflect negatively on the Cuban political system. Nyad finally completed the swim on her fifth attempt in 2013, after 53 hours in the water. A long and, to some, rhetorical swim.

Unexpected Locations

Rhetorical elements can reveal themselves in places we might easily overlook. For example, the typeface in which this book is printed has a rhetorical dimension. Though readers are not directed to notice the statement being made by typeface, each individual font was designed to convey a particular quality, character, or tone. Most textbooks are set in a typeface that appears to readers as serious, intentional, and, of course, legible. The typeface for a wedding invitation, however, might be selected to convey elegance or romance. Certainly if the type in this book were set in a font ordinarily reserved for a wedding invitation, a reader would immediately notice this unusual choice. So, we might say that typeface is selected, like the music in a hotel elevator, in order to not be noticed.¹⁷

Effective Symbolic Expression

While persuasion has long been an important goal of rhetoric, we should perhaps expand the definition of rhetoric to include other goals such as achieving clarity, awakening our sense of beauty, or bringing about mutual understanding. Thus, we can define the art of rhetoric as follows: The systematic study and intentional practice of effective symbolic expression. *Effective* here will mean achieving the purpose of the symbol-user, whether that purpose is persuasion, clarity, beauty, or mutual understanding.

The art of rhetoric can render symbol use more persuasive, beautiful, memorable, forceful, thoughtful, clear, and thus generally more compelling. In all of these ways, rhetoric is the art of employing symbols effectively. Rhetorical theory is the systematic presentation of rhetoric's principles, its various social functions, and how the art

achieves its goals. Messages crafted according to the principles of rhetoric will be an On called rhetorical discourse, or simply rhetoric. An individual practicing the art of rhetoric will occasionally be referred to as a rhetor.

An Overview of Rhetoric

As we have noted, for most of its history, the art of rhetoric has focused on persuasion by employing the symbol system of language. This traditional approach to rhetoric is still important, but recently both rhetoric's goals and the symbolic resources available to those practicing the art have expanded dramatically. Even the centrality of human agents has been challenged by some recent rhetorical movements such as posthumanism, Actor-Network Theory, and new materialism. Such developments have led some scholars to write of different kinds of rhetoric, even different rhetorics. Steven Mailloux notes that "there are oral, visual, written, digital, gestural, and other kinds; and under written rhetoric, there are various genres such as autobiographies, novels, letters, editorials, and so forth"

Does this mean that all communication, regardless of goal or symbol system employed, is rhetoric? Some scholars make communication and rhetoric synonymous, but this seems to ignore genuine and historically important distinctions among types of communication ranging from information and reports through casual conversations to outright propaganda. I will be taking the position that rhetorical discourse is a particular type of communication possessing several identifying characteristics. What, then, are the features of rhetorical discourse that set it apart from other types of communication? The following section describes six distinguishing qualities of rhetorical discourse as we encounter it in writing, speaking, the arts, and other media of expression.

RHETORICAL DISCOURSE

This section considers six distinguishing characteristics of rhetorical discourse; the marks the art of rhetoric leaves on messages. Rhetorical discourse characteristically is (1) planned, (2) adapted to an audience, (3) shaped by human motives, (4) responsive to a situation, (5) persuasion-seeking, and (6) concerned with contingent issues. Not all writing or speaking that might meaningfully be termed rhetoric satisfies all of these criteria, but the criteria will serve as a starting point for identifying, understanding, and responding to rhetorical discourse. We begin by considering rhetoric's most fundamental quality.

Rhetoric Is Planned

Regardless of the goal at which it aims, rhetorical discourse involves forethought or planning. Thinking of rhetoric as planned symbol use directs our attention to the choices people make about how they will address their audiences. Issues that arise in planning a message include the following:

- —Which arguments will I advance?
- —Which evidence best supports my point?
- —How will I order and arrange my arguments and evidence?
- —What resources of language and other symbol systems are available to me, given my topic and audience?

The planned nature of rhetoric has long been recognized as one of its defining features. Some early rhetorical theorists developed elaborate systems to assist would-be orators in planning their speeches. The Roman writer Cicero, for instance, used the term *inventio* (invention) to describe the process of discovering the arguments and evidence for a persuasive case. He then provided specific methods for inventing arguments quickly and effectively. Cicero also discussed the effective ordering of arguments and appeals under the heading *dispositio* (arrangement), while he used the term *elocutio* to designate the process of finding the right linguistic style for one's message, whether elegant or conversational.

Such concerns, already extensively studied in the ancient world, reflect the planned quality that characterizes rhetorical discourse. In subsequent chapters, we will look more closely at a number of rhetorical systems designed to assist the planning of messages.

Rhetoric Is Adapted to an Audience

Concern for forethought or planning suggests a second characteristic of rhetorical discourse. Rhetoric is planned with some audience in mind. *Audience* should not be understood strictly in the traditional sense of a large group of people seated in rows of chairs in a large hall. Some audiences are of this type, most are not.

When you speak to a small group of employees at work, they are your audience and you may adapt your discourse to them. The author of a letter while writing to the editor of a local newspaper also keeps the audience in mind, though the audience is not made up of people whom the author can see or knows personally in most cases. Similarly, a novelist writes with particular groups of readers in mind who constitute his or her audience. A politician may address a vast and diverse national audience by means of mass media.

Typically, a rhetor must make an educated guess about the audience he or she is addressing. This imagined audience is the only one present when a message is actually being crafted, and it often guides the inventional process in important ways. The audience that hears, reads, or otherwise encounters a message may be quite similar to the imagined audience, but even highly trained writers or speakers guess wrongly at times. In demand as a speaker, Wayne Booth pointed out that even when he thought he knew his audience, he was sometimes mistaken:

I always wrote with some kind of imaginary picture of listeners responding with smiles, scowls, or furrowed brows. Such prophecies often proved to be wildly awry: An imagined audience of thirty teachers who would have read the materials I sent them in advance turned out, in the reality faced a week or so later, to be ten teachers, along with two hundred captive freshmen reluctantly attending as part of their "reading" assignment; the audience for a "public lecture" was discovered to contain nobody from the public, only teachers.¹⁹

Booth's experience is not at all unusual. Nevertheless, some effort to estimate one's audience has always been, and remains, a crucial component in the rhetorical process.

Rhetorical discourse forges links between the rhetor's views and those of an audience. This means that speakers, writers, and designers must attend to an audience's values, experiences, beliefs, and aspirations. Twentieth-century rhetorical theorist Kenneth Burke (1897–1993) used the term *identification* to refer to the bond between rhetors and their audiences, finding identification crucial to cooperation, consensus,

compromise, and action. Two other rhetorical theorists have written that rhetoric involves "continuous adaptation of the speaker to [an] audience." ²⁰

An Overview of Rhetoric

Audiences and Attention

Our discussion of audience adaptation should not neglect the obvious concern that a speaker or writer has for keeping an audience's wandering attention. Richard Lanham has described rhetoric as "the economics of attention," that is, as a study concerned with managing the limited resource of audience attentiveness. ²¹ This interest in attention focuses *our* attention on a relatively new concern for students of rhetoric: Scientific studies of the brain are revealing some of the secrets of the audience and of persuasion.

Researchers at the University of Utah medical school took a major step toward understanding how we pay attention to various stimuli in our environment. Lead researcher Jeffery Anderson comments, "This study is the first of its kind to show how the brain switches attention from one feature to the next." Apparently, different parts of the brain process information from the different senses, and a "map" within the brain directs our attention to particular stimuli at any particular moment. "The research uncovers how we can shift our attention to different things with precision," says Anderson. "It's a big step in understanding how we organize information." Rhetorical scholars will no doubt be interested in studying such attention maps. The issue of attention is now widely studied, with some research suggesting that our attention spans are getting shorter. 23

Scientists are not the only ones studying attention. Brian Boyd, an expert on narrative, notes that "To hold an audience, in a world of competing demands on attention, an author needs to be an inventive intuitive psychologist."²⁴ Rhetorical theorists from ancient times to the present would agree—attracting and holding audience attention requires that the skillful rhetor become a student of the human mind, that is, of psychology. Attracting and holding audience attention is a central concern of the public advocate, and much of the art of rhetoric is directed to achieving this goal.

Rhetoric Reveals Human Motives

A third quality of rhetoric is closely related to the concern for the audience. Any study of rhetoric will reveal people acting symbolically in response to their motives, a term taking in commitments, goals, desires, or purposes that lead to action. Rhetors address audiences with goals in mind, and the planning and adaptation processes that mark rhetoric are governed by the desire to achieve these goals.

Motives that animate rhetorical discourse include making converts to a point of view, seeking cooperation to accomplish a task, building a consensus that enables group action, finding a compromise that breaks a stalemate, forging an agreement that makes peaceful coexistence possible, wishing to be understood, or simply having the last word on a subject. Rhetors accomplish such goals by aligning their own motives with an audience's commitments. For this reason, the history of rhetoric is replete with efforts to understand human values, to identify factors prompting audiences to action, and to grasp the symbolic resources for drawing people together.

Of course, there are good and bad motives. Imagine, for instance, a governor running for president. As you study the governor's public statements, you look for motives animating that rhetoric: Is the governor concerned with serving the public

good? Does she hope to see justice prevail? Is fame a motive, or greed? Perhaps all of these elements enter the governor's motivation. Of course, motives may be either admitted or concealed. The same politician would likely admit to desiring the public good but would be unlikely to admit to seeking fame, fortune, or even merely employment. Any informed critic of rhetoric must be aware that motives may be elusive or clearly evident, hidden, or openly admitted.

Rhetoric Is Responsive

The fourth quality of rhetorical discourse typically is a response either to a situation or to a previous rhetorical statement. By the same token, any statement, once advanced, is automatically an invitation for other would-be rhetors to respond. Rhetoric, then, is both "situated" and "dialogic." What does it mean for rhetoric to be situated? Simply that rhetoric is crafted in response to a set of circumstances, including a particular time, location, problem, and audience.

The situation prompting a rhetorical response may be a political controversy concerning welfare, a religious conflict over the role of women in a denomination, a debate in medical ethics over assisted suicide, the discussions about a policy that would control visitors in university dormitories, a natural disaster, or a theatrical performance in which a plea for racial harmony is advanced. Rhetoric is response-*making*.

But, rhetoric is also response-*inviting*. That is, any rhetorical expression may elicit a response from someone advocating an opposing view. Aware of this response-inviting nature of rhetoric, rhetors will imagine likely responses as they compose or "invent" their rhetorical appeals. They may find themselves coaxing their mental conception of a particular audience to respond the way they think the actual audience might. The response-*inviting* nature of rhetoric is easy to imagine when we are envisioning a setting such as a political campaign or a courtroom. But does rhetoric also invite response in less formal settings?

Think of a conversation between yourself and a friend regarding buying expensive tickets for a concert. You have given some thought to what you might say to persuade your friend to buy tickets for the concert, and you are even aware of the response your arguments will receive. Your first argument runs something like this: "Look, how often do you get to hear the Chicago Symphony live? And besides, it's only thirty bucks." You have argued from the rareness of the experience and the minimal costs involved. But your friend, ever the studied rhetor, is ready with a response: "Hey, thirty bucks is a lot of money, and I haven't paid my sister back the money she loaned me last week." Your friend has argued from the magnitude of the costs, and from a moral commitment to fulfilling prior obligations. Not to be denied your goal by such an eminently answerable argument, you respond: "But your sister has plenty of money, and thirty bucks is barely enough to buy dinner out."

And so it goes, each rhetorical statement invites a response. Maybe you persuade your friend, maybe you do not. But the rhetorical interaction will likely involve the exchange of statement and response so characteristic of rhetoric.

Rhetoric Seeks Persuasion

As we noted earlier in this chapter, the factor most often associated with rhetorical discourse has been its pursuit of persuasion. Though rhetoric often pursues other

goals, such as inquiry, beauty, or clear expression, it is important to recognize the centrality of persuasion throughout rhetoric's long history, perhaps especially in the Western world. Greek writers noted more than 2,500 years ago that rhetorical discourse sought persuasion, and a late twentieth-century rhetorical theorist can still be found stating straightforwardly that "the purpose of rhetoric is persuasion." It may be helpful, however, to imagine a spectrum running from texts with relatively little persuasive intent (e.g., a news report on a link between stress and childhood obesity) to texts that are strictly persuasive in nature (e.g., a candidate's campaign speech).

Rhetorical discourse often seeks to influence an audience to accept an idea, and then to act. For example, an attorney argues before a jury that the accused is guilty of a crime. The attorney seeks the jurors' acceptance of the idea that the defendant is guilty, and the resulting action of finding the defendant guilty. Or, perhaps I try to persuade a friend that a candidate should be elected mayor on the basis of the candidate's plans to improve education in the city. I want my friend to accept the idea that this candidate is the best person for the job, and to take the action of voting for my candidate. Let us shift our focus to the arts. A play reveals through the symbols of the theater the vicious nature of racism. The play's author hopes both to influence the audience's thinking about racism and to affect the audience's actions on racial matters.

How does rhetorical discourse achieve persuasion? Speaking in the most general terms, rhetoric employs various resources of symbol systems such as language. Four such resources have long been recognized as assisting the goal of persuasion: arguments, appeals, arrangement, and aesthetics.

Argument

An argument is made when a conclusion is supported by reasons. An argument is simply private reasoning made public with the goal of influencing an audience. In fact, researchers Hugo Mercier and Dan Sperber have suggested that the entire purpose of our public reasoning—of making arguments—is to demonstrate to *others* that we have support for our views. This view has been labelled as The Argumentative Hypothesis. In other words, reasoning is not principally a matter of clarifying our own thinking but of creating a rhetorical presentation of our views for an audience. "We outline an approach to reasoning," they write, "based on the idea that the primary function for which it evolved is the production and evaluation of arguments in communication."

Suppose that I wish to persuade a friend of the following claim: "The coach of the women's basketball team ought to be paid the same salary as the coach of the men's team." To support this claim, I then advance the following two reasons:

First, the coach of the women's team is an associate professor, just as is the coach of the men's team. Second, the women's coach has the same responsibilities as the men's coach: to teach two courses each semester, and to prepare her team to play a full schedule of games.

I have now made an argument, and have sought to persuade my friend through a demonstration of my reasoning. Rhetoricians have long associated argument with the public practice of rhetoric, as will become clear from subsequent chapters.

Though we typically think of arguments as occurring in traditional texts such as speeches or editorials, they are not limited to such verbal documents. For example, music critic Tom Strini has written of conductor Andreas Delfs' "uncommon grasp of Beethoven's dramatic rhetoric" and even of the conductor's ability to discover "Beethoven's grand plan" in his Ninth Symphony. Perhaps more surprising, however, is Strini's comment that Delfs' conducting allowed his audience to "follow Beethoven's arguments" in this famous symphony. Specifically, Strini takes the Ninth Symphony to be the great composer's argument in favor of democracy.²⁷ Was Beethoven's symphony, then, a public argument for the correctness of his political views? From a rhetorical point of view, the answer to this question may be yes!

Appeals

Appeals are symbolic strategies that aim to elicit an emotion or engage the audience's value commitments. We are all familiar with emotional appeals such as those to pity, anger, or fear. You probably also have encountered appeals to authority, to patriotism, or to organizational loyalty.

Appeals can be difficult to distinguish from arguments, the difference often being simply one of degree. An argument is directed to reason, an appeal to something more visceral such as an emotion, a conviction, or feeling of which we may not be consciously aware. For instance, an advertisement shows a young woman standing in front of an expensive new car while cradling a baby in her arms. The caption reads: "How much is your family's safety worth?" Though an argument is implied in the picture and caption, the advertisement is structured as an appeal to one's sense of responsibility. Even if reason responded, "Yes, safety is worth a great deal, but I still can't afford that car," the advertisement's appeal could perhaps still achieve its intended effect.

Arrangement

Arrangement refers to the planned ordering of a message to achieve the effect of persuasion, clarity, or beauty. A speaker makes the decision to place the strongest of his or her three arguments against animal experimentation last in a speech to a local civic organization. He or she believes that his or her strongest argument stands to have the greatest impact on his or her audience if it is the last point they hear.

Speakers and writers make many such decisions about arrangement in their messages, but the designers of a public building often make similar decisions. The Holocaust Museum in Washington, D.C., for instance, is physically arranged to make the strongest case possible against the racial hatred that resulted in the horrors of the concentration camps, and against all similar attitudes and actions. Careful planning went into decisions about which scenes visitors would encounter as they entered the museum, as they progressed through it, and as they exited. The great impact of this museum is enhanced by its careful arrangement, a concern the famous rhetorician Cicero referred to as *dispositio*.

Aesthetics

Aesthetics are elements adding form, beauty, and force to symbolic expression. Writers, speakers, composers, or other sources typically wish to present arguments and appeals in a manner that is attractive, memorable, or perhaps even shocking to the intended audience.

Abraham Lincoln's "Second Inaugural Address" is a striking example of language's aesthetic resources employed to memorable and moving effect. Consider the use of metaphor, allusion, consonance, rhythm, and rhyme in the following lines:

An Overview of Rhetoric

Fondly do we hope, fervently do we pray, that this mighty scourge of war may speedily pass away. Yet if God wills that it continue until all the wealth piled by the bondsman's two hundred and fifty years of unrequited toil shall be sunk, and until every drop of blood drawn with the lash shall be paid by another drawn with the sword, as was said three thousand years ago, so still it must be said, that the judgments of the Lord are true and righteous altogether.²⁸

Lincoln drew upon the aesthetic resources of language in a traditional way to make his speech more aesthetically appealing and thus more moving and memorable. In some cases, however, a source may decide intentionally to offend traditional aesthetic expectations to achieve greater persuasive impact. In the following passage, for example, Malcolm X answers some of the arguments of Rev. Martin Luther King, Jr. with provocative language that violates traditional conventions:

This is a real revolution. Revolution is always based on land. Revolution is never based on begging somebody for an integrated cup of coffee. Revolutions are never based on love-your-enemy and pray-for-those-who-spitefully-use-you. And revolutions are never waged singing "We Shall Overcome." Revolutions are based on bloodshed.²⁹

Malcolm X, like Abraham Lincoln, employs allusion, consonance, repetition, and other aesthetic devices to enhance his discourse and to make it more vivid, moving, and memorable. Though Malcolm X employs the aesthetic resources of language, it would not be quite accurate to say that his goal has been to make his speech more beautiful or pleasant to listen to. Rather, his goal is to shock his audience out of complacency, and to get them to reject one suggested course of action and to accept a different one.

The aesthetic dimension of rhetoric has always been important to the art. In the next chapter, we will see that one of the early Sophists, Gorgias, believed that the sounds of words, when manipulated with skill, could captivate audiences. The persuasive potential in the aesthetic resources of language is a persistent theme in rhetorical history.

Arguments, appeals, arrangement, and aesthetics each remind us that rhetoric is not only persuasive but also carefully planned discourse. Over its history, the art of rhetoric has developed around the realization that various resources available in symbol systems allow skilled practitioners to achieve various desired effects, including persuasion, clarity, beauty of expression, and capturing an audience's attention.

Rhetoric Addresses Contingent Issues

In an attempt to define the study of rhetoric, the Greek philosopher Aristotle (384–322 BCE) wrote that "it is the duty of rhetoric to deal with such matters as we deliberate upon without arts or systems to guide us" and when "the subjects of our deliberation are such as seem to present us with alternative possibilities." He added,

"About things that could not have been, and cannot be, other than they are, nobody who takes them to be of this nature wastes his time in delineation." ³⁰

Aristotle apparently thought that rhetoric comes into play when we are faced with practical questions about matters that confront everyone, and about which there are no definite and unavoidable answers. Such contingent questions require deliberation or the weighing of options, not proof of the type mathematicians might use. Rhetoric assists that process of weighing options when the issues facing us are contingent.

To deliberate is to reason through alternatives, and Aristotle says no one does this when things cannot be "other than they are." Rhetorical theorist Thomas Farrell (1947–2006) put the point this way: "It makes no sense to deliberate over things which are going to be the case anyway or things which could never be the case." So, the art of rhetoric would not address a question such as whether the sun will rise tomorrow morning, nor one such as whether France should be made the fifty-first American state. The one is an inevitable fact (it is "going to be the case anyway"), the other a virtual impossibility (it "could never be the case").

Rhetorical theorist Lloyd Bitzer (1931–2016), quoting the nineteenth-century writer Thomas De Quincey (1785–1859), has this to say about contingency: "Rhetoric deals mainly with matters which lie in that vast field 'where there is no *pro* and *con*, with the chance of right and wrong, true and false, distributed in varying proportions among them." Bitzer adds, "[R]hetoric applies to contingent and probable matters which are subjects of actual or possible disagreement by serious people, and which permit alternative beliefs, values, and positions." 32

Rhetoric addresses unresolved issues that do not dictate a particular outcome, and in the process it engages our value commitments. Thus, according to Farrell, Aristotle treated "the very best audiences as a kind of extension of self, capable of weighing the merits of practical alternatives." As individuals, we face many of the same kinds of issues, practical and moral questions that demand decisions or judgments. Of course, similar questions face us as members of the larger public. Is a just war possible? What subjects should be taught in our schools? How can health care be equitably distributed? When there are alternatives to be weighed and matters are neither inevitable nor impossible, we are facing contingent issues that invite the use of rhetoric.

We can shift our focus just a bit at this point and consider the practical results achieved by the art of rhetoric in democratic societies. We will see that when the art of rhetoric is taken seriously, studied carefully, and practiced well, it performs various vital social functions.

SOCIAL FUNCTIONS OF THE ART OF RHETORIC

We began this chapter by noting some unpleasant associations the art of rhetoric has carried with it through its history. But, though rhetoric can be used for wrong ends such as deception, it also plays many important social roles. Rhetoric's misuse is more likely when the art is available only to an elite, when it is poorly understood by audiences, or when it is unethically practiced. The six functions of rhetoric I will highlight are the following: (1) ideas are tested, (2) advocacy is assisted, (3) power is distributed, (4) facts are discovered, (5) knowledge is shaped, and (6) communities are built.

Rhetoric Tests Ideas

An Overview of Rhetoric

One of rhetoric's most important functions is that it allows ideas to be tested on their merits. The practice of rhetoric can provide a peaceful means for evaluating ideas publicly. To win acceptance for a concept I have to advocate it, and effective advocacy means thinking and acting rhetorically. That is to say, advocacy calls on one's knowledge of rhetoric. Testing ideas begins as I come up with my arguments (invention) and shape them into a structured message (arrangement), and it continues as an audience responds to my presentation.

The audience is a vital element in rhetoric's capacity to test ideas. In seeking an audience's consent, we recognize that the audience members will exercise critical judgment. Some audiences test ideas carefully while others are careless about this responsibility. The better equipped an audience is to test ideas advanced for their consideration, and the more care that goes into that testing, the greater assurance we have of the quality of ideas. This testing of ideas in public settings constitutes a distinct benefit to society. Thus, training in the art of rhetoric is just as important for audience members as it is for advocates.

The responses of both friendly critics and opponents help me strengthen my arguments and refine my ideas. Adapting to critical responses makes my case clearer, stronger, more moving, and more persuasive. The process of testing and refining ideas is tied directly to understanding and practicing the art of rhetoric.

What goes in to testing ideas rhetorically? To critically examine an idea means answering questions such as the following:

- —Do I trust the rhetor advocating the idea?
- —Is the idea clear or obscure?
- —Are the arguments supporting the idea convincing?
- —Is the evidence advanced in the idea's support recent and from reliable sources?
- —Have unnecessary appeals been employed to distract attention from faulty arguments?
- —Are contradictions present in the advocate's case?

Just as advocates rely on rhetorical resources, each of these questions also finds its answer in some dimension of the art of rhetoric. This means that audiences must also be rhetorically astute if the idea-testing function of the art of rhetoric is to be robust and trustworthy.

Rhetoric Assists Advocacy

Rhetoric is the method by which we advocate ideas we believe in. Rhetoric gives our private ideas a public voice, thus directing attention to them. Recall that Richard Lanham defines rhetoric as the study of "how attention is created and allocated."³⁴ For this reason, he speaks of rhetoric as teaching "the economics of attention."³⁵

Politics comes to mind as an activity requiring advocacy; political speeches, debates, and campaign ads promote ideas and candidates. Rhetoric is employed in preparing such messages. The same is true when lobbyists make their case to legislators, when constituents write letters to their representatives, and when committees debate the merits of a proposal. Similarly, in the arena of law, the art of rhetoric

helps attorneys prepare their clients' cases. Courtroom pleading itself has involved rhetorical skill since courts first appeared in the ancient world, and advocates in newer legal arenas such as environmental law also turn to rhetoric.

Advocacy in less structured settings often follows the principles taught by the art as well, whether or not advocates have had the benefit of formal education in rhetoric. For instance, when you express an artistic judgment to a friend—say, that Spike Lee's films are better than those of Steven Spielberg—you advance your reasons guided by some sense—trained or intuitive—of how to present ideas effectively.

The same holds true for a media project prepared for a course on documentary production. In a 20-minute video presenting interviews with breast cancer patients, a student builds a case for increased funding for research. The video will be shown not only to her class but also to funding agencies. Editorial decisions are made guided by principles such as the following: Which portions of the interviews will be used? Which interviews will come first and last? Will the interviewer herself play a prominent role in the video or will she remain in the background? Such judgments are made with some sense of how an effective case is constructed in the medium of video, within a limited amount of time, and before particular audiences.

Whether in formal contexts such as a courtroom or a less structured setting such as a conversation, the art of rhetoric is crucial to effective advocacy. Rhetoric is the study of effective advocacy; it provides a voice for ideas, thus drawing attention to them. This important function of rhetoric may easily be overlooked, but any time an idea moves from private belief to public statement the art of rhetoric is employed.

Understanding the art of rhetoric enhances one's skill in advocacy. We may at times wish that some persons or groups did *not* understand rhetoric, because we disagree with their aims or find their ideas repugnant. The solution to this problem would appear to be an improved understanding of rhetoric on our part. When we disagree with a point of view, rhetoric helps us to prepare an answer, to advance a counterargument. This brings us to the third benefit of the art of rhetoric, its capacity to distribute power.

Rhetoric Distributes Power

Our discussion of rhetoric's role in advocacy raises the closely related issue of rhetoric and power. Due to its capacity for influencing decisions, rhetoric is a form of social power. When we think of rhetoric and power, certain questions come to mind:

- —Who is allowed to speak in a society?
- —On what topics are we permitted to speak?
- —In which settings is speech allowed?
- —What kind of language is it permissible to employ?
- —Which media are available to which advocates, and why?

Talk Is Action

The answers these questions receive have a lot to do with the distribution of power or influence. Issues of power and its distribution have always been central to rhetorical theory. James Berlin writes, "Those who construct rhetorics ... are first and foremost concerned with addressing the play of power in their own day." Berlin is asserting,

then, that even the guidelines one sets out as normative for writing and speaking are influenced by, perhaps developed in the service of, existing power structures.

An Overview of Rhetoric

When we contrast talk to action in statements like, "Let's stop talking and do something," we may be misleading ourselves regarding language's great power to shape our thinking and thus our actions. Rhetorical theorists have long recognized that language and power are intimately connected, and that power involves more than physical force or monetary resources. Speaking and writing are forms of action, and thus rhetoric might be understood as the study of how symbols are used effectively as a source of power. We can identify three types of power with which rhetoric is closely associated.

Personal Power

First, rhetoric contributes to *personal* power. The art provides an avenue to success and advancement by sharpening our expressive skills. Seminars in effective speaking, writing, and even in vocabulary building suggest that the relationship between personal success and language is widely acknowledged. Human resources specialist Rebecca R. Hastings has written, "To be successful, young workers need to develop a lot more than job-specific knowledge, experts say. Of the so-called soft skills needed for success in the workplace, communication skills are particularly critical." Clear, effective, and persuasive expression is not simply a matter of demonstrating your sophistication; it is an important means of advancing toward the goals you have set for yourself.

Psychological Power

Second, rhetoric is a source of *psychological* power, that is, the power to shape thought. Symbols and thoughts are intricately connected; we may change the way people think simply by altering their symbolic framework through a skillfully crafted message. In addition to its capacity to affect action, rhetoric is a means by which one person alters the psychological world of another. Indeed, symbols are perhaps our only avenue into the mental world.

Advertising provides an example of rhetoric's psychological power. Through the strategic use of symbols, advertisers seek to shape our psychological frame and thus our behavior. The repeated symbolic association in advertising between a very thin body and personal attractiveness has led many individuals to become dissatisfied with their appearance. This alteration in one's psychological world can have harmful consequences when it begins to affect a behavior such as eating. For this reason, rhetoric's power to alter the mental world of an audience must be approached with great care.

Political Power

Third, rhetoric is a source of *political* power. The distribution of political influence is often a matter of who gets to speak, where they are allowed to speak, and on what subjects. As we shall see in Chapter 11, French philosopher Michel Foucault explored this intersection of rhetoric and political power in a society. He suggested that power is not a fixed, hierarchical social arrangement, but rather a fluid concept closely connected to the symbolic strategies that hold sway at any particular time. In other words, political power is, for Foucault, directly related to the practice of rhetoric.

Some groups have a greater opportunity to be heard than do others, a fact that raises a concern for the "privileging" of some perspectives or ideologies. An ideology is a system of belief, or a framework for interpreting the world.³⁸ An unexamined

ideology may prevent its adherents from seeing things "as they are." Thus, we need to be wary of rhetoric's use to concentrate as well as to distribute power.³⁹ When rhetoric is employed to advocate ideas, but its capacity to test ideas is subverted, the reign of unexamined ideology becomes a real possibility.

Rhetoric Discovers Facts

Rhetoric tests ideas, assists advocacy, and distributes power. A fourth important function of rhetoric is that it helps us to discover facts and truths crucial to decision-making. Rhetoric assists this important task in at least three ways.

First, in order to prepare a case, you must locate evidence to support your ideas. This investigative process is an integral part of the art of rhetoric. Though we may have strong convictions, if we are to convince an audience to agree with us, these convictions have to be supported with evidence and arguments. Solid evidence allows better decisions on contingent matters. Second, crafting a message involves evaluating the available facts. This process of invention often suggests new ways of understanding facts and new relationships among facts. Third, the clash of arguments brings new facts to light and refines available ones.

Audiences expect advocates to be well informed. As an advocate you become a source of information crucial to decision-making. But your audience, which may include opponents, will also be evaluating the evidence you present. Some facts may be misleading, outdated, irrelevant, or not convincing. Thus, the art of rhetoric assists not just the discovery of new facts, but also an interactive process of determining which facts are actually relevant and convincing. Of course, rhetoric might also be employed to conceal facts, which reminds us again that rhetoric always raises ethical concerns. As we shall see in Chapter 2, the realization that rhetoric assists the discovery of facts is an ancient one, as is awareness that it might also obscure facts.

Rhetoric Shapes Knowledge

How do we come to agreements about what we know or value? How does a particular view of justice come to prevail in one community or culture? How does a value for equality under the law become established? How do we know that equality is better than inequality? Though the answer to any one of these questions is complex, an important connection exists between knowledge and rhetorical practices.

Rhetoric often plays a critical social role in determining what we accept as true, right, or probable. For this reason, rhetoric scholar Robert Scott (1928–2018) referred to rhetoric as "epistemic," that is, knowledge-building. What did he mean? Through rhetorical interaction, we come to accept some ideas as true and to reject others as false. Rhetoric's knowledge-building function derives from its tendency to test ideas. Once an idea has been thoroughly tested by a community, it becomes part of what is accepted as known. Of course, this acceptance as knowledge may be temporary; further rhetorical interactions may call into question what is currently accepted as known.

How Do We "Know"?

That knowledge develops rhetorically runs counter to our usual understanding of the sources of knowledge. We often think that knowledge comes through our direct experience, or through the indirect experience that we call education. Knowledge is

treated as an object to be discovered in the same way as an astronomer discovers a new star: The star was always out there, and the astronomer just happened to see it. Some knowledge fits this objective description better than other knowledge does.

Perhaps rhetoric plays a limited role in establishing this sort of knowledge. But, the star's age is less certain than is its existence, and may require argument among scientists to ascertain. Rhetoric now begins to play a role in establishing knowledge, for the scientists involved in the debate will likely draw on what they know of the art to persuade their peers. They will assess their audience, craft arguments they think will be persuasive, avoid ones that are less persuasive, arrange their arguments in an effective order, and provide evidence to support their claims. And that is not the end of the process—even if the majority of scientists do reach agreement, these same scientists may find themselves adapting their arguments to a new audience of nonspecialists, taking into consideration a new set of audience demands. The question of what we say we know will still be important. Knowledge about the universe's age has religious significance for many people. Do we know that the star's age should be taught in schools? Do we know that money should be invested in trying to launch a telescope to get a better look at the new star? Rhetorical interactions are involved in resolving these questions as well, and the way rhetoric is practiced is important in determining what finally is accepted as knowledge.

Rhetoric Builds Community

What defines a community? One answer to this question is that what people value, know, or believe in common defines a community. Some observers fear that Americans may be losing their sense of constituting a community in the face of growing pressures toward fragmentation. If this is the case, and if preserving a sense of community is a goal worth striving for, what can be done about this problem of social fragmentation?

Many of the processes by which we come to hold beliefs and values in common are rhetorical in nature. Michael J. Hogan, a scholar who has studied the relationship between rhetoric and community, writes that "rhetoric shapes the character and health of communities in countless ways" Many writers who have sought to understand the ways in which communities form have concluded that "communities are largely defined, and rendered healthy or dysfunctional, by the language they use to characterize themselves and others." If this is indeed the case, as Hogan and others have suggested, then it is important to explore the specific function played by rhetoric in building—or perhaps in undermining—communities.

Communities are not simply geographical entities bounded by borders or contained in particular districts of a city. Communities are made up of people who find common cause with one another, who see the world in a similar way, who have similar concerns and aspirations. Thus, a religious organization, a group of employees, and members of an ethnic group living in the same city might constitute communities. Not every aspect of such communities results from the practice of rhetoric. For example, ethnicity is not a function of discourse. But developing common values, common aspirations, and common beliefs very often is a result of what is said, by whom, and with what effect.

Consider, for example, the community that developed around the civil rights advocacy of Dr. Martin Luther King, Jr. in the 1950s and 1960s. Dr. King was a highly

skillful and knowledgeable practitioner of the art of rhetoric. He, and others working with him, created a community of value and action, and much of their work was accomplished by means of effective rhetorical discourse. More specifically, Dr. King advocated certain values in a persuasive manner. Among these were equality, justice, non-violence, and peace. He also tested particular ideas in public settings—ideas like racism, which he rejected, and ideas like unity among races, which he embraced. He brought facts to light for his audiences, such as facts about the treatment of African American people.

Dr. King provided a language for talking about racial harmony. His dream of a racially unified America and his advocacy of "nonviolent resistance" inspired many in the civil rights movement who made his terminology part of their own vocabulary. Through his rhetorical efforts, King built a community of discourse that enabled people to think and act with unity. He developed an active and effective community around powerful ideas to which he gave voice rhetorically.

Often members of a community—examples might include feminists, Orthodox Jews, or animal rights activists—do not know all of the other members of their community personally. In fact, any particular member of a large and diffuse community might know only a very small fraction of the people who would say they belong to the group. How is a sense of community maintained when a community is large and geographically diffuse? Certainly, the group's symbols, metaphors, and ways of reasoning function to create a common bond that promotes a strong sense of community despite physical separation. Moreover, communities are sustained over time by the rhetorical interactions of their members with one another and with members of other groups. As Hogan writes, "[C]ommunities are living creatures, nurtured and nourished by rhetorical discourse."

This section has discussed six functions performed by the practice of rhetoric: (1) testing ideas, (2) assisting advocacy, (3) distributing power, (4) discovering facts, (5) shaping knowledge, and (6) building community. These functions are closely related to major themes in the history of rhetoric and provide connections among subsequent chapters. The next section sets out some of these themes in greater detail.

CONCLUSION

We began this chapter by considering some common meanings of the term *rhetoric*, such as empty talk, beautiful language, or persuasion. Whereas these meanings frequently are associated with the term, rhetoric was defined as the study or practice of effective symbolic expression, we noted that rhetoric refers to a type of discourse marked by several characteristics that include being planned, adapted to an audience, and responsive to a set of circumstances. We have also considered some of rhetoric's social functions such as testing ideas, assisting advocacy, and building communities.

Recurrent Themes

Several important issues arise when we begin to think seriously about the art of rhetoric and its various uses. We will return to these themes as we consider the ways in which the art of rhetoric has developed over the past 2,500 years. The following issues will be revisited throughout this text:

An Overview of Rhetoric

Rhetoric and Power

As we have seen, rhetoric bears an important relationship to power in a society. The art of rhetoric itself brings a measure of power, and rhetorical practices play an important role in both distributing and concentrating power. Every culture makes decisions about who may speak, before which audiences, and on which topics. Altering these limitations will often mean violating such established norms, whether through the practice of rhetoric, rhetorical education, or both. If a segment of a society lacks the knowledge of rhetoric, or is denied the ability to practice rhetoric, does this mean that their access to power is correspondingly diminished? We will examine this question at several junctures in the history of rhetoric.

Rhetoric and Truth

Rhetoric discovers facts relevant to decision-making. Moreover, rhetoric helps to shape what we say we know or believe. What, then, is rhetoric's relationship to truth? Does rhetoric discover truth? Or, does rhetoric simply provide one the means of communicating truth discovered by other approaches, for instance, the scientific method? As we explore the history of rhetoric, we will uncover various answers to these questions. If truth is transcendent, rhetoric's role in its discovery or creation may be minimal. In fact, rhetoric might even be a threat to truth. If, on the other hand, truth is a matter of social agreements, rhetoric plays a major role in establishing what is true.

Rhetoric and Ethics

Persuasion is central to rhetoric. This means that rhetoric always raises moral or ethical questions. If persuasion is always wrong, then rhetoric shares this moral condemnation. If persuasion is acceptable, it is important to ask about ethical obligations of a speaker, writer, or artist. What are the moral restraints within which rhetoric ought to be practiced? Few people would want to live in a society in which rhetoric is practiced without any regard for ethical responsibility on the part of advocates.

Rhetoric and the Audience

The question of ethics is inseparable from the question of a rhetor's potential influence on an audience because rhetoric is a form of power, and ethical considerations attend rhetoric. How does rhetoric alter an audience's ways of thinking or prompt action on their part? Moreover, if audiences have some control over the quality of rhetoric, are we morally obliged to educate audiences about rhetoric? As we explore the history of rhetoric, the audience will often be a central concern.

Rhetoric and Society

Our discussion in this chapter has also raised the larger issue of rhetoric's role in developing and maintaining communities and societies. We have considered rhetoric's specific social functions. We depend on rhetoric to forge the compromises and achieve the cooperation needed to live and work together. Such functions are crucial to flourishing democracies. As we survey the history of rhetoric we will want to pay attention to the ways in which rhetoric shapes the values that provide societies

a corporate identity and a common direction. How is it that the skillful practice of rhetoric benefits a society, not just practically but morally as well?

These themes and questions will attend our discussion of rhetoric's history. The different answers to our questions suggested by a wide range of writers, and the reasons for their answers, make the history of rhetoric a rich and intriguing source of insight into the development of human thought, relationships, and culture. In Chapter 2, we will encounter most of these themes as we begin our study of rhetoric's long and rich history by looking at its controversial origins and early development in ancient Greece.

QUESTIONS FOR REVIEW

- 1. How are the following terms defined in the chapter?
 - rhetoric
 - the art of rhetoric
 - rhetorical discourse
 - rhetor
 - symbol
 - · rhetorical theory
 - the Argumentative Hypothesis
- 2. What are the marks or characteristics of rhetorical *discourse* discussed in this chapter?
- 3. Which specific resources of language are discussed under the heading "Rhetoric Is Planned"?
- **4.** What *social functions* of the art of rhetoric are discussed in this chapter?
- 5. Which three types of power are enhanced by an understanding of the art of rhetoric?
- **6.** What is meant by the statement that rhetoric addresses contingent issues?

QUESTIONS FOR DISCUSSION

1. The following artifacts, Abraham Lincoln's "Second Inaugural Address" and Emily Dickinson's poem "Success Is Counted Sweetest," were written at about the same time, and each is written with reference to the Civil War. The two pieces are often held to represent two different types of discourse: Lincoln's address is categorized as rhetoric, while Dickinson's work fits best into the category of poetry. Thinking back on the characteristics of rhetorical discourse discussed in this chapter, what case could be made, if any, for distinguishing Lincoln's work from Dickinson's? Do they belong to different literary categories? Refer back to the resources of language—argument, appeal, arrangement, and artistic devices—in thinking about these two pieces. Does each employ all four resources?

"Second Inaugural Address"

Abraham Lincoln

could be presented.

Fellow-countrymen: At this second appearing to take the oath of the presidential office, there is less occasion for an extended address than there was at first. Then a statement, somewhat in detail, of a course to be pursued seemed very fitting and proper. Now, at the expiration of 4 years, during which public declarations have been constantly called forth on every point and phase of the great contest which still

The progress of our arms, upon which all else chiefly depends, is as well known to the public as to myself, and it is, I trust, reasonably satisfactory and encouraging to all. With high hope for the future, no prediction in regard to it is ventured.

absorbs the attention and engrosses the energies of the nation, little that is new

On the occasion corresponding to 4 years ago, all thoughts were anxiously directed to an impending civil war. All dreaded it, all sought to avoid it. While the inaugural address was being delivered from this place, devoted altogether to saving the Union without war, insurgent agents were in the city seeking to destroy it with war seeking to dissolve the Union and divide the effects by negotiation. Both parties deprecated war, but one of them would make war rather than let it perish, and the war came. One-eighth of the whole population were colored slaves, not distributed generally over the Union, but localized in the Southern part of it. These slaves constituted a peculiar and powerful interest. All knew that this interest was somehow the cause of the war. To strengthen, perpetuate, and extend this interest was the object for which the insurgents would rend the Union by war, while the government claimed no right to do more than to restrict the territorial enlargement of it.

Neither party expected for the war nor the magnitude or the duration which it has already attained. Neither anticipated that the cause of the conflict might cease when, or even before the conflict itself should cease. Each looked for an easier triumph, and a result less fundamental and astounding. Both read the same Bible and pray to the same God, and each invokes His aid against the other. It may seem strange that any men should dare to ask a just God's assistance in wringing their bread from the sweat of other men's faces, but let us judge not that we be not judged. The prayer of both could not be answered. That of neither has been answered fully. The Almighty has His own purposes. Woe unto the world because of offenses, for it must needs be that offenses come, but woe to that man by whom the offence cometh. If we shall suppose that American slavery is one of those offenses which, in the providence of God, must needs come, but which having continued through His appointed time, He now wills to remove, and that He gives to both North and South this terrible war as the woe due to those by whom the offence came, shall we discern there any departure from those divine attributes which the believers in a living God always ascribe to Him? Fondly do we hope, fervently do we pray, that this mighty scourge of war may speedily pass away. Yet if God wills that it continue until all the wealth piled by the bondsman's two 250 years of unrequited toil shall be sunk, and until every drop of blood drawn with the lash shall be paid by another drawn with the sword, as was said 3,000 years ago, so still it must be said, that the judgments of the Lord are true and righteous altogether.

With malice toward none, with charity for all, with firmness in the right as God gives us to see the right, let us finish the work we are in, to bind up the nation's wounds, to care for him who shall have borne the battle, and for his widow and his

An Overview of Rhetoric

orphans, to do all which may achieve and cherish a just and a lasting peace among ourselves and with all nations. 43

"Success Is Counted Sweetest"

Emily Dickinson

Success is counted sweetest
By those who ne'er succeed.
To comprehend a nectar
Requires sorest need.
Not one of all the purple host
Who took the flag to-day
Can tell the definition,
So clear, of victory,
As he, defeated, dying,
On whose forbidden ear
The distant strains of triumph Break, agonized, and clear.

44

- 2. If rhetoric accomplishes the benefits and performs the functions discussed in this chapter, it might follow that rhetorical training should be a central component in education. Has training in rhetoric or some related discipline been part of your educational experience? Should education focus more on the skills that make up the art of rhetoric?
- **3.** Is rhetoric pervasive in private and social life, as the chapter suggests? In what realms of life, if any, does rhetoric appear to have little or no part to play? Where is its influence greatest, in your estimation? Where is it present, but hidden?
- **4.** Steven Mailloux has written that there are "oral, visual, written, digital, gestural" rhetorics. Which other types of rhetoric would you add to this list? What special types or genres would you include under the types you have added?
- 5. Respond to the claim that rhetoric is important to the process of building community. Has it been your experience, when people come together to form a community, that ways of speaking and reasoning in common are an important part of that process? Could a greater understanding of the art of rhetoric enhance this process of building a community?
- **6.** Some people have criticized rhetoric for being manipulative. Do you believe that rhetoric is, by its very nature, manipulative? If not, what ethical guidelines might be important for constraining the practice of rhetoric so that it does not become a tool for manipulation?
- 7. The following speech was delivered by Civil Rights activist Mrs. Fannie Lou Hamer on August 22, 1964 to the Credentials Committee of the Democratic National Convention in Atlantic City, New Jersey. Though she lacked formal rhetorical training, the speech stands as a powerful example of morally informed oratory. Drawing on this chapter's discussion of the social functions of argument, write a one-page analysis of how Mrs. Hamer's testimony before the committee illustrates any two of those functions.

Mr. Chairman, and to the Credentials Committee, my name is Mrs. Fannie Lou Hamer, and I live at 626 East Lafayette Street, Ruleville, Mississippi, Sunflower County, the home of Senator James O. Eastland and Senator Stennis.

It was the 31st of August in 1962 that eighteen of us traveled twenty-six miles to the county courthouse in Indianola to try to register to become first-class citizens. We was met in Indianola by policemen, Highway Patrolmen, and they only allowed two of us in to take the literacy test at the time. After we had taken this test and started back to Ruleville, we was held up by the City Police and the State Highway Patrolmen and carried back to Indianola where the bus driver was charged that day with driving a bus the wrong color.

After we paid the fine among us, we continued on to Ruleville, and Reverend Jeff Sunny carried me four miles in the rural area where I had worked as a timekeeper and sharecropper for eighteen years. I was met there by my children, who told me the plantation owner was angry because I had gone down—tried to register.

After they told me, my husband came, and said the plantation owner was raising Cain because I had tried to register. And before he quit talking the plantation owner came and said, "Fannie Lou, do you know—did Pap tell you what I said?"

And I said, "Yes, sir." He said, "Well I mean that." Said, "If you don't go down and withdraw your registration, you will have to leave." Said, "Then if you go down and withdraw," said, "you still might have to go because we're not ready for that in Mississippi." And I addressed him and told him and said, "I didn't try to register for you. I tried to register for myself."

I had to leave that same night. On the 10th of September 1962, sixteen bullets were fired into the home of Mr. and Mrs. Robert Tucker for me. That same night two girls were shot in Ruleville, Mississippi. Also, Mr. Joe McDonald's house was shot in.

And June the 9th, 1963, I had attended a voter registration workshop; was returning back to Mississippi. Ten of us was traveling by the Continental Trailway bus. When we got to Winona, Mississippi, which is Montgomery County, four of the people got off to use the washroom, and two of the people—to use the restaurant—two of the people wanted to use the washroom.

The four people that had gone in to use the restaurant was ordered out. During this time I was on the bus. But when I looked through the window and saw they had rushed out I got off of the bus to see what had happened. And one of the ladies said, "It was a State Highway Patrolman and a Chief of Police ordered us out." I got back on the bus and one of the persons had used the washroom got back on the bus, too.

As soon as I was seated on the bus, I saw when they began to get the five people in a highway patrolman's car. I stepped off of the bus to see what was happening and somebody screamed from the car that the five workers was in and said, "Get that one there." And when I went to get in the car, when the man told me I was under arrest, he kicked me.

I was carried to the county jail and put in the booking room. They left some of the people in the booking room and began to place us in cells. I was placed in a cell with a young woman called Miss Ivesta Simpson. After I was placed in the cell I began to hear sounds of licks and screams. I could hear the sounds of licks and horrible screams. And I could hear somebody say, "Can you say, 'yes, sir,' nigger? Can you say 'yes, sir'?"

And they would say other horrible names. She would say, "Yes, I can say 'yes, sir." "So, well, say it." She said, "I don't know you well enough." They beat her,

I don't know how long. And after a while she began to pray, and asked God to have mercy on those people.

And it wasn't too long before three white men came to my cell. One of these men was a State Highway Patrolman and he asked me where I was from. And I told him Ruleville. He said, "We are going to check this." And they left my cell and it wasn't too long before they came back. He said, "You are from Ruleville all right," and he used a curse word. And he said, "We're going to make you wish you was dead."

I was carried out of that cell into another cell where they had two Negro prisoners. The State Highway Patrolmen ordered the first Negro to take the blackjack. The first Negro prisoner ordered me, by orders from the State Highway Patrolman, for me to lay down on a bunk bed on my face. And I laid on my face, the first Negro began to beat me.

And I was beat by the first Negro until he was exhausted. I was holding my hands behind me at that time on my left side, because I suffered from polio when I was six years old. After the first Negro had beat until he was exhausted, the State Highway Patrolman ordered the second Negro to take the blackjack.

The second Negro began to beat and I began to work my feet, and the State Highway Patrolman ordered the first Negro who had beat to sit on my feet—to keep me from working my feet. I began to scream and one white man got up and began to beat me in my head and tell me to hush.

One white man—my dress had worked up high—he walked over and pulled my dress—I pulled my dress down and he pulled my dress back up. I was in jail when Medgar Evers was murdered.

All of this is on account of we want to register, to become first-class citizens. And if the Freedom Democratic Party is not seated now, I question America. Is this America, the land of the free and the home of the brave, where we have to sleep with our telephones off of the hooks because our lives be threatened daily, because we want to live as decent human beings, in America?

Thank you.⁴⁵

TERMS

Aesthetics Study of the persuasive potential in the form, beauty, or force of symbolic expression.

Appeals Symbolic methods that aim either to elicit an emotion or to engage the audience's loyalties or commitments.

Argument Discourse characterized by reasons advanced to support a conclusion. Reasoning made public with the goal of influencing an audience.

Argumentative Hypothesis The view that the entire purpose of our public reasoning—of making arguments—is to demonstrate to *others* that we have support for our views.

Arrangement The planned ordering of a message to achieve the greatest persuasive effect.

Dispositio Arrangement; Cicero's term for the effective ordering of arguments and appeals.

Elocutio Style; Cicero's term to designate the concern for finding the appropriate language or style for a message.

Ideology A system of belief, or a framework for interpreting the world.

Inventio Invention; Cicero's term describing the process of coming up with the arguments and appeals that would make up the substance of a persuasive case.

An Overview of Rhetoric

Motives Commitments, goals, desires, or purposes when they lead to action.

Rhetor Anyone engaged in preparing or presenting rhetorical discourse.

Rhetoric, **Art of** The study and practice of effective symbolic expression.

Rhetoric, Type of discourse Goal-oriented discourse that seeks, by means of the resources of symbols, to adapt ideas to an audience.

Rhetorical discourse Language crafted according to the principles of the art of rhetoric.

Rhetorical theory The systematic presentation of rhetoric's principles, descriptions of its various functions, and explanations of how rhetoric achieves its goals.

Symbol Any mark, sign, sound, or gesture that represents something based on social agreement.

NOTES

- 1 Plato, Gorgias, 463; trans. W. C. Helmbold (Indianapolis, IN: Bobbs-Merrill, 1952), 23–24.
- 2 John Locke, *An Essay Concerning Human Understanding* (1690; rpt. Oxford: Clarendon Press, 1894), 146.
- 3 Sander Gilman, Carole Blair, and David Parent, eds./trans. *Friedrich Nietzsche on Rhet-oric and Language* (New York: Oxford University Press, 1989), 21. (Emphasis in original.)
- 4 Wayne Booth, *The Vocation of a Teacher* (Chicago, IL: University of Chicago Press, 1988), xiv–xv.
- 5 Richard McKeon, *Rhetoric: Essays in Invention and Discovery*, ed. Mark Backman (Woodbridge, CT: Ox Bow Press, 1987), 108.
- 6 Richard Lanham, *The Electronic Word: Democracy, Technology, and the Arts* (Chicago, IL: University of Chicago Press, 1993), 51.
- 7 Stanford News Service, "The New Literacy: Study Finds Richness and Complexity in Students' Writing" (October 12, 2009), http://news.stanford.edu/pr/2009/pr-lunsford-writing-101209.html. Accessed May 2, 2011.
- 8 Laurie Gries, Still Life with Rhetoric: A New Materialist Approach for Visual Rhetorics (Utah State University Press, 2015), 11.
- 9 Deirdre McCloskey, "The Neglected Economics of Talk," *Planning for Higher Education* 22 (Summer 1994): 11–16, 14.
- 10 Deirdre McCloskey, "The Rhetoric of the Economy and the Polity, Whether or Not 'In Crisis," *Prudentia* (December, 2010), www.deirdremccloskey.com/articles/polity.php.
- 11 Michael Billig, *Arguing and Thinking: A Rhetorical Approach to Social Psychology* (1989; Cambridge University Press, 1996), 57.
- 12 For a scholarly yet entertaining look at the ways we go about persuading one another in everyday life, see: Robert Cialdini's insightful book, *Influence: The Psychology of Persuasion* (1984; New York: William Morrow, 1993).
- 13 Jane Donawerth, ed., *Rhetorical Theory by Women before 1900* (Lanham, MD: Rowan & Littlefield, 2002), xiv.
- 14 George Kennedy, translator's introduction to *Aristotle on Rhetoric: A Theory of Civic Discourse* (Oxford: Oxford University Press, 1991), 7.
- 15 Carole Blair has written an intriguing essay on the rhetoric of the Vietnam Memorial, which appears in the book *Critical Questions* (New York: St. Martin's Press, 1994). Barry

- Brummett considers the rhetoric of a wide variety of cultural artifacts in *Rhetoric in Popular Culture* (New York: St. Martin's Press, 1994).
- 16 "Ocean swimmer Diana Nyad, 60, Preparing for Cuba-to-Florida Swim," Los Angeles Times (July 27, 2010), https://latimesblogs.latimes.com/outposts/2010/07/ocean-swimmer-diana-nyad-60-preparing-for-cubatoflorida-swim.html.
- 17 If you would be interested in seeing an extended treatment of this question of the rhetoric of typeface, watch the movie *Helvetica*, a documentary devoted entirely to the history and interpretation of the titular typeface.
- 18 Steven Mailloux, "One Size Doesn't Fit All: The Contingent Universality of Rhetoric," in *Sizing Up Rhetoric*, ed. David Zarefsky and Elizabeth Benacka (Long Grove, IL: Waveland Press, 2008), 7–19, 9.
- 19 Booth, xiv.
- 20 Chaim Perelman and Lucy Olbrechts-Tyteca, The New Rhetoric: A Treatise on Argumentation, trans. John Wilkinson and Purcell Weaver (Notre Dame: University of Notre Dame Press, 1969), 23–24.
- 21 Richard Lanham, *The Economics of Attention: Style and Substance in the Age of Information* (Chicago, IL: University of Chicago Press, 2007).
- 22 "Utah researchers discover how brain is wired for attention," KurzweilAI.net (November 2, 2010), www.kurzweilai.net. Accessed November 2, 2010. The original study appeared November 1, 2010 in online edition of the *Proceedings of the National Academy of Sciences*, www.pnas.org/.
- 23 Leon Watson, "Humans have shorter attention span than goldfish, thanks to smartphones," *The Telegraph* (May 15, 2015), www.telegraph.co.uk/science/2016/03/12/humans-have-shorter-attention-span-than-goldfish-thanks-to-smart/. Accessed August 31, 2016.
- 24 Brian Boyd, On the Origin of Stories (Cambridge, MA: Harvard University Press, 2009), 232.
- 25 Joseph Wenzel, "Three Perspectives on Argument," in *Perspectives on Argumentation: Essays in Honor of Wayne Brockriede*, ed. Robert Trapp and Janice Schuetz (Prospect Heights, IL: Waveland, 1990), 13.
- 26 Hugo Mercier and Dan Sperber, "How do Humans Reason? Arguments for an Argumentative Theory", *Behavioral and Brain Sciences* 34 (2011): 57–111, 58.
- 27 Tom Strini, "A Taut Take on Beethoven's Ninth," *The Milwaukee Journal Sentinel* (May 13, 2006).
- 28 Abraham Lincoln, "Second Inaugural Address," in *The World's Great Speeches*, ed. Lewis Copeland (New York: Dover Publications, 1958), 316–317.
- 29 George Breitman, ed. Malcolm X Speaks (New York: Grove Press, 1966), 50. Quoted in: Robert L. Scott, "Justifying Violence: The Rhetoric of Militant Black Power," in The Rhetoric of Black Power, ed. Robert L. Scott and Wayne Brockriede (New York: Harper and Row, 1969), 132.
- 30 Aristotle, Rhetoric, trans. W. Rhys Roberts (New York: Modern Library, 1954), 27.
- 31 Thomas Farrell, *Norms of Rhetorical Culture* (New Haven, CT: Yale University Press, 1995), 77.
- 32 Lloyd Bitzer, "Political Rhetoric," in *Landmark Essays on Contemporary Rhetoric*, ed. Thomas Farrell (Mahwah, NJ: Hermagoras Press), 1–22, 7.
- 33 Farrell, 79.
- 34 Lanham, *The Electronic Word*, 227.
- 35 Richard Lanham, "The Economics of Attention," *Michigan Quarterly Review* 36 (Spring 1997): 270.
- 36 On the relationship of rhetoric and power, see: James A. Berlin, "Revisionary Histories of Rhetoric: Politics, Power, and Plurality," in *Writing Histories of Rhetoric*, ed. Victor Vitanza (Carbondale, IL: Southern Illinois University Press, 1994), 112–127.

37 Rebecca R., Hastings, "Communication Skills Key for Young Workers", Society for Human Resource Management (March 8, 2012), www.shrm.org/resourcesandtools/hr-topics/employee-relations/pages/communicationskillskey.aspx. Accessed September 1, 2016.

An Overview of Rhetoric

- 38 See: Michael Billig, *Ideology and Opinion* (London: Sage, 1991).
- 39 Billig, Ideology, 5.
- 40 One of the earliest explorations of this issue is found in: Robert L. Scott, "On Viewing Rhetoric as Epistemic," *Central States Speech Journal* 18 (February 1967): 9–16. See also: Lloyd F. Bitzer, "Rhetoric and Public Knowledge," in *Rhetoric, Philosophy, and Literature: An Exploration* (West Lafayette, IN: Purdue University Press, 1978), 67–93.
- 41 See: *Rhetoric and Community: Studies in Unity and Fragmentation*, ed. Michael J. Hogan (Columbia, SC: University of South Carolina Press, 1998), Introduction, xv.
- 42 Hogan, 292.
- 43 Lincoln, 316-317.
- 44 Emily Dickinson, "Success Is Counted Sweetest." Reprinted by permission of the publishers and the Trustees of Amherst College from *The Poems of Emily Dickinson*, ed. Thomas H. Johnson (Cambridge, MA: The Belknap Press of Harvard University, copyright © 1951, 1955, 1979 by the President and Fellows of Harvard College).
- 45 Used with permission of the Fannie Lou Hamer Memorial Garden and Museum Foundation, Charles McLaurin, President of the Board.

Notes

- 1 Plato, *Gorgias*, 463; trans. W. C. Helmbold (Indianapolis, IN: Bobbs-Merrill, 1952), 23–24.
- 2 John Locke, *An Essay Concerning Human Understanding* (1690; rpt. Oxford: Clarendon Press, 1894), 146.
- 3 Sander Gilman, Carole Blair, and David Parent, eds./trans. *Friedrich Nietzsche on Rhetoric and Language* (New York: Oxford University Press, 1989), 21. (Emphasis in original.)
- Wayne Booth, *The Vocation of a Teacher* (Chicago, IL: University of Chicago Press, 1988), xiv–xv.
- 5 Richard McKeon, *Rhetoric: Essays in Invention and Discovery*, ed. Mark Backman (Woodbridge, CT: Ox Bow Press, 1987), 108.
- 6 Richard Lanham, *The Electronic Word: Democracy, Technology, and the Arts* (Chicago, IL: University of Chicago Press, 1993), 51.
- Stanford News Service, "The New Literacy: Study Finds Richness and Complexity in Students' Writing" (October 12, 2009), http://news.stanford.edu/pr/2009/pr-luns ford-writing-101209.html. Accessed May 2, 2011.
- 8 Laurie Gries, Still Life with Rhetoric: A New Materialist Approach for Visual Rhetorics (Utah State University Press, 2015), 11.
- 9 Deirdre McCloskey, "The Neglected Economics of Talk," *Planning for Higher Education* 22 (Summer 1994): 11–16, 14.
- Deirdre McCloskey, "The Rhetoric of the Economy and the Polity, Whether or Not 'In Crisis," *Prudentia* (December, 2010), www.deirdremccloskey.com/articles/polity. php.
- 11 Michael Billig, *Arguing and Thinking: A Rhetorical Approach to Social Psychology* (1989; Cambridge: Cambridge University Press, 1996), 57.
- 12 For a scholarly yet entertaining look at the ways we go about persuading one another in everyday life, see: Robert Cialdini's insightful book, *Influence: The Psychology of Persuasion* (1984; New York: William Morrow, 1993).
- 13 Jane Donawerth, ed., *Rhetorical Theory by Women before 1900* (Lanham, MD: Rowan & Littlefield, 2002), xiv.

- 14 George Kennedy, translator's introduction to Aristotle on Rhetoric: A Theory of Civic Discourse (Oxford: Oxford University Press, 1991), 7.
- 15 Carole Blair has written an intriguing essay on the rhetoric of the Vietnam Memorial, which appears in the book *Critical Questions* (New York: St. Martin's Press, 1994). Barry Brummett considers the rhetoric of a wide variety of cultural artifacts in *Rhetoric in Popular Culture* (New York: St. Martin's Press, 1994).
- 16 "Ocean swimmer Diana Nyad, 60, Preparing for Cuba-to-Florida Swim," *Los Angeles Times* (July 27, 2010), https://latimesblogs.latimes.com/outposts/2010/07/ocean-swimmer-diana-nyad-60-preparing-for-cubatoflorida-swim.html.
- 17 If you would be interested in seeing an extended treatment of this question of the rhetoric of typeface, watch the movie *Helvetica*, a documentary devoted entirely to the history and interpretation of the titular typeface.
- 18 Steven Mailloux, "One Size Doesn't Fit All: The Contingent Universality of Rhetoric," in *Sizing Up Rhetoric*, ed. David Zarefsky and Elizabeth Benacka (Long Grove, IL: Waveland Press, 2008), 7–19, 9.
- 19 Booth, xiv.
- 20 Chaim Perelman and Lucy Olbrechts-Tyteca, The New Rhetoric: A Treatise on Argumentation, trans. John Wilkinson and Purcell Weaver (Notre Dame: University of Notre Dame Press, 1969), 23–24.
- 21 Richard Lanham, *The Economics of Attention: Style and Substance in the Age of Information* (Chicago, IL: University of Chicago Press, 2007).
- 22 "Utah researchers discover how brain is wired for attention," KurzweilAI.net (November 2, 2010), www.kurzweilai.net. Accessed November 2, 2010. The original study appeared November 1, 2010 in online edition of the *Proceedings of the National Academy of Sciences*, www.pnas.org/.
- 23 Leon Watson, "Humans have shorter attention span than goldfish, thanks to smartphones," *The Telegraph* (May 15, 2015), www.telegraph.co.uk/science/2016/03/12/humans-have-shorter-attention-span-than-goldfish-thanks-to-smart/. Accessed August 31, 2016.
- Brian Boyd, *On the Origin of Stories* (Cambridge, MA: Harvard University Press, 2009), 232.
- Joseph Wenzel, "Three Perspectives on Argument," in *Perspectives on Argumentation: Essays in Honor of Wayne Brockriede*, ed. Robert Trapp and Janice Schuetz (Prospect Heights, IL: Waveland, 1990), 13.
- Hugo Mercier and Dan Sperber, "How do Humans Reason? Arguments for an Argumentative Theory", *Behavioral and Brain Sciences* 34 (2011): 57–111, 58.
- Tom Strini, "A Taut Take on Beethoven's Ninth," *The Milwaukee Journal Sentinel* (May 13, 2006).
- Abraham Lincoln, "Second Inaugural Address," in *The World's Great Speeches*, ed. Lewis Copeland (New York: Dover Publications, 1958), 316–317.
- 29 George Breitman, ed. Malcolm X Speaks (New York: Grove Press, 1966), 50. Quoted in: Robert L. Scott, "Justifying Violence: The Rhetoric of Militant Black Power," in The Rhetoric of Black Power, ed. Robert L. Scott and Wayne Brockriede (New York: Harper and Row, 1969), 132.
- 30 Aristotle, Rhetoric, trans. W. Rhys Roberts (New York: Modern Library, 1954), 27.
- Thomas Farrell, *Norms of Rhetorical Culture* (New Haven, CT: Yale University Press, 1995), 77.
- 32 Lloyd Bitzer, "Political Rhetoric," in *Landmark Essays on Contemporary Rhetoric*, ed. Thomas Farrell (Mahwah, NJ: Hermagoras Press), 1–22, 7.

- 33 Farrell, 79.
- 34 Lanham, The Electronic Word, 227.
- 35 Richard Lanham, "The Economics of Attention," *Michigan Quarterly Review* 36 (Spring 1997): 270.
- On the relationship of rhetoric and power, see: James A. Berlin, "Revisionary Histories of Rhetoric: Politics, Power, and Plurality," in *Writing Histories of Rhetoric*, ed. Victor Vitanza (Carbondale, IL: Southern Illinois University Press, 1994), 112–127.
- 37 Rebecca R., Hastings, "Communication Skills Key for Young Workers", Society for Human Resource Management (March 8, 2012), www.shrm.org/resourcesandtools/hrtopics/employee-relations/pages/communicationskillskey.aspx. Accessed September 1, 2016.
- 38 See: Michael Billig, *Ideology and Opinion* (London: Sage, 1991).
- 39 Billig, Ideology, 5.
- 40 One of the earliest explorations of this issue is found in: Robert L. Scott, "On Viewing Rhetoric as Epistemic," *Central States Speech Journal* 18 (February 1967): 9–16. See also: Lloyd F. Bitzer, "Rhetoric and Public Knowledge," in *Rhetoric, Philosophy, and Literature: An Exploration* (West Lafayette, IN: Purdue University Press, 1978), 67–93.
- 41 See: *Rhetoric and Community: Studies in Unity and Fragmentation*, ed. Michael J. Hogan (Columbia, SC: University of South Carolina Press, 1998), Introduction, xv.
- 42 Hogan, 292.
- 43 Lincoln, 316–317.
- Emily Dickinson, "Success Is Counted Sweetest." Reprinted by permission of the publishers and the Trustees of Amherst College from *The Poems of Emily Dickinson*, ed. Thomas H. Johnson (Cambridge, MA: The Belknap Press of Harvard University, copyright © 1951, 1955, 1979 by the President and Fellows of Harvard College).
- 45 Used with permission of the Fannie Lou Hamer Memorial Garden and Museum Foundation, Charles McLaurin, President of the Board.

- 1 Richard Leo Enos, *Greek Rhetoric before Aristotle* (Prospect Heights, IL: Waveland, 1993), 4.
- 2 Enos, 5.
- 3 Enos, 6.
- 4 Enos, 7–8.
- 5 Brian Boyd, On the Origin of Stories: Evolution, Cognition, and Fiction (Cambridge, MA: Harvard University Press, 2009), 234.
- 6 Jane Sutton, "The Marginalization of Sophistical Rhetoric and the Loss of History," in *Rethinking the History of Rhetoric: Multidisciplinary Essays on the History of Rhetoric*, ed. Takis Poulakos (Boulder, CO: Westview, 1993), 87.
- 7 The Greek Sophists, trans. John Dillon and Tania Gergel (New York: Penguin Books, 2003), xi.
- 8 John Poulakos, "Terms for Sophistical Rhetoric," in *Rethinking the History of Rhetoric: Multidisciplinary Essays on the History of Rhetoric*, ed. Takis Poulakos (Boulder, CO: Westview, 1993), 53–74, p. 56.
- 9 Poulakos, 57.
- 10 Susan Jarratt and Rory Ong, "Aspasia: Rhetoric, Gender, and Colonial Ideology," in *Reclaiming Rhetorica: Women in the Rhetorical Tradition*, ed. Andrea A. Lunsford (Pittsburgh, PA: University of Pittsburgh Press, 1995), 14.

- 11 Dillon and Gergel, ix.
- 12 Josiah Ober, Mass and Elite in Ancient Athens: Rhetoric, Ideology, and the Power of the People (Princeton, NJ: Princeton University Press, 1989), 7.
- 13 The First Philosophers: The Presocratics and the Sophists, trans. Robin Waterfield (Oxford: Oxford University Press, 2009), xxxii.
- 14 Jarratt and Ong, 14.
- 15 H. D. F. Kitto, *The Greeks* (1951; Baltimore, MD: Penguin Books, 1968), 120.
- 16 Kitto, 115.
- 17 Dillon and Gergel, x.
- 18 John Poulakos, *Sophistical Rhetoric in Classical Greece* (Columbia, SC: University of South Carolina Press, 1995), 16–17.
- 19 Maurice Balme and Gilbert Lawall, *Athenaze: An Introduction to Ancient Greek*, Book II (New York: Oxford University Press, 1991), 105–106.
- Jacqueline de Romilly, *The Great Sophists in Periclean Athens*, trans. Janet Lloyd (Oxford: Clarendon Press, 1992), 30.
- 21 The First Philosophers, xxvii.
- 22 Greek Sophists, xi.
- W. K. C. Guthrie, *The Sophists* (Cambridge: Cambridge University Press, 1971), 20. Guthrie quotes Plato's *Gorgias*.
- John Poulakos, "Toward a Sophistic Definition of Rhetoric," *Philosophy and Rhetoric* 16 (1983): 35–48; see: Ober.
- 25 H. D. Rankin, Sophists, Socratics and Cynics (London: Croom Helm, 1983), 15. Other helpful discussions of this period in Greek thought, and of the Sophists, include: Harold Barrett, The Sophists: Rhetoric, Democracy and Plato's Idea of Sophistry (Novato, CA: Chandler and Sharp, 1987); G. B. Kerferd, The Sophistic Movement (New York and Cambridge: Cambridge University Press, 1981); John Sallis, Being and Logos: Reading the Platonic Dialogues (Atlantic Highlands, NJ: Humanities Press, 1986).
- 26 Jarratt and Ong, 12.
- 27 Enos, ix.
- 28 See: Michael Gagarin, "Telling Stories in Athenian Law," *Transactions of the American Philological Association* 133 (2003): 2, 197–207.
- 29 Michael Gagarin, *The Murder of Herodes: A Study of Antiphon 5* (New York: Verlag Peter Lang, 1989), 14.
- 30 Lynda Walsh, "The Rhetoric of Oracles," *Rhetoric Society Quarterly* 33(3) (Summer, 2003): 55–78, p. 60.
- 31 Ober, 7–8.
- On the meaning of *Sophist*, see: Edward Schiappa, *Protagoras and Logos: A Study in Greek Philosophy and Rhetoric* (Columbia, SC: University of South Carolina Press, 1991), Chap. 1.
- 33 The First Philosophers, xi.
- 34 Poulakos, "Terms," 58.
- 35 See, for example: Susan C. Jarratt, *Rereading the Sophists: Classical Rhetoric Refigured* (Carbondale, IL: Southern Illinois University Press, 1991).
- 36 See: Guthrie.
- 37 Michael Gagarin, Antiphon the Athenian: Oratory, Law and Justice in the Age of the Sophists (Austin, TX: University of Texas Press, 2002), 1.
- 38 Gagarin, Antiphon, 5, 6.
- 39 Gagarin, Antiphon, 16-17.

- 40 Gagarin, Antiphon, 18.
- 41 See: Mario Untersteiner, *The Sophists* (Oxford: Oxford University Press, 1954).
- 42 Guthrie, 25. Rankin writes that *areté* "combines the factors both of high moral virtue and worldly success" (13).
- 43 The First Philosophers, xv.
- 44 Ober, 11.
- 45 Poulakos, "Terms," 57.
- 46 Friedrich Nietzsche, *Friedrich Nietzsche on Rhetoric and Language*, ed. trans. S. Gilman, C. Blair, and D. Parent (New York: Oxford University Press, 1989), 19.
- 47 See the discussion of *kairos* in Ekaterina V. Haskins, *Logos and Power in Isocrates* and *Aristotle* (Columbia: University of South Carolina Press, 2004), 66–70.
- Dale Sullivan, "*Kairos* and the Rhetoric of Belief," *Quarterly Journal of Speech* 78 (August 1992): 320.
- 49 Jarratt and Ong, 16.
- 50 Rankin, 14.
- 51 See: Schiappa, Chap. 1.
- 52 See: Glen Warren Bowersock, *Greek Sophists in the Roman Empire* (Oxford: Clarendon Press, 1969).
- Andrew Ford, "The Price of Art in Isocrates: Formalism and the Escape from Politics," in *Rethinking the History of Rhetoric*, ed. Takis Poulakos (Boulder, CO: Westview, 1992), 37.
- 54 Ford, 37.
- 55 Ford, 37.
- 56 Jarratt, xx.
- 57 Poulakos, Sophistical Rhetoric, 25.
- 58 Richard A. Katula and James J. Murphy eds., *A Synoptic History of Classical Rhet-oric* (Davis, CA: Hermagoras Press, 1995), 28.
- 59 See Gagarin, Murder, 47.
- 60 Jarratt, 42.
- 61 Greek Sophists, xv.
- 62 Sutton, 87.
- On Gorgias' philosophy of *logos*, see: Charles P. Segal, "Gorgias and the Psychology of the *Logos*," *Harvard Studies in Classical Philology* 66 (1962): 99–155.
- 64 Enos, 72.
- 65 See: John O. Ward, "Magic and Rhetoric from Antiquity to the Renaissance: Some Ruminations," *Rhetorica* 6 (Winter 1988): 1, 57–118, p. 58.
- 66 Segal, 110. Quoted in Bruce E. Gronbeck, "Gorgias on Rhetoric and Poetic: A Rehabilitation," *Southern Speech Communication Journal* 38 (Fall 1972): 1, 27–38.
- 67 Gronbeck, 33.
- 68 George Kennedy, Classical Rhetoric and Its Christian and Secular Tradition from Ancient to Modern Times, 2nd ed. (Chapel Hill, NC: University of North Carolina Press, 1999), 35.
- 69 Jeffrey Walker, *Rhetoric and Poetics in Antiquity* (Oxford: Oxford University Press, 2000), 154.
- Jacqueline de Romilly, *Magic and Rhetoric in Ancient Greece* (Cambridge, MA: Harvard University Press, 1975).
- 71 de Romilly, Magic, 16.
- 72 de Romilly, Magic, 6.
- 73 Jane P. Tompkins, "The Reader in History: The Changing Shape of Literary Response," in *Reader-Response Criticism: From Formalism to Post-Structuralism*, ed. Jane P. Tompkins (Baltimore, MD: Johns Hopkins University Press, 1980), 203–204.

- 74 G. M. A. Grube, *The Greek and Roman Critics* (Toronto: University of Toronto Press, 1965), 16.
- 75 Kennedy, 64.
- Gorgias, *Encomium on Helen* 9 and 10, trans. LaRue VanHook, *Classical Weekly* 6 (1913): 122–123. Quoted in Gronbeck, 34.
- 77 Kennedy, 35.
- 78 de Romilly, *Magic*, 3.
- 79 de Romilly, Magic, 4.
- 80 de Romilly, Magic, 5.
- 81 Grube, 16.
- 82 Rosamond Kent Sprague, ed. *The Older Sophists* (Columbia, SC: University of South Carolina Press, 1972), 50.
- 83 For a discussion of the relationship between truth and memory, see Haskins, 12–15.
- 84 Kennedy, 66.
- 85 Balme and Lawall, 106.
- 86 Billig, 40. Billig cites Philostratus, *Lives of the Sophists*, trans. W. C. Wright (London: Loeb Classical Library, 1965); Rankin, 30, ff.
- 87 Plato, Theaetetus, 151e-152a.
- 88 Rankin, 32.
- 89 Jarratt and Ong, 15.
- 90 Poulakos, "Terms," 58–59.
- 91 Poulakos, "Terms," 60.
- 92 de Romilly, Great Sophists, viii.
- 93 Michael C. Leff, "What Is Rhetoric?," *Rethinking Rhetorical Theory, Criticism, and Pedagogy: The Living Art of Michael C. Leff*, ed. A. de Velasco, J. A. Campbell, and D. Henry (East Lansing, MI: Michigan State University Press, 2016), 471–481, p. 479.
- 94 Brian Vickers, In Defense of Rhetoric (Oxford: Clarendon Press, 1988), 155.
- Ford, "The Price of Art in Isocrates," 37.
- Michael C. Leff, "Tradition and Agency in Humanistic Rhetoric," Rethinking Rhetorical Theory, Criticism, and Pedagogy: The Living Art of Michael C. Leff, ed. A. de Velasco, J. A. Campbell, and D. Henry (East Lansing, MI: Michigan State University Press, 2016), 7–24, p. 15.
- 97 See Haskins, 19. Also, Takis Poulakos, *Speaking for the Polis: Isocrates' Rhet-orical Education* (Columbia, SC: University of South Carolina Press, 1997).
- 98 Quoted in Sowerby, 108.
- 99 Vickers, 150.
- 100 The quote is from Against the Sophists. Quoted in Vickers, 150.
- 101 Vickers, 154.
- 102 Poulakos, Sophistical Rhetoric, 134.
- 103 Walker, 175.
- 104 On the importance of writing as an intellectual activity at this time, see: Kathleen E. Welch, *The Contemporary Reception of Classical Rhetoric: Appropriations of Ancient Discourse* (Hillsdale, NJ: Lawrence Erlbaum, 1990), 16–17.
- 105 Ford, 38.
- 106 Antidosis, 254. Quoted in Vickers, 156.
- 107 Enos, 114.
- 108 Jarratt and Ong, 13.
- 109 C. Jan Swearingen, "A Lover's Discourse: Diotima, Logos, and Desire," in *Reclaiming Rhetorica: Women in the Rhetorical Tradition*, ed. Andrea Lunsford (Pittsburgh, PA: University of Pittsburgh Press, 1995), 25–26.

- 110 Jarratt and Ong, 14.
- Ellen Meiksins Wood, *Peasant–Citizen and Slave: The Foundations of Athenian Democracy* (London: Verso, 1988), 115. Quoted in Jarratt and Ong, 13.
- 112 Jarratt and Ong, 10.
- 113 Jarratt and Ong, 12.
- 114 Jarratt and Ong, 13.
- Jarratt and Ong, 15. *The* Menexenus *in The Dialogues of Plato*, trans. B. Jowett (New York: Random House, 1937), 235.
- 116 Cheryl Glenn, "Locating Aspasia on the Rhetorical Map," in *Listening to Their Voices: The Rhetorical Activities of Historical Women*, ed. Molly Meijer Wertheimer (Columbia, SC: University of South Carolina Press, 1997), 19–41, p. 21.
- Sappho, "Please," trans. Paul Roche, *Isle of Lesbos*, www.sappho.com/poetry/sappho.html#not%20one%20word. Accessed September 30, 2016.
- 118 Friedrich Nietzsche on Rhetoric and Language, 3.
- 119 de Romilly, Great Sophists, 4.

- To view the complete text of any of Plato's dialogues, visit www.persues.tufts.edu. For a discussion of Plato's *Phaedrus* as rhetoric, see: Jane V. Curran, "The Rhetorical Technique of Plato's *Phaedrus*," *Philosophy and Rhetoric* 19 (1986): 66–72. There are many excellent sources on Plato, his philosophy, and his thoughts on specific topics. Such works that take up issues discussed in this chapter include: I. M. Crombie, *An Examination of Plato's Doctrines*, v. I (London: Routledge and Kegan Paul, 1961); Michael Despland, *The Education of Desire: Plato and the Philosophy of Religion* (Toronto: University of Toronto Press, 1985); and R. M. Hare, *Plato* (Oxford: Oxford University Press, 1982).
- 2 On these dialogues, see, for example: Plato's *Sophist*, trans. William S. Cobb (Savage, MD: Rowman and Littlefield, 1990); Plato, *Protagoras*, trans. Christopher Charles Whiston Taylor (Oxford: Clarendon Press, 1991); and B. A. F. Hubbard and E. S. Karnofsky, *Plato's Protagoras: A Socratic Commentary* (Chicago, IL: University of Chicago Press, 1982).
- 3 Brian Vickers, In Defense of Rhetoric (Oxford: Oxford University Press, 1988), 148.
- 4 For a detailed discussion of this question, see: Vickers, Chapter 3, "Territorial Disputes: Philosophy versus Rhetoric," 148–213.
- 5 George Kennedy, *Classical Rhetoric and Its Christian and Secular Tradition*, 2nd ed. (Chapel Hill, NC: University of North Carolina Press, 1999), 54.
- 6 A good introduction to these dialogues and their relationship to one another is: Seth Benardete, *The Rhetoric of Morality and Philosophy: Plato's Gorgias and Phaedrus* (Chicago, IL: University of Chicago Press, 1991). Also helpful: Adele Spitzer, "The Self-Reference of the *Gorgias*," *Philosophy and Rhetoric* 8 (1975): 1–22.
- 7 All passages are from *Gorgias*, trans. W. C. Helmbold (Indianapolis, IN: Bobbs-Merrill, 1952). Other translations include: Plato, *Gorgias*, trans. T. Irwin (Oxford: Clarendon Press, 1979). For a detailed introduction to and commentary on *Gorgias*, see: George Kimball Plochmann and Franklin E. Robinson, *A Friendly Companion to Plato's Gorgias* (Carbondale, IL: Southern Illinois University Press, 1988).
- 8 Vickers, 153.
- 9 Josiah Ober, *Mass and Elite in Democratic Athens* (Princeton, NJ: Princeton University Press, 1989), 105.

- 10 George Kennedy, *The Art of Persuasion in Greece* (Princeton, NJ: Princeton University Press, 1963), 185.
- 11 Bruce Gronbeck, "Gorgias on Rhetoric and Poetic: A Rehabilitation," *Southern Speech Communication Journal* 38 (Fall 1972): 35.
- 12 For discussions of Plato's theory of art as it relates to rhetoric, see: Hans-George Gadamer, *The Idea of the Good in Platonic-Aristotelian Philosophy* (New Haven, CT: Yale University Press, 1986), 46–49, 78–80; and Rupert C. Lodge, *Plato's Theory of Art* (1953; New York: Russell and Russell, 1975), Chaps. III and IV.
- 13 Crombie suggests that rhetoric's product would be "spiritual order," evidenced in "justice and self-restraint," and that these are "the qualities that the scientific orator would aim to produce in his hearers" (197).
- On the themes treated in Gorgias, see: Edwin Black, "Plato's View of Rhetoric," *Quarterly Journal of Speech* 44 (December 1958): 361–374.
- 15 Jessica Moss, "Right Reason in Plato and Aristotle: On the Meaning of *Logos*," *Phronesis* 59(3) (2014): 181–230, p. 188.
- Janet M. Atwill, *Rhetoric Reclaimed: Aristotle and the Liberal Arts Tradition* (Ithaca, NY: Cornell University Press, 1998), 6.
- 17 Edward Schiappa, "Did Plato Coin *Rhetorike*?" *American Journal of Philology* 111 (1990): 457–470. See also, Kennedy, *Classical Rhetoric*, 59.
- 18 Vickers, 97.
- 19 Ober, 107.
- 20 Spitzer, 10.
- 21 Marcel Detienne, *The Creation of Mythology*, trans. Margaret Cook (Chicago, IL: University of Chicago Press, 1986), 85.
- 22 Cicero, De Oratore, 1. 11. 47. Quoted in Enos, 91.
- 23 Vickers, 113–120.
- 24 Richard Leo Enos, *Greek Rhetoric before Aristotle* (Prospect Heights, IL: Waveland, 1993), 92.
- 25 Enos, 72.
- Some scholars have argued that as Plato criticizes rhetoric in *Gorgias*, he also employs the art. See, for example: Jan Swearingen, *Rhetoric and Irony* (Oxford: Oxford University Press, 1991).
- 27 Jacqueline de Romilly, *The Great Sophists in Periclean Athens*, trans. Janet Lloyd (Oxford: Clarendon Press, 1992), 71.
- John Poulakos, *Sophistical Rhetoric in Classical Greece* (Columbia, SC: University of South Carolina Press, 1995), 79.
- 29 Plato, *Phaedrus*, trans. W. C. Helmbold and W. G. Rabinowitz (Indianapolis, IN: Liberal Arts Press, 1956), 12. All passages from *Phaedrus* are taken from this edition unless otherwise indicated.
- This view may remind you of Freud's id, ego, and superego, and, while there are some similarities in the two theories, the analogy is not exact.
- Jacqueline de Romilly, *Magic and Rhetoric in Ancient Greece* (Cambridge, MA: Harvard University Press, 1975), 51.
- 32 George Maximilian Antony Grube, *Plato's Thought* (1935; rpt. Boston, MA: Beacon Press, 1958), 214–215.
- T. Benson and Michael H. Prosser, *Readings in Classical Rhetoric* (Boston, MA: Allyn and Bacon, 1969), 22.
- Regarding the relationship between the harmony of the individual and the harmony of the state, Moline explains that on Plato's view:

[T]he individual is wise in the same way, and in the same part of himself, as the city ... And the part which makes the individual brave is the same as that which makes the city brave, and in the same manner, and everything which makes for virtue is the same in both ... The city was just because each of the three classes in it was fulfilling its own task. Jon Moline, *Plato's Theory of Understanding* (Madison, WI: University of Wisconsin Press, 1981), 55.

- 35 Moline, 65.
- 36 Moline, 65.
- 37 Grube, 212.
- 38 Hare, 54.
- 39 Susan Jarratt, *Rereading the Sophists: Classical Rhetoric Refigured* (Carbondale, IL: Southern Illinois University Press, 1991), xvi.
- 40 Patricia Bizzell and Bruce Herzberg, *The Rhetorical Tradition: Readings from Classical Times to the Present* (Boston, MA: St. Martin's Press, 2000), 28.
- 41 de Romilly, *Magic*, 58.
- 42 See: Walter J. Ong, *Orality and Literacy: The Technologizing of the Word*, 2nd ed. (New York: Routledge, 2002).

- 1 George Kennedy, trans. *Aristotle on Rhetoric: A Theory of Civic Discourse* (New York: Oxford University Press, 1991), introduction, p. 3. To view the complete text of Aristotle's *Rhetoric* visit www.perseus.tufts.edu.
- 2 Jonathan Barnes, ed., *The Cambridge Companion to Aristotle* (Cambridge, UK: Cambridge University Press, 1995), 4. On Aristotle's life and his relationship with Plato, see: Geoffrey Ernest Richard Lloyd, *Aristotle: The Growth and Structure of His Thought* (Cambridge, UK: Cambridge University Press, 1968), Chaps. 1–3.
- 3 Barnes, 4.
- 4 Forbes I. Hill, "The Rhetoric of Aristotle," in *A Synoptic History of Classical Rhetoric*, ed. James J. Murphy (Davis, CA: Hermagoras Press, 1983), 19.
- 5 Barnes, 8.
- 6 Kennedy's translation, cited in note 1, is helpful. Other widely used translations of the *Rhetoric* include those by Lane Cooper (New York: Appleton-Century-Crofts, 1932); and W. Rhys Roberts (New York: Modern Library, 1954).
- 7 Kennedy, 5.
- 8 Hill, 24.
- 9 Unless otherwise indicated, all quotations from *Rhetoric* are from: Aristotle, *Rhetoric*, trans. W. Rhys Roberts (New York: Modern Library, 1954).
- 10 Kathleen Welch quotes Ellen Quandahl to the effect that "The Rhetoric is Difficult to Read, Full of Discrepancies, Gaps, Repetitions," in The Contemporary Reception of Classical Rhetoric: Appropriations of Ancient Discourse (Hillsdale, NJ: Lawrence Erlbaum, 1990), 22; Quandahl, "Aristotle's Rhetoric: Reinterpreting Invention," Rhetoric Review 4 (January 1986): 128.
- 11 Kennedy, appendix I, 289–290.
- 12 See: Larry Arnhart, *Aristotle on Political Reasoning* (DeKalb, IL: Northern Illinois University Press, 1981), 18–19.
- 13 Arnhart, 19.
- 14 Arnhart, 39.

- 15 Kennedy, 34, note 24.
- 16 See: Gerard A. Hauser, "The Example in Aristotle's *Rhetoric*: Bifurcation or Contradiction?" *Philosophy and Rhetoric* 1 (1968): 78–90.
- On Aristotle's theory of the enthymeme, see: Lloyd Bitzer, "Aristotle's Enthymeme Revisited," *Quarterly Journal of Speech* 45 (1959): 399–408.
- 18 See: Jeffrey Walker, *Rhetoric and Poetics in Antiquity* (Oxford: Oxford University Press, 2000), 175.
- 19 Bitzer, 408.
- For a detailed discussion of Aristotle's divisions of speeches, see: Ekaterina V. Haskins, *Logos and Power in Isocrates and Aristotle* (Columbia, SC: University of South Carolina Press, 2004), 58 ff.
- 21 The Kennedy translation of Aristotle's *Rhetoric* has been very helpful in preparing this section.
- Jane Sutton, "The Marginalization of Sophistical Rhetoric and the Loss of History," in *Rethinking the History of Rhetoric*, ed. Takis Poulakos (Boulder, CO: Westview 1993), 75–90.
- 23 Sutton, 84.
- 24 Quoted in Sutton, 84.
- 25 On Aristotle's thinking about happiness, see: Anthony Kenny, *Aristotle on the Perfect Life* (Oxford: Clarendon Press, 1992).
- 26 Walker, 9.
- 27 Kennedy trans., Bk. 1, Chap. 3, sect. 3. (Emphasis added.)
- For a detailed answer to the question of rhetoric's subject matter, see Fr. William Grimaldi's "The Role of the *PISTEIS* in Aristotle's Methodology," appendix to *Aristotle, Rhetoric I: A Commentary* (New York: Fordham University Press, 1980), 349–356.
- 29 John H. Randall, Jr., Aristotle (New York: Columbia University Press, 1960), 253.
- 30 On the difference between ancient and modern understandings of the relationship of thought and expression, see: Welch, 15 ff.
- See, for example: George Maximilian Antony Grube, translator's introduction to Longinus, *On the Sublime* (Indianapolis, IN: Library of Liberal Arts, 1957), x.
- 32 Barnes, 267.
- 33 Arnhart, 37. See also: David Charles, *Aristotle's Philosophy of Action* (Ithaca, NY: Cornell University Press, 1984), 176–188.
- 34 William W. Fortenbaugh, Aristotle on Emotion (New York: Barnes and Noble, 1975), 17.
- 35 Fortenbaugh, 17.
- 36 Fortenbaugh, 17.
- 37 Fortenbaugh, 18.
- James M. May, Trials of Character: The Eloquence of Ciceronian Ethos (Chapel Hill, NC: University of North Carolina Press, 1988), 2.
- 39 On *topoi* and Roman *loci*, see: Michael Leff, "The Topics of Argumentative Invention in Latin Rhetorical Theory from Cicero to Boethius," *Rhetorica* 1 (Spring 1983): 23–44.
- 40 Kennedy translation.
- 41 See, for instance, Leff.
- 42 On this view, see: Fr. William M. A. Grimaldi, "The Aristotelian Topics," *Traditio* 14 (1958): 1–16.
- Thomas M. Conley, "Logical Hylomorphism' and Aristotle's *Koinoi Topoi*," *Central States Speech Journal* 29 (Summer 1978): 92–97.
- Donovan J. Ochs, "Aristotle's Concept of Formal Topics," *Speech Monographs* 36 (1969): 419–425.

- 1 Joy Connolly, *The State of Speech: Rhetoric and Political Though in Ancient Rome* (Princeton, NJ: Princeton University Press, 2013), 1.
- 2 Manfred Fuhrmann, *Cicero and the Roman Republic*, trans. W. E. Yuill (Oxford: Blackwell, 1992), 18.
- 3 Erich Gruen, "Codes of Rome," Times Literary Supplement (July 3, 2009), 10.
- 4 The passage is cited by Connolly, 232.
- 5 James M. May, *Trials of Character: The Eloquence of Ciceronian Ethos* (Chapel Hill, NC: University of North Carolina Press, 1988), 6.
- 6 Connolly, 31.
- 7 Quoted in Connolly, 12.
- 8 Harold C. Gotoff, *Cicero's Caesarian Speeches: A Stylistic Commentary* (Chapel Hill, NC: University of North Carolina Press, 1993), x.
- 9 Michael Mooney, *Renaissance Thought and Its Sources* (New York: Columbia University Press, 1979), 36.
- 10 Graham Anderson, Sage, Saint, and Sophist: Holy Men and Their Associates in the Early Roman Empire (London: Routledge, 1994).
- 11 See: Ann Vasaly, *Representations: Images of the World in Ciceronian Oratory* (Berkeley, CA: University of California Press, 1993).
- 12 George Maximilian Antony Grube, translator's introduction to Longinus, *On the Sublime* (Indianapolis, IN: Library of Liberal Arts, 1957), ix.
- 13 Calvin Troup, *Temporality, Eternity, and Wisdom: The Rhetoric of Augustine's Confessions* (Columbia, SC: University of South Carolina Press, 1999), 13.
- 14 Fuhrmann, 19.
- 15 Longinus, *On the Sublime*, trans. George Maximilian Antony Grube (Indianapolis, IN: Library of Liberal Arts, 1957), x.
- 16 Connolly, 3.
- 17 Christian Habicht, *Cicero the Politician* (Baltimore, MD: Johns Hopkins University Press, 1990), 1.
- Habicht, 2. Habicht provides a brief and readable account of Cicero's political career. For a more detailed account of Cicero's life, see: David R. Shackleton-Bailey, *Cicero* (New York: Charles Scribner's Sons, 1971).
- 19 See: Christopher P. Craig, Form as Argument in Cicero's Speeches: A Study of Dilemma, American Classical Studies, #31 (Atlanta, GA: Scholars Press, 1993).
- 20 Gotoff, xii.
- 21 Mooney, 8.
- On the *Rhetorica ad Herennium*, see: James J. Murphy, "The Age of Codification: Hermagoras and the Pseudo-Ciceronian *Rhetorica ad Herennium*," in *A Synoptic History of Classical Rhetoric*, ed. Richard Katula and James J. Murphy (Davis, CA: Hermagoras Press, 1983), 77–89. For a detailed discussion of *De Inventione* (and Cicero's other works on rhetoric), see: Donovan J. Ochs, "Cicero's Rhetorical Theory," in *A Synoptic History of Classical Rhetoric*, 90–150.
- 23 Craig, 14.
- 24 Brian Vickers, In Defense of Rhetoric (Oxford: Oxford University Press, 1988), 28.
- 25 Cicero, *De Inventione*, trans. H. M. Hubbell (Cambridge, MA: Loeb Classical Library, 1976a).
- 26 M. L. Clark, Rhetoric at Rome (New York: Barnes and Noble, 1953), 53.
- 27 Habicht, 2.

- 28 De Inventione, I. 1.
- 29 De Inventione, I. iv. 5.
- 30 De Inventione, I. vi. 9.
- 31 Connolly, 129.
- 32 My translation. The passage is also cited by Connolly, 231.
- 33 De Inventione, I. vii. 9. (Emphasis added.)
- 34 *De Inventione*, I. 11–19.
- 35 De Inventione, I. viii. 11.
- 36 De Inventione, I. viii. 16.
- 37 Connolly, 73.
- A good survey of the complex territory of loci and topical systems is Michael C. Leff's "The Topics of Argumentative Invention in Latin Rhetorical Theory," *Rhetorica* (1), 1983, 23–44. Reprinted in *Rethinking Rhetorical Theory, Criticism, and Pedagogy: The Living Art of Michael C. Leff* (East Lansing MI: Michigan State University Press, 2016), 65–86.
- 39 Donovan J. Ochs, "Cicero's Topica: A Process View of Invention," in Explorations in Rhetoric: Essays in Honor of Douglas Ehninger, ed. Ray E. McKerrow (Glenview, IL: Scott, Foresman, 1982), 117. Quoted in Kathleen E. Welch, The Contemporary Reception of Classical Rhetoric: Appropriations of Ancient Discourse (Hillsdale, NJ: Lawrence Erlbaum, 1990), 60.
- 40 Edward P. J. Corbett, "The *Topoi* Revisited," in *Rhetoric and Praxis: The Contribution of Classical Rhetoric to Practical Reasoning*, ed. Jean Deitz Moss (Washington, DC: Catholic University of America Press, 1986), 47. Quoted in Welch, 60–61.
- 41 Cicero discusses these loci in De Inventione, Book I. xxiv-xxviii.
- 42 De Inventione, I. xxiv. 34.
- 43 Leff, 27.
- 44 De Inventione, I. xxvi–xxviii.
- 45 Leff, 28.
- 46 Cicero, *De Oratore*, trans. E. W. Sutton and H. Rackham (Cambridge, MA: Harvard University Press, 1976b), ix ff.
- 47 On the Aristotelian influences in *De Oratore*, see: May, 3 ff.
- 48 De Oratore, I. ii. 5.
- 49 De Oratore, I. xxviii. 128. Quoted in Mooney, 36.
- 50 De Oratore, I. vii. 30.
- 51 Connolly, 14.
- 52 De Oratore, III. xvi. 60-61.
- 53 De Oratore, I. iii. 12. See also: Stef E. W. Bugter, "Sensus Communis in the works of M. Tullius Cicero," in Common Sense: The Foundations for Social Sciences. ed. F. L. Van Holthoon and David R. Olson (New York: University Press of America, 1987), 83–98.
- 54 Ancient Rhetoric: From Aristotle to Philostratus, trans. Thomas Habinek (New York: Penguin, 2017), 70.
- 55 De Oratore, I. v. 18. (Emphasis added.)
- 56 De Oratore, II. xliv. 187 ff.
- 57 De Oratore, II. xliv. 188.
- 58 De Oratore, I. v. 20.
- 59 De Oratore, I. viii. 34.
- 60 De Oratore, II. lvi. 228.
- 61 De Oratore, I. vi. 229.

- 62 De Oratore, II. lviii-lix. 338-339.
- 63 De Oratore, II. lx.
- 64 De Oratore, II. lx. 247.
- 65 De Oratore, II. lxi. 250.
- 66 De Oratore, II. lxi. 251.
- 67 De Oratore, II. lxi. 252.
- 68 De Oratore, II. lxi. 253.
- 69 De Oratore, II. 1xiv. 258-259.
- 70 De Oratore, II. lxiv. 260.
- 71 R. G. M. Nisbet, "The Speeches," in *Cicero*, ed. T. A. Dorey (New York: Basic Books, 1965), 56.
- 72 Quoted in Prentice A. Meador, Jr., "Quintilian and the *Institutio Oratoria*," in *A Synoptic History of Classical Rhetoric* (Davis, CA: Hermagoras Press, 1983), 151–176.
- On Quintilian's teaching, see: James J. Murphy, "Roman Writing Instruction as Described by Quintilian," in *A Short History of Writing Instruction*, ed. James J. Murphy (Davis, CA: Hermagoras Press, 1990), 19–76.
- Quintilian, *Institutio Oratoria*, 4 vols., trans. H. E. Butler (Cambridge, MA: Harvard University Press, 1959–1963), Loeb Classical Library. See also: James J. Murphy, Quintilian, *On the Early Education of the Citizen Orator*, trans. John S. Watson (Indianapolis, IN: Library of Liberal Arts, 1965).
- 75 Quintilian, Institutes of Oratory, I. 1.
- 76 A. Wallace-Hadrill, *Rome's Cultural Revolution* (Cambridge: Cambridge University Press, 2008), 47.
- 77 Quoted in Meador, 155.
- 78 Quintilian, *Institutes*, II. 11.
- 79 Leff, 33.
- 80 Quintilian, *Institutes*, v. 10, 125. Quoted in Leff, 34.
- For a more detailed discussion of ancient systems for dividing speeches, see: Vickers, 67–72.
- 82 Longinus, *On the Sublime*, trans. George Maximilian Antony Grube (Indianapolis, IN: Library of Liberal Arts, 1957). All passages from *On the Sublime* are from this translation.
- 83 Vickers, 307.
- 84 Grube, x-xi.
- 85 Jane Tompkins, "The Reader in History: The Changing Shape of Literary Response," in *Reader Response Criticism*, ed. Jane P. Tompkins (Baltimore, MD: Johns Hopkins University Press, 1980), 203.
- 86 Tompkins, 203.
- 87 Longinus, 10.
- 88 Longinus, 10.
- 89 Longinus, 17. Translation of the love ode of Sappho by J. A. Symonds (1883).
- 90 Longinus, 29.
- 91 Longinus, 29.
- 92 Longinus, 31.
- 93 Vickers, 310. Longinus quote from On the Sublime, 29.2.
- 94 Grube, 325.
- 95 Grube, 325.
- 96 Grube, 325, 326.

- 97 Grube, 328.
- 98 Katula and Murphy, 206.
- 99 George Kennedy, *A Comparative Rhetoric* (New York: Oxford University Press, 1998), 211.
- 100 Thomas M. Conley, Rhetoric in the European Tradition (New York: Longman, 1994), 60.

- George A. Kennedy, Classical Rhetoric and Its Christian and Secular Tradition from Ancient to Modern Times (Chapel Hill, NC: University of North Carolina Press, 1980), 174.
- 2 Karin Margareta Fredborg, "Twelfth-Century Ciceronian Rhetoric: Its Doctrinal Development and Influences," in *Rhetoric Revalued*, ed. Brian Vickers (Binghamton, NY: Center for Medieval and Early Renaissance Studies, 1982), 88.
- 3 Roger P. Parr writes that "Geoffrey's [of Vinsauf] primary source, particularly for the devices of style, was *Rhetorica ad Herennium*." Translator's introduction to Geoffrey of Vinsauf's *Documentum do Modo et Arte Dictandi et Versificandi* (Milwaukee, WI: Marquette University Press, 1968), 2.
- 4 Fredborg, 88.
- 5 Kathleen E. Welch, *The Contemporary Reception of Classical Rhetoric* (New York: Lawrence Earlbaum, 1990), 15.
- 6 John O. Ward, "From Antiquity to the Renaissance: Glosses and Commentaries on Cicero's *Rhetorica*," in *Medieval Eloquence: Studies in the Theory and Practice of Medieval Rhetoric*, ed. James J. Murphy (Berkeley, CA: University of California Press, 1978), 54, 74. Quoted in Vickers, 216.
- 7 Joseph M. Miller, Michael H. Prosser, and Thomas, Benson, eds., *Readings in Medieval Rhetoric* (Bloomington, IN: Indiana University Press, 1974), xiii.
- 8 Miller, xiii.
- 9 See, for example: Vickers, Chapt. 4: Medieval Fragmentation, 214–253.
- 10 Christian Habicht, *Cicero the Politician* (Baltimore, MD: Johns Hopkins University Press, 1990), 2.
- 11 Eleonore Stump and David Vincent Mecon, ed., *The Cambridge Companion to Augustine* (Cambridge UK: Cambridge University Press, 2014), 2–3.
- 12 Quoted in Calvin Troup, *Temporality, Eternity, and Wisdom: The Rhetoric of Augustine's Confessions* (Columbia, SC: University of South Carolina Press, 1999), 1. Augustine, *Confessions*, 9.2.4.
- 13 Troup, 4.
- 14 Kennedy, 149 ff.
- 15 Confessions 1.9.14. Quoted in Troup, 11.
- 16 Troup, 27.
- James J. Murphy, "Saint Augustine and the Debate about a Christian Rhetoric," *Quarterly Journal of Speech* 46 (1960): 400–410.
- 18 Augustine, *On Christian Doctrine*, trans. D. W. Robertson (Indianapolis, IN: Library of Liberal Arts), 7.
- 19 Renato Barilli, *Rhetoric*, trans. Giuliana Menozzi (Minneapolis, MN: University of Minnesota Press, 1989), 42.
- W. R. Johnson, "Isocrates Flowering: The Rhetoric of Augustine," *Rhetoric and Philosophy* 9(4) (1976): 220.

- 21 James J. Murphy, ed., *Three Medieval Rhetorical Arts* (Berkeley, CA: University of California Press, 1971), xiii.
- 22 Murphy, *Arts*, xiii. (Emphasis added.)
- 23 Ward, 27.
- 24 Johnson, 220.
- 25 Augustine, Christian Doctrine, 118.
- 26 Johnson, 221.
- 27 Troup, 13.
- 28 Percival Cole, Later Roman Education in Ausonius, Capella, and the Theodosian Code (New York: Columbia University Press, 1909), 16. Quoted in Readings in Medieval Rhetoric, 1.
- 29 On Capella, see: Kennedy, 175–177.
- 30 Joy Connolly, *The State of Speech* (Princeton, NJ: Princeton University Press, 2013), 219.
- 31 Barilli, 42.
- 32 Boethius, *De Topicis Differentiis*, trans. Eleanor Stump (Ithaca, NY: Cornell Press, 1978). See also: Michael C. Leff, "The Logician's Rhetoric: Boethius' *De Differentiis Topicis*, Book IV," in *Medieval Eloquence*, 3–24; also see: Michael C. Leff, "The Topics of Argumentative Invention in Latin Rhetorical Theory from Cicero to Boethius," *Rhetorica* 1, 1 (Spring 1983): 23–44, reprinted in *Rethinking Rhetorical Theory, Criticism, and Pedagogy: The Living Art of Michael C. Leff*, ed. A. de Velasco, J. A. Campbell, and D. Henry (East Lansing, MI: Michigan State University Press, 2016), 65–86.
- 33 Ward, 55.
- 34 Barilli, 43.
- 35 Charles G. Nauert, Jr., *Humanism and the Culture of Renaissance Europe* (Cambridge: Cambridge University Press, 1995), 18.
- 36 Nauert, 18.
- Albrecht Classen, "Female Explorations of Literacy: Epistolary Challenges to the Literary Canon in the Late Middle Ages," in *Disputatio Volume 1: The Late Medieval Epistle*, ed. Carol Poster and Richard Utz (Evanston IL: Northwestern University Press, 1996), 89–122, p. 90. Shirley Kersey, "Medieval Education of Girls and Women," *Educational Horizons* 58(4) (1980): 188–192, p. 188.
- 38 Kersey, 88.
- 39 Albrecht Classen, "Female Exploration of Literacy: Epistolary Challenges to the Literary Canon in the Late Middle Ages," in Poster and Utz, 90.
- 40 Marjory Curry Woods, "The Teaching of Writing in Medieval Europe," in *A Short History of Writing Instruction*, ed. James J. Murphy (Davis, CA: Hermagoras Press, 1990), 80–81.
- In this section I am following Murphy's analysis of later medieval rhetoric as manifested principally in three arts.
- 42 Ward, 44.
- 43 Murphy, Arts, xxiii. Quoted in Vickers, 225.
- 44 Kennedy, 190.
- 45 Ranulf Higden and Margaret Jennings, C. S. J., "The *Ars compendi sermones* of Ranulph Higden," in *Medieval Eloquence: Studies in the Theory and Practice of Medieval Rhetoric*, ed. James J. Murphy, 113.
- 46 Murphy, Arts, 109–216.
- 47 Murphy, *Arts*, 112.

- 48 Murphy, *Arts*, 112–113.
- 49 Murphy, Arts, 114.
- 50 Murphy, *Arts*, 120.
- 51 Murphy, Arts, 124.
- 52 Murphy, Arts, 138.
- 53 Murphy, *Arts*, 128.
- 54 Murphy, Arts, 129.
- 55 Murphy, Arts, 146.
- Disputatio de Rhetorica et de Virtutibus Sapientissimi Regis Carli et Albini Magistri, (The Rhetoric of Alcuin and Charlemagne), trans. Wilbur Samuel Howell (New York: Russell and Russell, 1965).
- 57 Paul Kristeller, "Philosophy and Rhetoric from Antiquity to the Renaissance," part 5 in *Renaissance Thought and Its Sources*, ed. Michael Mooney (New York: Columbia University Press, 1979), 241. Quoted in James J. Murphy and Martin Camargo, "The Middle Ages," Chap. 2 in *The Present State of Scholarship in Historical and Contemporary Rhetoric*, ed. Winifred Bryan Horner (Columbia, SC: University of Missouri Press, 1990), 59.
- Patricia Bizzell and Bruce Herzberg, eds., "The Arts of Letter Writing and Preaching," *The Rhetorical Tradition* (Boston, MA: St. Martin's Press, 1990), 377.
- Nadia Margolis, "The Cry of the Chameleon': Evolving Voices in the Epistles of Christine de Pisan," in *Disputatio: An International Transdisciplinary Journal of* the Late Middle Ages, ed. C. Poster and R. Utz (Evanston, IL: Northwestern University Press, 1996), 37–70.
- Martin Camargo, "Where's the Brief?: The Ars Dictaminis and Reading/Writing between the Lines," in *Disputatio: An International Transdisciplinary Journal of the Late Middle Ages*, ed. C. Poster and R. Utz, vol. 1: The Late Medieval Epistle, 1–17, p. 2. Quotation from Guido Faba's *Summa Dictaminis* (1228–1229) (1996).
- 61 Carol Poster and Richard Utz, eds., *Disputatio: An International Transdisciplinary Journal of the Late Middle Ages* (Evanston, IL: Northwestern University Press), 3.
- 62 Camargo, 4.
- 63 Camargo, 7.
- 64 Murphy, Arts, 3.
- 65 Nicholas Mann, "The Origins of Humanism," in *The Cambridge Companion to Renaissance Humanism*, ed. Jill Kraye (Cambridge: Cambridge University Press, 1996), 5.
- 66 Charles B. Faulhaber, "The *Summa Dictaminis* of Guido Faba," in Murphy, *Medieval Eloquence*, 85.
- 67 Faulhaber, 87.
- 68 Faulhaber, 86.
- 69 Faulhaber, 86.
- 70 Barilli, 49.
- 71 Barilli, 49.
- 72 Camargo, 3.
- 73 Kennedy, 186. (Emphasis added.) See also: Murphy, Arts, 7.
- 74 Murphy, Arts, 17.
- 75 Murphy, *Arts*, 8. This translation of *The Principles of Letter Writing* is by Professor Murphy.
- 76 Murphy, Arts, 11.
- 77 Murphy, Arts, 13-14.

- 78 Murphy, *Arts*, 13.
- 79 Murphy, *Arts*, 15.
- 80 Vickers, 235.
- 81 Vickers, 235–236.
- Faulhaber, 108.
- 83 Classen, 91.
- 84 Malcolm Richardson, "Women, Commerce, and Rhetoric in Medieval England," in *Listening to Their Voices: The Rhetorical Activities of Historical Women*, ed. Molly Meijer Wertheimer (Columbia, SC: University of South Carolina Press, 1997), 133–149, p. 136.
- 85 Richardson, 140.
- 86 Richardson, 145.
- 87 Richardson, 146.
- 88 Richardson, 147.
- 89 Richardson, 146.
- 90 Richardson, 147.
- 91 On women and writing in the Middle Ages, see: John Ward, "Women and Latin Rhetoric from Hortsvit to Hildegard," in *The Changing Tradition: Women in the History of Rhetoric*, ed. Christine Mason and Rebecca Sutcliffe Sutherland (Calgary, AB: University of Calgary Press, 1999), 121–132.
- 92 Kristie S. Fleckenstein, "In the Blood of the Word: Embodied Rhetorical Authority in St. Catherine of Siena's *Dialogue*," in *Sizing up Rhetoric*, ed. David Zarefsky and E. Benacka (Long Grove, IL: Waveland, 2008), 287–295, pp. 287, 294, 289–290.
- 93 Edmond Faral, *Les Artes Poétique du XIIe et XIIIe Siécle* (Paris: Champion, 1924), xv. Ouoted in Vickers, 239.
- 94 Vickers, 239.
- 95 Ernest Gallo, "The *Poetria Nova* of Geoffrey of Vinsauf," in *Medieval Eloquence:* Studies in the Theory and Practice of Medieval Rhetoric, ed. James J. Murphy (Berkely, CA: University of California Press, 1978), 68.
- 96 Murphy, *Arts*, 29.
- 97 Murphy, *Arts*, 30. Vinsauf's *The New Poetics* in Murphy's collection is translated by Jane Baltzell Kopp.
- 98 Murphy, Arts, 34.
- 99 Murphy, Arts, 35.
- 100 Gallo, 71.
- 101 Woods, 84.
- 102 Woods, 86.
- 103 Woods, 87.
- 104 Vickers, 242.
- 105 Gallo, 76.
- 106 Poetria Nova, lines 126–131. Quoted in Gallo, 77.
- 107 Gallo, 77.
- 108 Classen, 91.
- 109 From Joan M. Ferrante, "The French Courtly Poet, Marie de France," in *Medieval Women Writers*, ed. Katharina M. Wilson (Athens: University of Georgia Press, 1984), 65.
- 110 Ferrante, 64.
- 111 Ferrante, 64.

- 112 Ward, 64.
- 113 Ward, 64.

- Brian Vickers, "On the Practicalities of Renaissance Rhetoric," in *Rhetoric Revalued*, ed. Brian Vickers (Binghamton, NY: Center for Medieval and Early Renaissance Studies, 1982b), 133.
- 2 Baldassare Castiglione, *The Courtier* (1528); Gerald P. Mohrmann, "The Civile Conversation: Communication in the Renaissance," *Speech Monographs* 39 (1972): 193–204.
- Charles Nauert, Jr., *Humanism and the Culture of Renaissance Europe* (Cambridge: Cambridge University Press, 1995), 25.
- 4 Vickers, "Practicalities," 135.
- 5 Katharina Wilson, ed., *Women Writers of the Renaissance and Reformation* (Athens: University of Georgia Press, 1987), xxii–xxiii.
- 6 Wilson, xxiii.
- 7 See: Patricia Bizzell and Bruce Herzberg, eds., The Rhetorical Tradition: Readings from Classical Times to the Present (Boston, MA: St. Martin's Press, 2000), 463–498. Bizzell and Herzberg include in their collection works by Christine de Pisan and Laura Cereta.
- 8 Wilson, xxiii. See also in this volume: Anne R. Larsen, "Les Dames des Roches," 232–259.
- 9 George Kennedy, Classical Rhetoric and Its Secular and Secular Traditions (Chapel Hill NC: University of North Carolina Press, 1999), 231.
- 10 Kennedy, 1999, 231.
- 11 Kevin M. Stevens, "Books Fit for a Portuguese Queen: The Lost Library of Catherine of Austria and the Milan Connection (1540)," *Documenting the Early Modern Book World*, ed. Malcolm Walsby and Natasha Constantinidou (Boston, MA: Brill, 2013), 85–116, p. 102.
- 12 A good source on women scholars of the Renaissance is Patricia H. Labalme, ed., *Beyond Their Sex: Learned Women of the European Past* (New York: New York University Press, 1984); and Christopher Lund, Department of Spanish and Portuguese, Brigham Young University, brought de Castro to my attention.
- 13 Jenny R. Redfern, "Christine de Pisan and The Treasure of the City of Ladies: A Medieval Rhetorician and Her Rhetoric," in *Reclaiming Rhetorica: Women in the Rhetorical Tradition*, ed. Andrea A. Lunsford (Pittsburgh, PA: University of Pittsburgh Press, 1995), 73–92, p. 74.
- 14 Redfern, 77.
- 15 Redfern, 74.
- 16 Redfern, 74.
- 17 Redfern, 74.
- 18 Redfern, 78.
- 19 Redfern, 75.
- 20 Redfern, 76.
- 21 Nadia Margolis, "The Cry of the Chameleon: Evolving Voices in the Epistles of Christine de Pizan," *Disputatio: An International Transdisciplinary Journal of the Late Middle Ages* (1996: 1) 37–70, p. 43.

- 22 Kennedy, 1999, 230.
- 23 Don Paul Abbott, "The Renaissance," in *The Present State of Scholarship in Historical and Contemporary Rhetoric*, ed. Winifred Bryan Horner (Columbia, SC: University of Missouri Press, 1990a), 89.
- Nauert, 5.
- Nicholas Mann, "The Origins of Humanism," in *The Cambridge Companion to Renaissance Humanism*, ed. Jill Kraye (Cambridge: Cambridge University Press, 1996), 1.
- Lauro Martines, Jr., *Power and Imagination: City States in Renaissance Italy* (New York: Knopf, 1979), 204.
- 27 Martines, 204.
- 28 Nauert, 18.
- 29 Nauert, 18.
- 30 Nauert, 193.
- 31 Martines, 204.
- 32 Martines, 204.
- 33 Mann, 5.
- 34 Martines, 204.
- 35 Martines, 204.
- 36 Nauert, 5.
- 37 Paul Oskar Kristeller, *Renaissance Thought: The Classic, Scholastic, and Humanist Strains* (New York: Harper and Row, 1961), 13.
- 38 See: John Monfasani, *George of Trebizond: A Biography and a Study of His Rhet-oric and Logic* (Leiden, the Netherlands: Brill, 1976).
- 39 George Kennedy, *Classical Rhetoric in Its Secular and Christian Tradition* (Chapel Hill, NC: University of North Carolina Press, 1999), 232.
- 40 James J. Murphy, ed., *Peter Ramus's Attack on Cicero*, trans. Carole Newlands (Davis, CA: Hermagoras Press, 1992), xv.
- 41 Izora Scott, Controversies over the Imitation of Cicero in the Renaissance (1910; rpt. Davis, CA: Hermagoras Press, 1991), 113.
- 42 Murphy, "Introduction" to Ramus, xxxii.
- 43 Nauert, 25.
- 44 Majid Fakhry, *Averroes: His Life, Work and Influence* (Oxford: Oneworld Publications, 2001), 32.
- 45 Kristeller, 60–63.
- 46 Mann, 9.
- 47 Mann, 9.
- 48 Scott, 7.
- 49 Gene Brucker, *Florence: The Golden Age 1138–1737* (Berkeley, CA: University of California Press, 1969), 202.
- 50 Mann, 5.
- 51 Nauert, 25.
- 52 Jerrold E. Seigel, *Rhetoric and Philosophy in Renaissance Humanism* (Princeton, NJ: Princeton University Press, 1968), 34.
- 53 Martines, 199.
- 54 Euginio Garin, *Portraits of the Quattrocentro* (New York: Harper and Row, 1963).
- 55 Garin, 19.
- Morris Bishop, *Petrarch and His World* (Bloomington, IN: Indiana University Press, 1963), 21.

- 57 Bishop, 21.
- 58 Scott, 7.
- 59 Nauert, 24.
- 60 Mann, 6.
- 61 Nauert, 24.
- Walter J. Ong, S. J., Ramus: Method and the Decay of Dialogue: From the Art of Discourse to the Art of Reason (Cambridge, MA: Harvard University Press, 1983), 137.
- 63 Kristeller, 60-63.
- John G. Burke, "Hermetism as a Renaissance World View," in *The Darker Vision of the Renaissance: Beyond the Fields of Reason*, ed. Robert Kinsman (Berkeley, CA: University of California Press, 1974), 95.
- William A. Covino, Magic, Rhetoric and Literacy: An Eccentric History of the Composing Imagination (Albany, NY: State University of New York Press, 1994), 19.
- 66 Burke, 102.
- 67 Kristeller, 129.
- 68 Pico, Oration on the Dignity of Man. Quoted in Anthony Grafton, "Humanism, Magic and Science," in *The Impact of Humanism on Western Europe*, ed. Anthony Goodman and Angus MacKay (London: Longman, 1990), 111.
- 69 Vickers, "Practicalities," 137.
- 70 Kelley, 35.
- 71 Kelley, 35–36; Vickers, "Practicalities," 266.
- 72 Mack, "Humanist Rhetoric and Dialectic," 86.
- 73 Seigel, 137.
- 74 Kelley, 88.
- 75 Quoted in Seigel, 139.
- 76 Vickers, "Practicalities," 187.
- 77 Seigel writes that Valla "was almost certainly a loyal Christian, even though he may have been a somewhat eccentric one" (145).
- 78 Seigel, 142.
- 79 Seigel, 251.
- 80 Quoted in Michael Mooney, Renaissance Thought and Its Sources (NY: Columbia University Press, 1979), 45.
- 81 Vickers, "Practicalities," 185.
- 82 Donald R. Kelley, *Renaissance Humanism* (Boston, MA: Twayne, 1991), 82.
- 83 Brian Vickers, *In Defense of Rhetoric* (Oxford: Oxford University Press, 1988), 182–183.
- 84 Vickers, In Defense of Rhetoric, 273.
- 85 Vickers, In Defense of Rhetoric, 272.
- 86 Vickers, In Defense of Rhetoric, 271.
- 87 Garin, 19.
- 88 Michael Mooney, *Vico in the Tradition of Rhetoric* (Princeton, NJ: Princeton University Press, 1985), xii.
- 89 Kennedy, 1999, 245.
- James K. Cameron, "Humanism in the Low Countries," in *The Impact of Humanism on Western Europe* (London: Routledge, 1990) 148.
- 91 Cameron, 30.
- 92 Cameron, 148.

- 93 Kennedy, 1999, 246.
- 94 Vickers, "Practicalities," 133.
- 95 Peter Mack, "Humanist Rhetoric and Dialectic," in *The Cambridge Companion to Renaissance Humanism*, ed. Jill Kraye (Cambridge: Cambridge University Press, 1996), 88.
- 96 Don Paul Abbott, "Rhetoric and Writing in Renaissance Europe and England," in A Short History of Writing Instruction, ed. James J. Murphy (Davis, CA: Hermagoras Press, 1990b), 95. See also: Wilbur Samuel Howell, Logic and Rhetoric in England: 1500–1700 (New York: Russell and Russell, 1961).
- 97 Vickers, "Practicalities," 269.
- 98 On Renaissance interest in figures of speech, see: Sylvia Adamson, Gavin Alexander, and Katrin Ettenhuber, eds., *Renaissance Figures of Speech* (Cambridge: Cambridge University Press, 2011).
- 99 Thomas O. Sloane, *On the Contrary: The Protocol of Traditional Rhetoric* (Washington, DC: Catholic University of America Press, 1997), especially Chap. 4, "Disputatiousness."
- 100 Mack, "Humanist Rhetoric and Dialectic," 91.
- 101 Kennedy, 1999, 245.
- 102 Richard Lanham, *A Handbook of Rhetorical Terms*, 2nd ed. (Berkeley, CA: University of California Press, 1991), 80. Lanham quotes Edward O. Wilson, *Biophilia* (Cambridge, MA: Harvard University Press, 1984), 60.
- 103 Lawrence J. Prelli, ed., *Rhetorics of Display* (Columbia, SC: University of South Carolina, 2006), 5.
- 104 For an excellent discussion of Agricola and his influence, see: Ong, Chap. V.
- 105 George Kennedy, Classical Rhetoric and Its Christian and Secular Tradition from Ancient to Modern Times (Chapel Hill, NC: University of North Carolina Press, 1980), 208–210.
- 106 Nauert, 104.
- 107 Kelley, 89.
- 108 Nauert, 104.
- 109 Nauert, 104.
- 110 Agricola, *De Inventione Dialectica* (Nieuwkoop, The Netherlands: De Graaf Publishers, 1967), 192. Quoted in Mack, 86.
- 111 Ong, 104.
- 112 See: Peter Mack, Renaissance Argument: Valla and Agricola in the Tradition of Rhetoric and Dialectic (Leiden, the Netherlands: E. J. Brill, 1993), Chaps. VII and VIII.
- 113 Ong, 93.
- 114 Ong, 93.
- 115 Mack, "Humanist Rhetoric and Dialectic," 89.
- Peter Ramus, *The Questions of Brutus* (1549) trans. Carole Newlands (Davis, CA: Hermagoras Press, 1992), 8.
- 117 Peter Ramus, *Rhetoricae Distinctiones in Quintilianum* (Arguments in Rhetoric against Quintilian), trans. Carole Newlands (1549; DeKalb, IL: Northern Illinois University Press, 1986), 84.
- 118 Ong, 49.
- 119 Richard Lanham, *The Electronic Word: Democracy, Technology and the Arts* (Chicago, IL: University of Chicago Press, 1993), 157–158.
- 120 See: Mack, *Renaissance Argument*, Chap. XVII. For another perspective on rhetoric in this period, see: Mark U. Edwards, Jr., *Printing*, *Propaganda*, *and Martin Luther* (Berkeley, CA: University of California Press, 1994).

- 121 For a detailed discussion of English rhetorical works in the Renaissance, see: Arthur F. Kinney, *Humanist Poetics: Thought, Rhetoric, and Fiction in Sixteenth-Century England* (Amherst, MA: University of Massachusetts Press, 1986).
- See: Garry Wills, *Rome and Rhetoric: Shakespeare's Julius Caesar* (New Haven, CT: Yale University Press, 2011).
- 123 Kennedy, 1999, 247.
- Ryan Stark, *Rhetoric*, *Science and Magic in Seventeenth-Century England* (Washington, DC: Catholic University Press, 2009).
- 125 Stark, 83.
- 126 Stark, 68.
- Jane Donawerth, ed., Editor's Introduction to *Rhetorical Theory by Women before* 1900 (Lanham, MD: Rowen and Littlefield, 2002), xxii–xxiii.
- 128 Donawerth, xxiii.
- 129 Donawerth, xl.
- 130 Donawerth, xl.

- 1 Barbara Wamick, *The Sixth Canon: Belletristic Rhetorical Theory and Its French Antecedents* (Columbia, SC: University of South Carolina Press, 1993), 129.
- 2 Ryan John Stark, "Margaret Cavendish and Composition Style," *Rhetoric Review* 17(2) (Spring 1999), 264–281.
- 3 Sarah Heller Mendelson, *The Mental World of Stuart Women: Three Studies* (Amherst, MA: University of Massachusetts Press, 1987), 35.
- 4 On Vico's life and relationship to rhetoric, see: Michael Mooney, *Vico in the Tradition of Rhetoric* (Princeton, NJ: Princeton University Press, 1985).
- 5 Peter Burke, Vico (Oxford: Oxford University Press, 1985), 2.
- 6 Burke, 55.
- 7 Burke, 3.
- 8 Ernesto Grassi, *Rhetoric as Philosophy* (University Park, PA: Pennsylvania State Press, 1980), 5.
- 9 Grassi, 6.
- 10 Mooney, 126.
- 11 Mooney, 8–9.
- 12 Mooney, 127.
- 13 Grassi, 7.
- 14 Mooney, 135.
- 15 Grassi, 8.
- 16 Mooney, 29.
- Giambattista Vico, On the Study Methods of Our Time, trans. Elio Gianturco (Ithaca, NY: Cornell University Press, 1990), 87. Quoted in Donald Philip Verene, "Gadamer and Vico on Sensus Communis and the Tradition of Humane Knowledge," in The Philosophy of Hans-George Gadamer, ed. Lewis E. Hahn (Chicago, IL: Open Court Publishers, 1997), 140.
- 18 Richard Manson, *The Theory of Knowledge of Giambattista Vico* (n.p.: Archon Books, 1969), 35.
- 19 Katherine E. Gilbert and Helmut Kuhn, *A History of Esthetics* (Bloomington, IN: University of Indiana Press, 1953), 268. Quoted in Willson Havelock Coates, *The Emergence of Liberal Humanism* (New York: McGraw-Hill, 1966), 219.

- 20 Burke, 53.
- 21 Burke, 53.
- 22 Joseph Mali, *The Rehabilitation of Myth: Vico's "New Science"* (Cambridge: Cambridge University Press, 1992), 183; Mooney, 66.
- 23 Mooney, 66.
- 24 Mali, 183.
- 25 Mali, 167.
- Vico, "Orazione in Morte di Donna Angela Cimmino Marchesa di Petrella," in *Opere di G. B. Vico*, ed. Fausto Nicolini, 8 vols. in 11 (Rome: Bari Laterza, 1911–1914), 7:170. Quoted in Grassi, 7.
- 27 Paul Hazzard, *La Pensée Européenne au XVIIIe siècle de Montesquieu Lessing* (Paris: Arthone Fayard, 1963), 43. Quoted in editor's introduction to Giambattista Vico, *On Humanistic Education: Six Inaugural Orations, 1699–1707*, ed. Donald Phillip Verene, trans. G. A. Pinton and A. W. Shippe (Ithaca, NY: Cornell University Press, 1993), 2–3.
- 28 Mali, 7.
- 29 Laurence Coupe, Myth (New York: Routledge, 1997), 119.
- 30 Mali, 3.
- 31 Mali, 150. Benjamin Bennett writes that in *The Birth of Tragedy*, Nietzsche asserts that myth is "unconscious metaphysics." "Nietzsche's Idea of Myth: The Birth of Tragedy from the Spirit of Eighteenth-Century Aesthetics," *Proceedings of the Modern Language Association* 91: 3 (May, 1979), 420–422, p. 421.
- 32 Mali, 9.
- 33 Verene, 83.
- 34 Verene, 138.
- 35 Hayden White, *Tropics of Discourse: Essays in Cultural Criticism* (Baltimore, MD: Johns Hopkins University Press, 1986).
- Wilbur Samuel Howell, *Eighteenth-Century British Logic and Rhetoric* (Princeton, NJ: Princeton University Press, 1971), 6.
- Winifred Bryan Horner, "Writing Instruction in Great Britain: Eighteenth and Nineteenth Centuries," in *A Short History of Writing Instruction*, ed. James J. Murphy (Davis, CA: Hermagoras Press, 1990), 124–125.
- 38 Stephen H. Browne, "Shandyean Satire and the Rhetorical Arts in Eighteenth-Century England," *Southern Communication Journal* 55 (Winter 1990): 191–205.
- 39 Warnick, 66. Warnick cites Richard Sher on this point.
- 40 Horner, 136–137.
- 41 Horner, 121–150.
- 42 Kenneth Cmiel, Democratic Eloquence: The Fight over Popular Speech in Nineteenth-Century America (New York: William Morrow, 1990), 36.
- 43 See: Howell, Eighteenth-Century, 145–256.
- 44 G. P. Mohrmann, introduction to Thomas Sheridan, *A Discourse Being Introductory to His Course of Lectures on Elocution and the English Language* (1759; rpt. Los Angeles: Augustan Reprint Society, 1969), ii.
- 45 Quoted in Howell, Eighteenth-Century, 155.
- 46 Quoted in Howell, Eighteenth-Century, 156.
- 47 Sheridan, A Discourse, 25–26.
- 48 Thomas Sheridan, *A Course of Lectures on Elocution* (London: W. Strahan, 1762). On Sheridan's work, see: Wallace A. Bacon, "The Elocutionary Career of Thomas Sheridan," *Speech Monographs* 31 (March 1964): 1–53.

- 49 Thomas Sheridan, *A General Dictionary of the English Language* (London: J. Dodsley, 1780). This work was published to provide a guide to proper pronunciation of English words.
- 50 Cmiel, 40.
- Hugh Blair, *Lectures on Rhetoric and Belles Lettres*, ed. Harold Harding, 2 vols. (Carbondale, IL: Southern Illinois University Press, 1965), v. 1, x–xi.
- 52 Warnick, especially Chap. 1.
- 53 Warnick, 4.
- 54 Warnick, 34.
- 55 Herman Cohen, "Belles-lettres," in Encyclopedia of Composition and Rhetoric: Communication from Ancient Times to the Information Age, ed. Theresa Enos (New York: Routledge, 2004), 71.
- 56 Richard Whately, *Elements of Rhetoric*, ed. Douglas Ehninger (1828, rpt. Carbondale, IL: Southern Illinois University Press, 1963), xxiv.
- 57 Henry Home, Lord Kames, *The Elements of Criticism* (1761; rpt. New York: Barnes and Burr, 1865), 133.
- 58 Kames, 270.
- 59 Warnick, 112.
- 60 Blair, 21.
- 61 Blair, 183.
- 62 Blair, 183–184.
- 63 Blair, 184.
- 64 Blair, 185–186.
- 65 Blair, 226.
- See: James L. Golden and Edward P. J. Corbett, The Rhetoric of Blair, Campbell, and Whately (New York: Holt, Rinehart and Winston, 1968). Other sources include: Douglas Ehninger, "George Campbell and the Revolution in Inventional Theory," Southern Speech Journal 15 (1950): 270–276; "Campbell, Blair, and Whately Revisited," Southern Speech Journal 28 (1963), 169–182; "Campbell, Blair, and Whately: Old Friends in a New Light," Western Speech 19 (1955): 263–269; Douglas McDermott, "George Campbell and the Classical Tradition," Quarterly Journal of Speech 49 (1963): 403–409; Dominic LaRusso, "Root or Branch? A Reexamination of Campbell's 'Rhetoric," Western Speech 32 (1968): 85–91; Herman Cohen, "William Leechman's Anticipation of Campbell," Western Speech 32 (1968), 92–99.
- 67 George Campbell, *The Philosophy of Rhetoric*, ed. Lloyd F. Bitzer (Carbondale, IL: University of Southern Illinois Press, 1963). On Campbell, see: Bitzer's Introduction to *The Philosophy of Rhetoric*; Howell, *Eighteenth-Century*, 577–612.
- On Campbell's thought and work, see: Arthur Walzer, *George Campbell: Rhetoric in the Age of Enlightenment* (Albany NY: State University of New York Press, 2003).
- 69 Gerard Hauser, "Empiricism, Description, and the New Rhetoric," *Philosophy and Rhetoric* 5 (1972): 24–44; Lloyd Bitzer, "Hume's Philosophy in George Campbell's *The Philosophy of Rhetoric*," *Philosophy and Rhetoric* 2 (1969): 136–166; Vincent Bevilaqua, "Philosophical Origins of George Campbell's *Philosophy of Rhetoric*," *Speech Monographs* 32 (1965): 7–8.
- 70 Bitzer, Introduction to The Philosophy of Rhetoric, xiii.
- 71 Bitzer, Introduction to *The Philosophy of Rhetoric*, xiii.
- 72 Campbell, xlv.
- 73 George Kennedy, Classical Rhetoric and Its Christian and Secular Tradition from Ancient Times, 1st ed. (Chapel Hill, NC: University of North Carolina Press, 1980),

- 234; see also: Thomas Conley, *Rhetoric in the European Tradition* (New York: Longman, 1990), 217–220.
- Douglas Ehninger, Introduction to Richard Whately, *Elements of Rhetoric* (1828; Carbondale, IL: Southern Illinois University Press, 1963), xxv–xxvi.
- 75 Kennedy, 240.
- 76 Wilbur Samuel Howell, "John Locke and the New Rhetoric," *Quarterly Journal of Speech* 53 (1967): 319–321.
- 77 Campbell, 1.
- 78 Campbell, 1.
- 79 Warnick, 68.
- 80 Quoted in Howell, Eighteenth-Century, 587.
- 81 Campbell, 77.
- 82 Quoted in Howell, Eighteenth-Century, 587.
- 83 Warnick, 67.
- 84 Warnick, 67.
- 85 Campbell, 108.
- 86 Campbell, 108.
- 87 Wamick, 119.
- 88 McKerrow, 7 ff.
- 89 McKerrow, 9.
- 90 Whately, 90.
- 91 Whately, 91.
- 92 Whately, 112.
- 93 Whately, 112–113.
- 94 Whately, 116.
- Jane Donawerth, "Poaching on Men's Philosophy of Rhetoric: Eighteenth- and Nine-teenth-Century Rhetorical Theory by Women," *Philosophy and Rhetoric*, 33(3) (2000), 243–258.
- 96 Donawerth, 244.
- 97 Donawerth, 245.
- 98 Donawerth, 246.

- David A. Frank, "A Traumatic Reading of Twentieth-Century Rhetorical Theory: The Belgian Holocaust, Malines, Perelman, and de Man," *Quarterly Journal of Speech* 93(3) (August 2007): 308–343.
- For a good collection of essays on Perelman's rhetorical theory, see: Ray D. Dearin, ed., *The New Rhetoric of Chaim Perelman* (Lanham, MD: University Press of America, 1989).
- 3 Chaim Perelman and L. Olbrechts-Tyteca, *The New Rhetoric: A Treatise on Argumentation*, trans. John Wilkinson and Purcell Weaver (Notre Dame, IN: University of Notre Dame Press, 1969), 6.
- 4 Perelman and Olbrechts-Tyteca, 20.
- 5 Perelman and Olbrechts-Tyteca, 4.
- 6 Perelman and Olbrechts-Tyteca, 14.
- On Perelman and Olbrechts-Tyteca's theory of audience, see: Alan Gross, "A Theory of the Rhetorical Audience: Reflections on Chaim Perelman," *Quarterly Journal of Speech* 85 (May 1999): 202–211.

- 8 Perelman and Olbrechts-Tyteca, 30.
- 9 If you would like to read more about the concept of the universal audience, see: John R. Anderson, "The Audience as a Concept in the Philosophic Rhetoric of Perelman, Johnstone, and Natanson," *Southern Speech Communication Journal* 38 (Fall 1972): 39–50; Lisa S. Ede, "Rhetoric vs. Philosophy: The Role of the Universal Audience in Chaim Perelman's *The New Rhetoric*," *Central States Speech Journal* 32 (Summer 1981): 118–125; John W. Ray, "Perelman's Universal Audience," *Quarterly Journal of Speech* 64 (December 1978): 361–375; and Allen Scult, "Perelman's Universal Audience: One Perspective," *Central States Speech Journal* 27 (Fall 1976): 176–180.
- 10 Perelman and Olbrechts-Tyteca, 30.
- 11 Chaim Perelman, *The New Rhetoric and the Humanities*, trans. William Kluback (Dordrecht, Holland: D. Reidel, 1979), 46.
- 12 Perelman and Olbrechts-Tyteca, 20–21.
- 13 Perelman and Olbrechts-Tyteca, 65.
- 14 Perelman and Olbrechts-Tyteca, 65.
- 15 Perelman and Olbrechts-Tyteca, 66.
- 16 Chaim Perelman, Justice (New York: Random House, 1967), 78.
- 17 Perelman, Justice, 78.
- 18 Perelman, Justice, 82.
- 19 Perelman, Rhetoric and Humanities, 118.
- 20 Perelman and Olbrechts-Tyteca, 31.
- 21 Perelman and Olbrechts-Tyteca, 36.
- 22 Perelman and Olbrechts-Tyteca, 36.
- 23 Perelman and Olbrechts-Tyteca, 37.
- 24 Perelman and Olbrechts-Tyteca, 116–117.
- Hubert L. Dreyfus and Paul Rabinow, "What Is Maturity? Habermas and Foucault on 'What Is Enlightenment?" in *Foucault: A Critical Reader*, ed. David Couzens Hoy (Oxford: Basil Blackwell, 1986), 110.
- 26 Dreyfus and Rabinow, 110. See also: Jürgen Habermas, *Toward a Rational Society: Student Protest, Science and Politics*, trans. J. Shapiro (Boston, MA: Beacon Press, 1970).
- 27 Jürgen Habermas, Communication and the Evolution of Society, trans. Thomas McCarthy (Boston, MA: Beacon Press, 1979), 7.
- See: Jürgen Habermas, *Knowledge and Human Interests*, trans. Jeremy J. Shapiro (Boston, MA: Beacon Press, 1971); and *Theory and Practice*, trans. John Viertel (Boston, MA: Beacon Press, 1973).
- 29 McCarthy in editor's introduction to *The Philosophical Discourse of Modernity*, ed. Thomas McCarthy (Cambridge MA: MIT Press, 1987), xii.
- 30 Habermas, Toward a Rational Society, 74–75.
- 31 Susan Wells, *Sweet Reason: Rhetoric and the Discourses of Modernity* (Chicago, IL: University of Chicago Press, 1996), 114.
- 32 Jürgen Habermas, The Lure of Technocracy (Cambridge: Polity Press, 2016), 48.
- Jürgen Habermas, *The Philosophical Discourse of Modernity: Twelve Lectures*, trans. Frederick Lawrence (Cambridge, MA: MIT Press, 1987), ed. Thomas McCarthy, x.
- 34 Thomas McCarthy, introduction to Habermas, *Philosophical Discourse*, x.
- Jürgen Habermas, *Theory and Practice*, trans. John Viertel (Boston, MA: Beacon Press, 1973), 25.
- 36 McCarthy in editor's introduction to *Philosophical Discourse*, x.

- 37 Wells, 115.
- Translator's introduction to Jürgen Habermas, *Truth and Justification*, trans. Barbara Fultner (Cambridge, MA: MIT Press, 2003), xvii.
- 39 Wells, 103.
- 40 McCarthy in editor's introduction to *Philosophical Discourse*, xvi.
- 41 Thomas McCarthy, *The Critical Theory of Jürgen Habermas* (Cambridge, MA: MIT Press, 1978), 292.
- 42 Wells, 119.
- 43 Sonja K. Foss, Karen A. Foss, and Robert Trapp, *Contemporary Perspective on Rhetoric*, 2nd ed. (Prospect Heights, IL: Waveland, 1991), 241–272.
- 44 McCarthy, Critical Theory, 273.
- 45 See, for example: Dick Howard, "A Politics in Search of the Political," *Theory and Society* 1 (Fall 1974): 271–306.
- 46 Wells, 116.
- 47 Joy Connolly, *The State of Speech: Rhetorical and Political Thought in Ancient Rome* (Princeton, NJ: Princeton University Press, 2013), 9.
- 48 Marie Fleming, *Emancipation, and Illusion: Rationality and Gender in Habermas' Theory of Modernity* (University Park, PA: Penn State University Press, 1999), 86.
- 49 Herbert Simons, ed., *The Rhetorical Turn: Invention and Persuasion in the Conduct of Inquiry* (Chicago, IL: University of Chicago Press, 1990), 8.
- See, for example: Thomas Kuhn, The Structure of Scientific Revolutions, 3rd ed. (Chicago, IL: University of Chicago Press, 1996); Evelyn Fox Keller, "Secrets of Life, Secrets of Death," in The Rhetoric of the Human Sciences, ed. J. Nelson, A. Megill and D. McCloskey (Madison, WI: University of Wisconsin Press, 1987); Charles Bazerrman, Shaping Written Knowledge: The Genre and Activity of the Experimental Article in Science (Madison, WI: University of Wisconsin Press, 1988); Lawrence Prelli, A Rhetoric of Science: Inventing Scientific Discourse (Columbia, SC: University of South Carolina Press, 1989); Alan Gross, The Rhetoric of Science (Cambridge, MA: Harvard University Press, 1990); Randy Alan Harris, ed., Landmark Essays in the Rhetoric of Science (London: Routledge, 1997; 2nd ed. 2017).
- 51 Brian Boyd, *On the Origins of Stories* (Cambridge, MA: Harvard University Press, 2010), 411.
- 52 See: Karl Popper, Conjectures and Refutations: The Growth of Scientific Knowledge (London: Routledge, 1963).
- 53 Mary Midgley, Evolution as a Religion (London: Routledge, 1985), 30.
- Mary Midgley, *Science as Salvation: A Modern Myth and Its Meaning* (London and New York: Routledge, 1992).
- 55 Mary Midgley, *The Myths We Live By* (New York: Routledge, 2004), xii–xiii.
- Midgley, Myths, xii. Segal notes Karl Popper's related idea: "Popper even maintains that scientific theories remain myth-like, for theories, like myths, can never be proved, only disproved, and therefore 'remain essentially uncertain or hypothetical," 34.
- 57 Midgley, Myths, xii.
- 58 Mary Midgley, Salvation, 57.
- 59 Midgley, Salvation, 13.
- John Lyne, "Rhetoric and Scientific Communities," in *Rhetoric and Community: Studies in Unity and Fragmentation*, ed. Michael J. Hogan (Columbia, SC: University of South Carolina Press, 1998), 266.

- 61 Wells, 93.
- 62 Charles Bazerman, Shaping Written Knowledge: The Genre and Activity of the Experimental Article in Science (Madison, WI: University of Wisconsin Press 1988), 321.
- 63 See: Kenneth Gergen and Mary Gergen, "Narrative Form and the Construction of Psychological Sciences," in *Narrative Psychology*, ed. T. R. Sarbin (New York: Praeger Press, 1986), 22–44; and Lisa J. Disch, "More Truth than Fact: Storytelling as Critical Understanding in the Writings of Hannah Arendt," *Political Theory* (1994): 665–694.
- 64 Charles Arthur Willard, "Argumentation and Postmodern Critique," in *Perspectives on Argumentation: Essays in Honor of Wayne Brockriede*, ed. Robert Trapp and Janice E. Schuetz (Prospect Heights, IL: Waveland, 1990), 221–231, p. 225.
- 65 Wells, 56.
- 66 See, for example: Stephen B. Katz, "The Ethic of Expediency: Classical Rhetoric, Technology, and the Holocaust," *College English* 54 (March 1992): 255–275.
- 67 Wells, 56.
- 68 Simons, 4.
- 69 Lyne, 268.
- 70 Lyne, 268–269.
- Alan Gross, Joseph Harmon, and Michael Reidy, Communicating Science: The Scientific Article from the seventeenth Century to the Present (Oxford, UK: Oxford University Press, 2002), viii. See also: Alan G. Gross, Starring the Text: The Place of Rhetoric in Science Studies (Carbondale, IL: Southern Illinois University Press, 2006).
- 72 Lyne, 272–273.
- 73 Lyne, 273.
- 74 Lyne, 275.
- 75 Deirdre N. McCloskey, *The Rhetoric of Economics* (Madison, WI: University of Wisconsin Press, 1985; 2nd ed. 1998), 36. Quoted in Herbert W. Simons, "The Rhetoric of Inquiry as an Intellectual Movement," *The Rhetorical Turn*, ed. Herbert W. Simons (Chicago, IL: University of Chicago Press, 1990), 1–31, p. 9.
- 76 Deirdre N. McCloskey, *Bourgeois Dignity Why Economics Can't Explain the Modern World* (Chicago, IL: University of Chicago Press, 2011), 40.
- 77 John Van Maanen, *Tales of the Field: On Writing Ethnography* (Chicago, IL: University of Chicago Press, 1988), 5.
- 78 Jeanne Fahnestock, *Rhetorical Figures in Science* (New York: Oxford University Press, 1999).
- 79 Clifford Geertz, *Works and Lives: The Anthropologist as Author* (Stanford, CA: Stanford University Press, 1988), 4–5.
- 80 Geertz, 84. For other views of rhetoric's role in anthropology, see: Renato Rosaldo, "Where Objectivity Lies," in *The Rhetoric of the Human Sciences*, 87–110; Christopher Nash and Martin Warner, eds., *Narrative in Culture* (London: Routledge, 1988); and Alessandro Duranti, *From Grammar to Politics: Linguistic Anthropology in a Western Samoan Village* (Berkeley, CA: University of California Press, 1994).
- See: John Angus Campbell, "Darwin, Thales, and the Milkmaid: Scientific Revolution and Argument from Common Beliefs and Common Sense," in *Perspectives on Argument*, ed. Robert Trapp and Janice Schuetz (Prospect Heights, IL: Waveland, 1990a), 207–220; and John Angus Campbell, "Scientific Discovery and Rhetorical Invention: The Path of Darwin's Origin," *The Rhetorical Turn*, 58–90.

- 82 Campbell, "Darwin, Thales," 209.
- 83 Campbell, "Darwin, Thales," 209.
- 84 Campbell, "Darwin, Thales," 210.
- 85 Campbell, "Darwin, Thales," 213.
- 86 Campbell, "Darwin, Thales," 214.
- 87 Campbell, "Darwin, Thales," 215.
- 88 Campbell, "Darwin, Thales," 216.
- 89 Dilip P. Gaonkar, "The Idea of Rhetoric in the Rhetoric of Science," Southern Communication Journal, 58(4) (Summer, 1993): 255–327, pp. 258–259. Also published in Alan G. Gross and William M. Keith, eds., Rhetorical Hermeneutics: Invention and Interpretation in the Age of Science (Albany, NY: State University of New York Press, 1997).
- 90 Gaonkar, 259.
- 91 Gaonkar, 266.
- 92 Gaonkar, 267.
- 93 Gaonkar, 267.
- David J. Depew and John Lyne, "The Productivity of Scientific Rhetoric," *POROI:* An Interdisciplinary Journal of Rhetorical Analysis and Invention, 9(1) (2013): 1–21, pp. 1–2.

CHAPTER 10

- James L. Kinneavy, "Contemporary Rhetoric," in *The Present State of Scholarship in Historical and Contemporary Rhetoric*, ed. Winifred Bryan Horner (Columbia, SC: University of Missouri Press, 1990), 192.
- 2 Michael C. Leff, "What Is Rhetoric?" *Rethinking Rhetorical Theory, Criticism, and Pedagogy: The Living Art of Michael C. Leff*, eds. A. de Velasco, J. A. Campbell, and D. Henry (East Lansing, MI: Michigan State University Press, 2016), 471–481, p. 476.
- 3 Kenneth Burke's numerous works include: Counter-Statement (Berkeley, CA: University of California Press, 1968); Language as Symbolic Action (Berkeley, CA: University of California Press, 1966); A Grammar of Motives (Berkeley, CA: University of California Press, 1969); A Rhetoric of Motives (Berkeley, CA: University of California Press, 1969); and The Rhetoric of Religion (Berkeley, CA: University of California Press, 1961).
- 4 See, for example: Don Abbott, "Marxist Influences on the Rhetorical Theory of Kenneth Burke," *Philosophy and Rhetoric* 4 (1974): 217–233.
- 5 Sidney Hook, "The Technique of Mystification," *Critical Responses to Kenneth Burke*, ed. William H. Ruckert (Minneapolis, MN: University of Minnesota Press, 1969), 89. Quoted in Abbott, 217.
- 6 George Kennedy, *Classical Rhetoric and Its Christian and Secular Tradition* (Chapel Hill, NC: University of North Carolina Press, 1999), 94.
- 7 Kenneth Burke, *A Rhetoric of Motives* (Berkeley, CA: University of California Press, 1969), 55.
- 8 Lloyd Bitzer, "Political Rhetoric," in *Landmark Essays on Contemporary Rhetoric*, ed. Thomas Farrell (Mahwah, NJ: Lawrence Earlbaum, 1998), 1–22, p. 9.
- 9 Burke, Rhetoric of Motives, 22.
- 10 Bitzer, "Political Rhetoric," 9-10.
- 11 Burke, Rhetoric of Motives, xiv.

- 12 Burke used the Latin phrase, ad purificandum bellum.
- 13 Thomas Farrell, *Norms of Rhetorical Culture* (New Haven, CT: Yale University Press, 1995), 204.
- 14 Burke, *Rhetoric of Motives*, 43.
- 15 Burke, Language as Symbolic Action, 16.
- 16 Burke, Language as Symbolic Action, 45.
- 17 Burke, Language as Symbolic Action, 45.
- 18 Burke, Language as Symbolic Action, 45.
- 19 Burke, Language as Symbolic Action, 50.
- 20 Burke, A Grammar of Motives (Berkeley, CA: University of California Press, 1969).
- 21 Burke, Grammar of Motives, xvi.
- 22 Burke, Grammar of Motives, 15.
- 23 Richard Lanham, *The Electronic Word: Democracy, Technology and the Arts* (Chicago, IL: University of Chicago Press, 1993), 56.
- Burke, *The Philosophy of Literary Form: Studies in Symbolic Action*, 3rd ed. (1941; Berkeley, CA: University of California Press, 1973). Also: *Counter-Statement*.
- 25 Burke, *Counter-Statement* (1931; Berkeley, CA: University of California Press, 1968), 124.
- 26 Burke, Counter-Statement, 124.
- 27 Burke, Counter-Statement, 124.
- 28 Burke, Counter-Statement, 125.
- 29 Burke, Counter-Statement, 126.
- 30 Brian Vickers, In Defense of Rhetoric (Oxford: Oxford University Press, 1988), 441.
- William A. Covino, *Magic, Rhetoric and Literacy: An Eccentric History of the Composing Imagination* (Albany, NY: State University of New York Press, 1994), 93.
- 32 Burke, *The Philosophy of Literary Form*, 4. Quoted in Covino, 91.
- 33 Lloyd Bitzer, "The Rhetorical Situation," *Philosophy and Rhetoric* 1 (December 1968), 1–14.
- 34 Bitzer, "The Rhetorical Situation," 3.
- 35 Bitzer, "The Rhetorical Situation," 6.
- 36 Bitzer, "The Rhetorical Situation," 7.
- 37 Bitzer, "The Rhetorical Situation," 8.
- 38 Bitzer, "The Rhetorical Situation," 8.
- 39 Bitzer, "The Rhetorical Situation," 10.
- Sources on rhetoric and narration include: Anne Dipardo, "Narrative Knowers, Expository Knowledge," Written Communication 7 (January 1990): 59–95; Doug Hesse, "Aristotle's Poetics and Rhetoric: Narrative as Rhetoric's Fourth Mode," in Rebirth of Rhetoric: Essays in Language, Culture, and Education, ed. Richard Andrews (London: Routledge, 1992); Richard Andrews, ed., Narrative and Argument (Milton Keynes, UK: Open University Press, 1989); and Renato Rosaldo, "Where Objectivity Lies," in The Rhetoric of the Human Sciences, ed. John Nelson, Allen Megill, and Donald McCloskey (Madison, WI: University of Wisconsin Press, 1987), 87–110.
- 41 Susan Wells, Sweet Reason: Rhetoric and the Discourses of Modernity (Chicago, IL: University of Chicago Press, 1996), 49.
- 42 Wells, 49.
- 43 On Bakhtin's thought, see: Michael Holquist, *Dialogism: Bakhtin and His World* (London: Routledge, 1990).
- 44 Holquist, 15.

- 45 Alastair Renfrew, Mikhail Bakhtin (London: Routledge, 2015), 7–8.
- 46 Renfrew, 23.
- 47 Mikhail Bakhtin, *Marxism and the Philosophy of Language*, trans. L. Matejka and I. R. Titunik (New York: Seminar Press, 1973), 9. See also: Liisa Steinby and Tintti Klapuri, eds., *Bakhtin and His Others: Intersubject(ivism), Chronotype*, *Dialogism* (London, UK: Anthem, 2013).
- 48 Bakhtin, Marxism, 86.
- 49 Charles Schuster, as quoted in editor's introduction to *Landmark Essays on Bakhtin*, *Rhetoric*, *and Writing*, ed. Frank Farmer (Mahwah, NJ: Hermagoras Press, 1998), 3.
- Josephine Donovan in *Feminism*, *Bakhtin*, *and the Dialogic*, ed. Dale M. Bauer and S. Jaret McKinstry (Albany, NY: State University of New York, 1991), 86.
- Wayne E. Booth, Introduction to Mikhail Bakhtin, *Problems of Dostoevsky's Poetics*, ed. and trans. Caryl Emerson (Minneapolis, MN: University of Minnesota Press, 1984), xix.
- 52 Wayne E. Booth, Introduction to Bakhtin, *Dostoevsky*, xx.
- 53 Bakhtin, Dostoevsky, xxv.
- 54 Bakhtin, Dostoevsky, 3.
- 55 Bakhtin, Dostoevsky, 6.
- Wayne Booth, The Rhetoric of Fiction (Chicago, IL: University of Chicago Press, 1961), 16. See also: Now Don't Try to Reason with Me (Chicago, IL: University of Chicago Press, 1970); Modern Dogma and the Rhetoric of Assent (Chicago, IL: University of Chicago Press, 1974).
- 57 Booth, Fiction, 20.
- 58 Booth, Fiction, 50.
- 59 Booth, Fiction, 60.
- 60 Booth, Fiction, 69.
- 61 Booth, Fiction, 78.
- 62 Brian Boyd, *On the Origin of Stories* (Cambridge, MA: Harvard University Press, 2010), 166.
- 63 Boyd, 411.
- 64 Walter Fisher, *Human Communication as Narration: Toward a Philosophy of Reason, Value, and Action* (Columbia, SC: University of South Carolina Press, 1987).
- 65 Fisher, 108–109.
- 66 Fisher, 95.
- 67 Fisher, 105.
- 68 Fisher, 137.
- 69 See: Ernest Bormann, "Fetching Good Out of Evil: A Rhetorical Use of Calamity," *Quarterly Journal of Speech*, 63 (1977): 130–139; "A Fantasy Theme Analysis of the Television Coverage of the Hostage Release and the Reagan Inauguration," *Quarterly Journal of Speech* 68 (1982): 133–145; *Restoring the American Dream* (Carbondale, IL: Southern Illinois University Press, 1985).
- 70 Ernest Bormann, "Fantasy and Rhetorical Vision: The Rhetorical Criticism of Social Reality," *Quarterly Journal of Speech* 59 (1972): 398.
- 71 Bormann, "Fantasy and Rhetorical Vision," 398.
- 72 Ernest C. Bormann, "Symbolic Convergence: Organizational Communication and Culture," in *Communication and Organizations: An Interpretive Approach*, ed. L. L. Putnam and M. E. Pacanowsky (Newbury Park: Sage, 1983), 99–122.
- 73 Bormann, "Symbolic Convergence," 102.

- 74 Bormann, "Symbolic Convergence," 108.
- 75 Bormann, "Symbolic Convergence," 120.
- 76 See: Carole Blair, Places of Public Memory: The Rhetoric of Memorials and Museums (Tuscaloosa, AL: University of Alabama Press, 2010); and Lester Olson, Benjamin Franklin's Vision of American Community (Columbia, SC: University of South Carolina Press, 2004).
- 77 Lawrence J. Prelli, ed., *Rhetorics of Display* (Columbia, SC: University of South Carolina, 2006), 1.
- 78 Prelli, 2.
- 79 See, for example: Laurie E. Gries, *Still Life with Rhetoric: A New Materialist Approach for Visual Rhetorics* (Boulder, CO: University Press of Colorado, 2015).
- 80 Scot Barnett and Casey Boyle, eds., *Rhetoric*, *through Everyday Things* (Tuscaloosa, AL: University of Alabama Press, 2016), 1. See also: Mel Y. Chen, *Animacies: Biopolitics, Racial Mattering and Queer Affect* (Durham, NC: Duke University Press, 2012).
- Bruno Latour, *We Have Never Been Modern* (Cambridge, MA: Harvard University Press, 1991). For a feminist application of Latour's work to questions in science, see: Donna J. Haraway, *Staying with the Trouble: Making Kin in the Chthulucene* (Durham, NC: Duke University Press, 2016).
- Laurie Gries, Still Life with Rhetoric: A New Materialist Approach for Visual Rhetorics (Logan, UT: Utah State University Press, 2015), xv.
- 83 Gries, 4.
- 84 Gries, 5.
- 85 Gries, 7.
- 86 Gries, 12.
- 87 Gries, 16.
- 88 Gries, 86.
- 89 Scot Barnett and Casey Boyle, *Rhetoric, through Everyday Things*, 2nd ed. (Tuscaloosa, AL: University of Alabama Press, 2016), 1.
- 90 Barnett and Boyle, 1.
- 91 Barnett and Boyle, 2.
- 92 Nicholas Carr, *The Shallows: What the Internet Is doing to Our Brains* (New York: Norton, 2011), 104. See also by Carr: *The Glass Cage: How Our Computers Are Changing Us* (New York: W.W. Norton, 2015), and *Utopia Is Creepy: And Other Provocations* (New York: W. W. Norton, 2016).
- 93 Zeynep Tufekci, March 10, 2018.
- 94 See, for example: Maria Lindh and Jan Nolin, "Information We Collect: Surveillance and Privacy in the Implementation of Google Apps for Education," *European Educational Research Journal* (2016): 1–20.
- 95 Clay Shirky, "Cleaning up Online Conversation," *Harvard Business Review: HBR Agenda 2011* (January 2011), http://hbr.org/web/extras/hbragenda-2011/clay-shirky (accessed June 5, 2011). See also Shirky's books on the online experience: *Here Comes Everybody: The Power of Organizing without Organizations* (New York: Penguin, 2009), and *Cognitive Surplus: Creativity and Generosity in a Connected Age* (New York: Penguin, 2010).
- Olive Thompson, "Clive Thompson on the New Literacy," *Wired* (August 24, 2009), www.wired.com/techbiz/people/magaz 4.ine/17-09/st_thompson (accessed June 4, 2011).
- 97 Douglas Eyman, *Digital Rhetoric: Theory, Method, Practice* (Ann Arbor, MI: University of Michigan Press, 2015), 1, 13

- 98 Eyman, 17.
- 99 Eyman 44.

CHAPTER 11

- 1 For a good survey of these developments, see: Ann Gill, *Rhetoric and Human Understanding* (Prospect Heights, IL: Waveland, 1994). See also: Charles C. Lemert and Garth Gillan, *Michel Foucault: Social Theory as Transgression* (New York: Columbia University Press, 1982), Chap. 1, "Foucault's Field."
- 2 Claudia Moscovici, *Double Dialectics: Between Universalism and Relativism in Enlightenment and Postmodern Thought* (Lanham, MD: Rowman and Littlefield, 2002), 2.
- 3 Jean-François Lyotard, *The Postmodern Condition: A Report on Knowledge*, trans. Geoff Bennington and Brian Massumi (Minneapolis, MN: University of Minnesota Press, 1988), xxiii.
- 4 Lyotard, xxiv. Quoted in Moscovici, 3.
- 5 Claude Lévi-Strauss, "The Structural Study of Myth," *The Journal of American Folklore* 68(270), (1955): 428–444, p. 429.
- 6 Lévi-Strauss, 430.
- 7 Lévi-Strauss, 430.
- 8 Lévi-Strauss, 430.
- 9 Claude Lévi-Strauss, Myth and Meaning (1978; rpt. Oxford, UK: Routledge), 1.
- 10 Lévi-Strauss, "Study," 431.
- 11 C. G. Prado, *Starting with Foucault: An Introduction to Genealogy* (Boulder, CO: Westview, 1995), 1.
- 12 James W. Bernauer, *Michel Foucault's Force of Flight: Toward an Ethics of Thought* (Atlantic Highlands, NJ: Humanities Press, 1990), 2.
- 13 Michel Foucault, *The Order of Things: An Archaeology of the Human Sciences (Les Mots et les Chose*, 1966; New York: Random House, 1970); and *The Archaeology of Knowledge*, trans. A. M. Sheridan Smith (*L'Archaeology du Savoir*, 1969; New York: Random House, 1972).
- 14 Michel Foucault, *The Birth of the Clinic: An Archaeology of Medical Perception*, trans. A. M. Sheridan Smith (New York: Random House, 1973).
- 15 See: Stuart Elden, Foucault's Last Decade (Cambridge, UK: Polity Press, 2016).
- Michel Foucault, *Michel Foucault: Power, Truth, Strategy*, ed. Meaghan Morris and Paul Patton (Sydney, Australia: Feral Publications, 1979), 32.
- 17 Alec McHoul and Wendy Grace, *A Foucault Primer: Discourse, Power and the Subject* (New York: New York University Press, 1997), 22.
- 18 McHoul and Grace, 22.
- 19 Michel Foucault, 8.
- 20 McHoul and Grace, 21.
- 21 On Foucault's study of madness, see: *The Cambridge Companion to Foucault*, ed. Gary Gutting (Cambridge, UK: Cambridge University Press, 1994), Chap. 2, "Madness and Mental Illness."
- For a sophisticated discussion of the meaning of *discourse* in Foucault's works, see: Manfred Frank, "On Foucault's Concept of Discourse," in *Michel Foucault: Philosopher*, trans. Timothy J. Armstrong (New York: Routledge, 1992), 99–116.
- 23 McHoul and Grace, 35.

- 24 Bernauer, 5.
- 25 Lemert and Gillan, 6.
- 26 McHoul and Grace, 39.
- 27 Edward Craig, ed., *Routledge Encyclopedia of Philosophy* (London: Routledge Press, 1998), 708.
- 28 Gary Gutting, Michel Foucault's Archaeology of Scientific Reason (New York: Cambridge University Press, 1989), 2.
- 29 Elden, 199.
- 30 McHoul and Grace, 18.
- 31 See: Gutting, Chap. 6: "The Archaeology of Knowledge."
- 32 Graham Burchell, Colin Gordon, and Peter Miller, eds., *The Foucault Effect: Studies in Governmentality* (Chicago, IL: University of Chicago Press, 1991), 59–60.
- 33 McHoul and Grace, 29.
- 34 Karlis Racevskis, *Michel Foucault and the Subversion of Intellect* (Ithaca, NY: Cornell University Press, 1983), 58. See also: Lemert and Gillan, Chap. 2, "Historical Archaeology."
- 35 Racevskis, 59.
- 36 David R. Shumway, *Michel Foucault* (Charlottesville, VA: University Press of Virginia, 1989), 56.
- 37 Shumway, 56.
- Joseph Rouse, "Power/Knowledge," in *The Cambridge Companion to Foucault*, ed. Gary Gutting (Cambridge: Cambridge University Press, 1994), 92.
- 39 Gutting, Cambridge Companion, 231.
- 40 Michel Foucault, "Discourse on Language," published with *Archaeology of Knowledge* (New York: Random House, 1972), 215–237, p. 216.
- 41 Foucault, "Discourse on Language," 216.
- 42 Foucault, Archaeology of Knowledge, 217.
- 43 Mel Y. Chen, *Animacies: Biopolitics, Racial Mattering and Queer Affect* (Durham, NC: Duke University Press, 2012), 12.
- 44 See: Gayatri Spivak, translator's preface to Jacques Derrida, *Of Grammatology* (*De la Grammatologie*, 1967; Baltimore, MD: Johns Hopkins University Press, 1976).
- John D. Caputo, ed., *Deconstruction in a Nutshell: A Conversation with Jacques Derrida* (New York: Fordham University Press, 1997), 79.
- 46 Jacques Derrida, *Positions*, trans. and ed. Alan Bass (1972; Chicago, IL: University of Chicago Press, 1981), 19.
- 47 John Caputo, Deconstruction, 205.
- 48 John Caputo, Deconstruction, 79.
- 49 John Caputo, Deconstruction, 52.
- 50 Christopher Norris, *Derrida* (Cambridge, MA: Harvard University Press, 1987), 22.
- James A. Aune, "Rhetoric after Deconstruction," in *Rhetoric and Philosophy*, ed. Richard A. Cherwitz (Hillsdale, NJ: Lawrence Erlbaum, 1990), 253–273, p. 256.
- 52 John Caputo, Deconstruction, 55.
- 53 Norris, 14.
- 54 Norris, 163.
- Jonathan Culler, On Deconstruction (Ithaca, NY: Cornell University Press, 1982),86.
- 56 Norris, 165–166.
- 57 Jacques Derrida, *Writing and Difference*, trans. Alan Bass (*L'écriture et la Différence*; Chicago, IL: University of Chicago Press, 1978), 42.

- 58 John Caputo, Deconstruction, 78.
- 59 John Caputo, Deconstruction, 78.
- 60 Jacques Derrida, *Speech and Phenomena*, trans. David Allison (*La Voix et le Phénomène*, 1967; Evanston, IL: Northwestern University Press, 1973).
- On Derrida's relationship to Habermas, see: Bill Martin, *Matrix and Line* (Albany, NY: State University of New York Press, 1992), Chap. 3, "What Is the Heart of Language? Habermas, Davidson, Derrida."
- 62 Norris, 27.
- 63 Norris, 23, 21.
- 64 Norris, 43.
- 65 John Caputo, Deconstruction, 41.
- 66 John Caputo, Deconstruction, 49.
- 67 John Caputo, Deconstruction, 55.
- 68 In addition to the other sources discussed in this section, see: Elizabeth A. Fay, *Eminent Rhetoric: Language, Gender, and Cultural Tropes* (Westport, CT: Bergen and Garvey, 1994); and Deborah Tannen, *Gender and Discourse* (Oxford: Oxford University Press, 1994).
- 69 Gesa E. Kirsch and Jacqueline J. Royster, "Feminist Rhetorical Practices: In Search of Excellence," *College Composition and Communication* 61(4), (2010): 640–672, pp. 641, 647. See also by these authors: *Feminist Rhetorical Practices: New Horizons for Rhetoric, Composition, and Literacy Studies* (Carbondale, IL: Southern Illinois University Press, 2012).
- General studies of the issue of the differences between men's and women's ways of speaking and knowing can be found in: Carol Gilligan, *In a Different Voice* (Cambridge, MA: Harvard University Press, 1982); Janice Moulton, "A Paradigm of Philosophy: The Adversary Method"; Genevieve Lloyd, "The Man of Reason," and Alison M. Jaggar, "Love and Knowledge" in *Women, Knowledge and Reality: Explorations in Feminist Philosophy*, ed. Ann Gary and Marilyn Pearsall (New York: Routledge, 1989); and Maryann Ayim, "Violence and Domination as Metaphors in Academic Discourse," in *Selected Issues in Logic and Communication*, ed. Trudy Govier (Belmont, CA: Wadsworth, 1988), Chap. 16.
- 71 Jana Sawicki, "Foucault, Feminism and Questions of Identity," in *The Cambridge Companion to Foucault*, ed. Garry Gutting (Cambridge: Cambridge University Press, 1994), 290.
- 72 Sawicki, 288. For an example of Foucault's ideas informing feminist critique of discourse see: Judith Butler, *Bodies that Matter* (London: Routledge, 2011).
- 73 Readings in Feminist Rhetorical Theory, ed. Karen Foss, Sonja Foss, and Cindy Griffin (Thousand Oaks, CA: Sage, 2004).
- 74 Readings, 7.
- 75 Cheris Kramarae, "Proprietors of Language" (1980), in *Readings*, 9–18, p. 9.
- 76 Quoted in Kramarae, 10.
- 77 Kramarae, 11.
- 78 Cheris Kramarae, "Women as a Muted Group" (1981), in Readings, 19–26, 19.
- 79 Kramarae, "Muted," 21.
- 80 Cheris Kramarae, "A Visiting Scholar: How Feminism Has and Has Not Influenced My Field" (1988), in *Readings*, 27–33.
- 81 Cheris Kramarae, "Feminist Theories of Communication" (1989), in *Readings*, 39–44, p. 39.
- 82 Leslie Di Mare, "Rhetoric and Women: The Private and Public Spheres," in Constructing and Reconstructing Gender: The Links Among Communication, Language,

- and Gender, ed. Linda A. M. Perry, Lynn H. Turner, and Helen M. Sterk (Albany, NY: State University of New York Press, 1992), 47.
- 83 Sonja Foss, *Rhetorical Criticism* (Prospect Heights, IL: Waveland, 1989), 151–152.
- 84 Foss, 152.
- Sonia Johnson, "Who's Afraid of the Supreme Court?" in *Readings*, 282–293, p. 284.
- 86 Foss, 152.
- 87 Dale Spender, Man Made Language (Boston, MA: Routledge and Kegan Paul, 1980), 59.
- Mary Daly, "A Call for the Castration of Sexist Religion: Theology after the Demise of 'God the Father," in *Readings*, 107–118, p. 109.
- 89 Spender, 52.
- 90 Women's Voices in Our Time: Statements by American Leaders, ed. Victoria L. DeFrancisco and Marvin D. Jensen (Prospect Heights, IL: Waveland, 1994), ix.
- 91 DeFrancisco and Jensen, ix. They cite studies by Karlyn K. Campbell, "Hearing Women's Voices," *Communication Education* 40 (January 1991): 33–48; and K. S. Vonnegut, "Listening for Women's Voices," *Communication Education* 41 (January 1992): 26–39.
- 92 Sally McConnell-Ginet, et al., *Women and Language in Literature and Society* (New York: Prager, 1980), 66.
- 93 Barrie Thorne and Nancy Henley, *Language and Sex: Difference and Dominance* (Rowley, MA: Newberry House, 1975).
- Joy Ritchie and Kate Ronald, eds., *Available Means: An Anthology of Women's Rhet-oric(s)* (Pittsburgh PA: University of Pittsburgh Press, 2001), xv.
- 95 Available Means, xvi.
- 96 Available Means, xvii.
- 97 Available Means, xvii.
- 98 Available Means, xvii.
- 99 Available Means, xviii.
- 100 Available Means, xxiii.
- 101 Available Means, xxiv.
- 102 Available Means, xxvii.
- 103 Foss, 155.
- Julia T. Wood, Gendered Lives (Belmont, CA: 1994), 91.
- 105 Wood, 91.
- 106 Wood, 91–92.
- 107 Foss, 153.
- 108 Foss, 154.
- 109 Foss, 154.
- 110 Karlyn Kohrs Campbell, Man Cannot Speak for Her: A Critical Study of Early Feminist Rhetoric (New York: Greenwood Press, 1989), 14.
- 111 Campbell, Man Cannot Speak for Her, 14.
- Sally Gearhart, "The Womanization of Rhetoric," *Women's Studies International Quarterly* 2 (1979): 195–201, p. 195.
- 113 Gearhart, 195.
- 114 Gearhart, 195.
- 115 Gearhart, 196.
- 116 George Kennedy, *The Art of Persuasion in Ancient Greece* (Princeton, NJ: Princeton University Press, 1963), 189.

- 117 Gearhart, 196.
- 118 Gearhart, 198.
- 119 Gearhart, 199.
- 120 Gearhart, 200.
- 121 Gearhart, 201.
- 122 S. J. Foss and C. L. Griffin, "Beyond Persuasion: A Proposal for an Invitational Rhetoric," *Communication Monographs* 62 (March 1995): 2–18.
- 123 Foss and Griffin, 2.
- 124 Foss and Griffin, 3.
- 125 Foss and Griffin, 5.
- Foss and Griffin, 5.
- 127 Foss and Griffin, 6.
- Diane Helene Miller, "The Future of Feminist Rhetorical Criticism," in *Listening to Their Voices: The Rhetorical Activities of Historical Women*, ed. M. M. Wertheimer (Columbia, SC: University of South Carolina Press, 1997), 359–380, pp. 361–362.
- 129 Miller, 362.
- 130 Miller, 363.
- Jane Sutton, *The House of My Sojourn: Rhetoric, Women and the Question of Authority* (Tuscaloosa, AL: University of Alabama Press, 2010), 128–129.
- Roland Barthes, "From Work to Text," in *Textual Strategies*, ed. Josue V. Harari, 73–81. Quoted in Miller, 366 (London: Methuen, 1980).
- 133 Miller, 367.
- 134 Miller, 370.
- 135 Miller, 371.
- A. R. de Velasco, "Transgressive Eloquence: Bell hooks, Cicero, and the Aims of Rhetorical Pedagogy," in *Sizing up Rhetoric*, ed. David Zarefsky and Elizabeth Benacka (Long Grove, IL: Waveland, 2008), 393–397, pp. 394, 396.
- LuMing Mao and Bo Wang, "Manifesting a Future for Comparative Rhetoric," *Rhetoric Review* 34(3), (2015): 239–274, p. 241.
- Keith Lloyd, Rethinking Rhetoric from an Indian Perspective: Implications in the "Nyaya Sutra," *Rhetoric Review* 26(4) (2007): pp. 365–384, 366, 367.
- 139 Lloyd, 369
- For example, see: Jon Sun-Gi, "Toward a Rhetoric of Communication, with Special Reference to the History of Korean Rhetoric," *Rhetorica* 28(3) (Summer 2010): 313–329.
- Marcel Detienne, *The Creation of Mythology*, trans. Margaret Cook (Chicago, IL: University of Chicago Press, 1986), 37–38.
- Brian Boyd, *On the Origin of Stories* (Cambridge, MA: Harvard University Press, 2010), 166.
- 143 George Kennedy, *Comparative Rhetoric: An Historical and Cross-Cultural Introduction* (New York: Oxford University Press, 1998).
- 144 Kennedy, Comparative, 4.
- 145 Kennedy, Comparative, 101.
- 146 Kennedy, Comparative, 103.
- 147 Kennedy, Comparative, 105.
- 148 Michael V. Fox, "Ancient Egyptian Rhetoric," *Rhetorica* 1 (1983): 21–22. Quoted in Kennedy, *Comparative Rhetoric*, 131.
- Burton Watson, *Basic Writings of Mo Tzu, Hsun Tzu, and Han Fei-Tzu* (New York: Columbia University Press, 1967), 73. Quoted in Kennedy, *Comparative Rhetoric*, 163–164.

- 150 Kennedy, Comparative Rhetoric, 164.
- James Crump, *Intrigues: Studies of the Chan-Kuo Tse* (Ann Arbor, MI: University of Michigan Press, 1964), 100.
- 152 Crump, 93.
- 153 Crump, 97.
- 154 Crump, 100.
- 155 Paul Rakita Goldin, "Miching Mallecho: The Zhanguo Ce and Classical Rhetoric," *Sino-Platonic Papers* 41 (October 1993): 1–26.
- 156 Goldin, 6.
- 157 Crump, 1.
- 158 Crump, 117.
- 159 Crump, 100.
- 160 Xing Lu, Rhetoric in Ancient China, Fifth to Third Century, B.C.E.: A Comparison with Classical Greek Rhetoric (Columbia, SC: University of South Carolina Press, 1998), 79.
- 161 Lu, 81. On Chinese rhetorical practices, see also: Mary Garrett, "Classical Chinese Conceptions of Argumentation and Persuasion," *Argumentation and Advocacy* 29 (1993): 105–115.
- 162 See: Carol Lea Clark, "Aristotle and Averroes: The Influences of Aristotle's Arabic Commentator upon Western European and Arabic Rhetoric," *Review of Communication* 7(4) (2007): 369–387.
- 163 Clark, 375.
- 164 See: Salim Kemal, *The Philosophical Poetics of Alfarabi, Avicenna and Averroes: The Aristotelian Reception* (London: Routledge, 2003).
- Majid Fakhry, *Averroes: His Life, Work and Influence* (Oxford, UK: Oneworld Publications, 2001), 41–42.
- 166 C. Butterworth in Averroes, *Averroes's "Three Short Commentaries on Aristotle's 'Topics,' 'Rhetoric,' and 'Poetics,'"* ed. and trans. C. Butterworth (Albany, NY: State University of New York Press, 1977), 30. Quoted in Clark, 375.
- 167 Clark, 377.
- 168 Clark, 378.
- 169 Kennedy, Comparative Rhetoric, 198.
- 170 Kennedy, Comparative Rhetoric, 199.
- 171 Kennedy, Comparative Rhetoric, 199.
- 172 Kennedy, Comparative Rhetoric, 203.
- 173 Kennedy, *Comparative Rhetoric*, 211. Similarly, Douglas Thomas writes of Nietzsche's observation about Greek competitiveness: "Nietzsche argues that since Homer's time this will to annihilation has been contained by the notion of *contest*." Douglas Thomas, *Reading Nietzsche Rhetorically* (New York: Guilford Press, 1999), 6.
- 174 Kennedy, Comparative Rhetoric, 217.
- 175 Kennedy, Comparative Rhetoric, 221.
- 176 Kennedy, Comparative Rhetoric, 221.

Bibliography

- Abbott, Don Paul. "Marxist Influences on the Rhetorical Theory of Kenneth Burke: Theory of Kenneth Burke," *Philosophy and Rhetoric* 4 (1974): 217–233.
- Abbott, Don Paul. "The Renaissance," in *The Present State of Scholarship in Historical and Contemporary Rhetoric*, ed. Winifred Bryan Horner, Columbia, MO: University of Missouri Press, 1990a: 82–106.
- Abbott, Don Paul. "Rhetoric and Writing in Renaissance Europe and England," in A Short History of Writing Instruction: From Ancient Greece to Twentieth-Century America, ed. James J. Murphy, Davis, CA: Hermagoras Press, 1990b: 148–171.
- Agricola, Rudolph. De Inventione Dialectica. Nieuwkoop, Holland: B. de Graaf, 1967.
- Alessandro, Duranti. From Grammar to Politics: Linguistic Anthropology in a Western Samoan Village. Berkeley, CA: University of California Press, 1994.
- Anderson, Graham. Sage, Saint, and Sophist: Holy Men and Their Associates in the Early Roman Empire. London: Routledge, 1994.
- Anderson, John R. "The Audience as a Concept in the Philosophic Rhetoric of Perelman, Johnstone, and Natanson," *Southern Speech Communication Journal* 38 (1972): 39–50.
- Andrews, Richard, ed. Narrative and Argument. Milton Keynes, UK: Open University Press, 1989.
- Andrews, Richard, ed. Rebirth of Rhetoric: Essays in Language, Culture, and Education. London: Routledge, 1992.
- Aristotle. Rhetoric. Trans. Lane Cooper. New York: Appleton-Century-Crofts, 1932.
- Aristotle. Rhetoric. Trans. W. Rhys Roberts. New York: Modern Library, 1954.
- Arnhart, Larry. Aristotle on Political Reasoning: A Commentary on the Rhetoric. DeKalb, IL: Northern Illinois University Press, 1981.
- Augustijn, Cornelis. *Erasmus: His Life, Works, and Influence*. Trans. J. C. Grayson. Toronto: University of Toronto Press, 1991.
- Augustine. On Christian Doctrine. Trans. D. W. Robertson. Indianapolis, IN: Library of Liberal Arts, 1958.
- Aune, James A. "Rhetoric after Deconstruction," in *Rhetoric and Philosophy*, ed. Richard A. Cherwitz, Hillsdale, NJ: Lawrence Erlbaum, 1990: 253–273.
- Bacon, Wallace A. "The Elocutionary Career of Thomas Sheridan," Speech Monographs 31 (1964): 1-53.
- Bakhtin, Mikhail. *Marxism and the Philosophy of Language*. Trans. L. Matejka and I. R. Titunik. New York: Seminar Press, 1973.
- Bakhtin, Mikhail. *Problems of Dostoevsky's Poetics*. Ed. and trans. Caryl Emerson. Minneapolis, MN: University of Minnesota Press, 1984.
- Barilli, Renato. *Rhetoric*. Trans. Giuliana Menozzi. Minneapolis, MN: University of Minnesota Press,
- Barnes, John A. A Pack of Lies: Towards A Sociology of Lying. Cambridge: Cambridge University Press, 1994.
- Barnes, Jonathan, ed. *The Cambridge Companion to Aristotle*. Cambridge: Cambridge University Press, 1995.

Barnett, Scot and Casey Boyle, ed. Rhetoric, through Everyday Things. Tuscaloosa AL: University of Alabama Press, 2016.

Barrett, Harold. The Sophists. Novato, CA: Chandler and Sharp, 1987.

Barthes, Roland. "From Work to Text," in *Textual Strategies*, ed. Josue Harari, Ithaca, NY: Cornell University Press, 1979: 17–72.

Bauer, Dale M. and S. Jaret McKinstry, eds. Feminism, Bakhtin, and the Dialogic. Albany, NY: State University of New York Press, 1991.

Bazerman, Charles. Shaping Written Knowledge: The Genre and Activity of the Experimental Article in Science. Madison, WI: University of Wisconsin Press, 1988.

Benardete, Seth. The Rhetoric of Morality and Philosophy: Plato's Gorgias and Phaedrus. Chicago, IL: University of Chicago Press, 1991.

Benson, Thomas W. and Michael. Prosser, eds. Readings in Classical Rhetoric. Boston, MA: Allyn and Bacon, 1969.

Berlin, James A. "Revisionary Histories of Rhetoric: Politics, Power, and Plurality," in Writing Histories of Rhetoric, ed. Victor Vitanza, Carbondale, IL: Southern Illinois University Press, 1994: 112–127.

Bernauer, James W. Michel Foucault's Force of Flight: Toward an Ethics of Thought. Atlantic Highlands, NJ: Humanities Press, 1990.

Bevilaqua, Vincent. "Philosophical Origins of George Campbell's Philosophy of Rhetoric," Speech Monographs 32 (1965): 7–8.

Billig, Michael. "Psychology, Rhetoric, and Cognition," History of the Human Sciences 2 (1989): 289-307.

Billig, Michael. Ideology and Opinion. London: Sage, 1991.

Billig, Michael. Arguing and Thinking, 2nd ed. Cambridge: Cambridge University Press, 1996.

Bishop, Morris. Petrarch and His World. Bloomington, IN: Indiana University Press, 1963.

Bitzer, Lloyd. "Aristotle's Enthymeme Revisited," Quarterly Journal of Speech 45 (1959): 399-408.

Bitzer, Lloyd. "The Rhetorical Situation," *Philosophy and Rhetoric* 1 (1968): 1–14.

Bitzer, Lloyd. "Hume's Philosophy in George Campbell's Philosophy of Rhetoric," Philosophy and Rhetoric 2 (1969): 136–166.

Bitzer, Lloyd. "Rhetoric and Public Knowledge," in Rhetoric, Philosophy, and Literature: An Exploration, ed. Don Burks, West Lafayette, IN: Purdue University Press, 1978: 67–93.

Bitzer, Lloyd. "Political Rhetoric," in Landmark Essays on Contemporary Rhetoric, ed. Thomas Farrell, Mahwah, NJ: Hermagoras Press, 1998: 1–22.

Bizzell, Patricia and Bruce Herzberg. The Rhetorical Tradition: Readings from Classical Times to the Present, 2nd ed. Boston, MA: St. Martin's Press, 2000.

Black, Edwin. "Plato's View of Rhetoric," Quarterly Journal of Speech 44 (December 1958): 361-374.

Blair, Hugh. Lectures on Rhetoric and Belles Lettres, ed. Harold Harding, Carbondale, IL: Southern University Press, 1965.

Boethius. De Topicis Differentiis. Trans. Eleanor Stump. Ithaca, NY: Cornell University Press, 1978.

Booth, Wayne. The Rhetoric of Fiction. Chicago, IL: University of Chicago Press, 1961.

Booth, Wayne. Now Don't Try to Reason with Me. Chicago, IL: University of Chicago Press, 1970.

Booth, Wayne. Modern Dogma and the Rhetoric of Assent. Chicago, IL: University of Chicago Press, 1974.

Booth, Wayne. The Vocation of a Teacher. Chicago, IL: University of Chicago Press, 1988.

Bormann, Ernest. "Fantasy and Rhetorical Vision: The Rhetorical Criticism of Social Reality," *Quarterly Journal of Speech* 59 (1972): 396–407.

Bormann, Ernest. "Symbolic Convergence: Organizational Communication and Culture," in *Communication and Organizations: An Interpretive Approach*, eds. Linda L. Putnam and Michael. E. Pacanowsky, Thousand Oaks, CA: Sage, 1983: 99–122.

Bowersock, Glen. W. Greek Sophists in the Roman Empire. Oxford: Clarendon Press, 1969.

Breitman, George, ed. Malcolm X Speaks. New York: Grove Press, 1966.

Browne, Stephen H. "Shandyean Satire and the Rhetorical Arts in Eighteenth-Century England," *Southern Communication Journal* 55 (1990): 191–205.

Brucker, Gene. Florence: The Golden Age 1138-1737. Berkeley, CA: University of California Press, 1969.

Brummett, Barry. Rhetoric in Popular Culture. New York: St. Martin's Press, 1994.

Burchell, Graham, Colin Gordon, and Peter Miller, eds. *The Foucault Effect*. Chicago, IL: University of Chicago Press, 1991.

Burke, John G. "Hermetism as a Renaissance World View," in *The Darker Vision of the Renaissance: Beyond the Fields of Reason*, ed. Robert Kinsman, Berkeley CA: University of California Press, 1974: 95–117.

Burke, Kenneth. A Rhetoric of Motives. Berkeley, CA: University of California Press, 1950.

Burke, Kenneth. The Rhetoric of Religion. Berkeley, CA: University of California Press, 1961.

Burke, Kenneth. Language as Symbolic Action. 1937. Berkeley, CA: University of California Press, 1966.

Burke, Kenneth. Counter-Statement. 1931. Berkeley, CA: University of California Press, 1968.

Burke, Kenneth. A Grammar of Motives. 1945. Berkeley, CA: University of California Press, 1969.

Burke, Kenneth. The Philosophy of Literary Form: Studies in Symbolic Action. 1941. Berkeley, CA: University of California Press, 1973.

Burke, Peter. Vico. Oxford: Oxford University Press, 1985.

Burke, Peter. "The Spread of Humanism," in *The Impact of Humanism in Western Europe*, eds. Anthony Goodman and Angus MacKay, London: Longman, 1990: 1–22.

Butler, Judith. Bodies that Matter: On the Discursive Limits of Sex. New York: Routledge, 1994.

Butterworth, Charles. Averroes's "Three Short Commentaries on Aristotle's 'Topics,' 'Rhetoric,' and 'Poetics'. Ed. and trans. C. Butterworth. Albany, NY: State University of New York Press, 1977.

Cameron, James K. "Humanism in the Low Countries," in *The Impact of Humanism on Western Europe*, eds. Anthony Goodman and Angus MacKay, London: Routledge, 1990: 137–163.

Campbell, George. The Philosophy of Rhetoric, 1776. ed. Lloyd F. Bitzer, Carbondale, IL: University of Southern Illinois Press, 1963.

Campbell, John Angus. "Darwin, Thales, and the Milkmaid: Scientific Revolution and Argument from Common Beliefs and Common Sense," in *Perspectives on Argument: Essays in Honor of Wayne Brockriede*, eds. Robert Trapp and Janice Schuetz, Prospect Heights, IL: Waveland, 1990a: 207–220.

Campbell, John Angus. "Scientific Discovery and Rhetorical Invention: The Path of Darwin's Origin," in *The Rhetorical Turn*, ed. Herbert Simon, Chicago, IL: University of Chicago Press, 1990b: 58–90.

Campbell, Karlyn K. Man Can Not Speak for Her: A Critical Study of Early Feminist Rhetoric. New York: Greenwood Press, 1989.

Campbell, Karlyn K. "Hearing Women's Voices," Communication Education 40 (1991): 33-48.

Caputo, John D., ed. Deconstruction in A Nutshell: A Conversation with Jacques Derrida. New York: Fordham University Press, 1997.

Caputo, John D. and Mark Yount. Foucault and the Critique of Institutions. University Park, PA: Pennsylvania State University Press, 1993.

Carleton, Walter. "What Is Rhetorical Knowledge?" Quarterly Journal of Speech 64 (1978): 313-328.

Casson, Lionel. Selected Satires of Lucian. New York: Norton, 1968.

Castiglione, Ballasdare. The Courtier. n.p. 1528.

Chaim, Perelman and Lucy Olbrechts-Tyteca. The New Rhetoric: A Treatise on Argumentation. 1958.

Trans. John Wilkinson and Purcell Weaver. Notre Dame, IN: University of Notre Dame Press, 1969.

Charles, David. Aristotle's Philosophy of Action. Ithaca, NY: Cornell University Press, 1984.

Chen, Mel Y. Animacies: Biopolitics, Racial Mattering and Queer Affect. Durham, NC: Duke University Press, 2012.

Cialdini, Robert. Influence: The Psychology of Persuasion. New York: William Morrow, 1993.

Cicero. Ed. T. A. Dorey. New York: Basic Books, 1965.

Cicero. De Inventione. Trans. H. M. Hubbell. Cambridge, MA: Loeb Classical Library, 1976a.

Cicero. De Oratore. Trans. E. W. Sutton and H. Rackham. Cambridge, MA: Loeb Classical Library, 1976b.

Clark, Carol Lea. "Aristotle and Averroes: The Influences of Aristotle's Arabic Commentator upon Western European and Arabic Rhetoric," *Review of Communication* 7(4) (2007): 369–387.

Clark, Martin L. Rhetoric at Rome. New York: Barnes and Noble, 1953.

Classen, Albrecht. "Female Explorations of Literacy: Epistolary Challenges to the Literary Canon in the Late Middle Ages," in *Disputatio Volume 1: The Late Medieval Epistle*, eds. Carol Poster and Richard Utz, Evanston, IL: Northwestern University Press, 1996: 89–122, p. 90.

Cmiel, Kenneth. Democratic Eloquence: The Fight over Popular Speech in Nineteenth-Century America. New York: William Morrow, 1990.

Coates, Willson Havelock. The Emergence of Liberal Humanism: An Intellectual History of Western Europe. New York: McGraw Hill, 1966.

- Cohen, Herman. "William Leechman's Anticipation of Campbell," Western Speech 32 (1968): 92-99.
- Cole, Percival. Later Roman Education in Ausonius, Capella, and the Theodosian Code. New York: Columbia University Press, 1909.
- Conley, Thomas M. "Logical Hylomorphism' and Aristotle's Koinoi Topoi," Central States Speech Journal 29 (1978): 92-97.
- Conley, Thomas M. Rhetoric in the European Tradition. Chicago, IL: University of Chicago Press, 1994.
- Connolly, Joy. The State of Speech: Rhetorical and Political Thought in Ancient Rome. Princeton, NJ: Princeton University Press, 2013.
- Corbett, Edward P. J. "The Topoi Revisited," in Rhetoric and Praxis: The Contribution of Classical Rhetoric to Practical Thinking, ed. Jean Deitz Moss, Washington, DC: Catholic University Press of America, 1986: 79–106.
- Coupe, Laurence. Myth. New York: Routledge, 1997.
- Covino, William A. Magic, Rhetoric and Literacy: An Eccentric History of the Composing Imagination. Albany, NY: State University of New York Press, 1994.
- Craig, Christopher P. Form as Argument in Cicero's Speeches: A Study of Dilemma. American Classical Studies, 31. Atlanta: Scholars Press, 1993.
- Craig, Edward, ed. Routledge Encyclopedia of Philosophy. London: Routledge, 1998.
- Crombie, I. M. An Examination of Plato's Doctrines. London: Routledge and Kegan Paul, 1961.
- Crump, James. Intrigues: Studies of the Chan-Kuo Tse. Ann Arbor, MI: University of Michigan Press, 1964. Culler, Jonathan. On Deconstruction. Ithaca, NY: Cornell University Press, 1982.
- Curran, Jane V. "The Rhetorical Technique of Plato's Phaedrus," Philosophy and Rhetoric 19 (1986): 66-72.
- de Lauretis, Teresa. "Queer Theory: Lesbian and Gay Sexualities," Differences: A Journal of Feminist Cultural Studies 3 (1991): iii-xviii.
- de Lauretis, Teresa. "Habit Changes," Differences: A Journal of Feminist Cultural Studies 6 (1994): 296–313.
- de Romilly, Jacqueline. *Magic and Rhetoric in Ancient Greece*. Cambridge, MA: Harvard University Press, 1975.
- de Romilly, Jacqueline. The *Great Sophists in Periclean Athens*. Trans. Janet Lloyd. Oxford: Clarendon Press, 1992.
- Dearin, Ray D., ed. The New Rhetoric of Chaim Perelman: Statement and Response. Lanham, MD: University Press of America, 1989.
- DeFrancisco, Victoria and Marvin Jensen, eds. Women's Voices in Our Time. Prospect Heights, IL: Waveland, 1994.
- DePew, David J. and John Lyne. "The Productivity of Scientific Rhetoric," *POROI: An Interdisciplinary Journal of Rhetorical Analysis and Invention* 9(1) (2013): 1–16.
- Derrida, Jacques. Speech and Phenomena (La Voix Et Le Phénomène). Trans. David Allison 1967. Evanston, IL: Northwestern University Press, 1973.
- Derrida, Jacques. Of Grammatology (De la Grammatologie). Trans. Gayatri Spivak 1967. Baltimore, MD: Johns Hopkins University Press, 1976.
- Derrida, Jacques. Writing and Difference (L' Écriture Et La Différence). Trans. Alan Bass. Chicago, IL: University of Chicago Press, 1978.
- Derrida, Jacques. *Positions*. Trans. and ed. Alan Bass. 1972. Chicago, IL: University of Chicago Press, 1981.
- Despland, Michael. The Education of Desire: Plato and the Philosophy of Religion. Toronto: University of Toronto Press, 1985.
- Detienne, Marcel. *The Creation of Mythology*. Trans. Margaret Cook. Chicago, IL: University of Chicago Press, 1986.
- de Velasco, A. R. "Transgressive Eloquence: Bell Hooks, Cicero, and the Aims of Rhetorical Pedagogy," in *Sizing up Rhetoric*, eds. David Zarefsky and Elizabeth Benacka, Long Grove, IL: Waveland Press, 2008: 393–397.
- Dickinson, Emily. The Poems of Emily Dickinson, ed. Thomas H. Johnson. Cambridge, MA: Belknap Press of Harvard University, 1983.
- Dillon, John and Tania Gergel, eds. The Greek Sophists. New York: Penguin Books, 2003.
- Dipardo, Anne. "Narrative Knowers, Expository Knowledge," Written Communication 7 (1990): 59-95.
- Disch, Lisa J. "More Truth than Fact: Storytelling as Critical Understanding in the Writings of Hannah Arendt," *Political Theory* 4 (1994): 665–694.

- Donawerth, Jane, ed. Rhetorical Theory by Women before 1900. Lanham, MD: Rowman and Littlefield,
- Donovan, Josephine. "Feminism, Bakhtin, and the Dialogic," in *In Feminism, Bakhtin, and the Dialogic*, eds. Dale M. Bauer and S. Jaret McKinstry, Albany, NY: State University of New York Press, 1991: 84–91.
- Ede, Lisa S. "Rhetoric Vs. Philosophy: The Role of the Universal Audience in Chaim Perelman's The New Rhetoric," Central States Speech Journal 32 (1981): 118–125.
- Edwards, Mark U., Jr. Printing, Propaganda, and Martin Luther. Berkeley, CA: University of California Press,
- Ehninger, Douglas. "George Campbell and the Revolution in Inventional Thinking," *Southern Speech Journal* 15 (1950): 270–276.
- Ehninger, Douglas. "Campbell, Blair, and Whately: Old Friends in a New Light," Western Speech 19 (1955): 263-269.
- Ehninger, Douglas. "Campbell, Blair and Whately Revisited," Southern Speech Journal 28 (1963): 169-182.

Elden, Stuart. Foucault's Last Decade. Cambridge: Polity Press, 2016.

Enos, Richard Leo. Greek Rhetoric before Aristotle. Prospect Heights, IL: Waveland, 1993.

Erasmus. Collected Works, Vol. 4, ed. J. K. Sowards, Toronto: University of Toronto Press, 1985.

Erickson, Keith V., ed. Aristotle: The Classical Heritage of Rhetoric. Metuchen, NJ: The Scarecrow Press, 1974.

Fahnestock, Jeanne. Rhetorical Figures in Science. Oxford: Oxford University Press, 1999.

Fakhry, Majid. Averroes: His Life, Work and Influence. Oxford: Oneworld Publications, 2001.

Faral, Edmond. Les Artes Poetique du XIIe et XIIIe Siècle. (1924) Paris: n.p., 1971.

Farmer, Frank, ed. Landmark Essays on Bakhtin, Rhetoric and Writing. Mahwah, NJ: Hermagoras Press, 1998.

Farrell, Thomas. "Social Knowledge II," Quarterly Journal of Speech 64 (1978): 329-334.

Farrell, Thomas. Norms of Rhetorical Culture. New Haven, CT: Yale University Press, 1995.

Faulhaber, Charles. "The Origins of Humanism," in Medieval Eloquence: Studies in the Theory and Practice of Medieval Rhetoric, ed. James J. Murphy, Berkeley, CA: University of California Press, 1978: 85–111.

Fay, Elizabeth A. Eminent Rhetoric: Language, Gender, and Cultural Tropes. Westport, CT: Bergen and Garvey, 1994.

Fisher, Walter. Human Communication as Narration: Toward a Philosophy of Reason, Value, and Action. Columbia, SC: University of South Carolina Press, 1987.

Fleckenstein, Kristie S. "In the Blood of the Word: Embodied Rhetorical Authority in St. Catherine of Siena's *Dialogue*," in *Sizing up Rhetoric*, eds. David Zarefsky and Elizabeth Benacka, Long Grove, IL: Waveland Press, 2008: 287–295.

Ford, Andrew. "The Price of Art in Isocrates: Formalism and the Escape from Politics," in *Rethinking the History of Rhetoric*, ed. Takis Poulakos, Boulder, CO: Westview, 1992: 31–42.

Fortenbaugh, William W. Aristotle on Emotion. New York: Barnes and Noble, 1975.

Fortenbaugh, William W. and David C. Mirhady, eds. *Peripatetic Rhetoric after Aristotle*. New Brunswick, NJ: Transaction, 1994.

Foss, Sonja. Rhetorical Criticism. Prospect Heights, IL: Waveland, 1989.

Foss, Sonja and C. L. Griffin. "Beyond Persuasion: A Proposal for an Invitational Rhetoric," *Communication Monographs* 62 (1995): 2–18.

Foss, Sonja, Karen A. Foss, and Robert Trapp. Contemporary Perspective on Rhetoric, 2nd ed. Prospect Heights, IL: Waveland, 1991.

Foucault, Michel. The Order of Things: An Archaeology of the Human Sciences (Les Mots Et Les Chose). 1966. New York: Random House, 1970.

Foucault, Michel. The Archaeology of Knowledge (L'archaeologie du Savoir). Trans. A. M. Sheridan Smith 1969. New York: Random House, 1972.

Foucault, Michel. The Birth of the Clinic: An Archaeology of Medical Perception. Trans. A. M. Sheridan Smith. New York: Random House, 1973.

Foucault, Michel. Discipline and Punish: The Birth of the Prison. London: Allen Lane, 1975.

Fox, Michael V. "Ancient Egyptian Rhetoric," Rhetorica 1 (1983): 21–22.

- Frank, David A. "A Traumatic Reading of Twentieth-Century Rhetorical Theory: The Belgian Holocaust, Malines, Perelman, and de Man," *Quarterly Journal of Speech* 93(3) (August 2007): 308–343.
- Frank, Manfred. "On Foucault's Concept of Discourse," in *Michel Foucault: Philosopher*, trans. Timothy J. Armstrong, New York: Routledge, 1992: 99-116.
- Fredborg, Karin Margareta. "Twelfth-Century Ciceronian Rhetoric: Its Doctrinal Development and Influences," in *Rhetoric Revalued*, ed. Brian Vickers, Binghamton, NY: Center for Medieval and Early Renaissance Studies, 1982: 87–97.
- Fuhrmann, Manfred. Cicero and the Roman Republic. Trans. W. E. Yuill. Oxford: Blackwell, 1992.
- Gadamer, Hans-George. The Idea of the Good in Platonic-Aristotelian Philosophy. New Haven, CT: Yale University Press, 1986.
- Gagarin, Michael. The Murder of Herodes: A Study of Antiphon 5. New York: Verlag Peter Lang, 1989.
- Gagarin, Michael. Antiphon and the Athenians: Oratory, Law and Justice in the Age of the Sophists. Austin, TX: University of Texas Press, 2002.
- Gagarin, Michael. "Telling Stories in Athenian Law," Transactions of the American Philological Association 133(2) (2003): 197–207.
- Gallo, Ernest. "The Poetria Nova of Geoffrey of Vinsauf," in Medieval Eloquence: Studies in the Theory and Practice of Medieval Rhetoric, ed. James J. Murphy, Berkeley, CA: University of California Press, 1978: 68–84.
- Gaonkar, Dilip Parameshwar. "The Idea of Rhetoric in the Rhetoric of Science," Southern Communication Journal 58 (1993): 255-327.
- Garin, Euginio. Portraits of the Quattrocentro. New York: Harper and Row, 1963.
- Garry, Ann and Marilyn Pearsall, eds. Women, Knowledge and Reality: Explorations in Feminist Philosophy. New York: Routledge, 1989.
- Gearhart, Sally Miller. "The Womanization of Rhetoric," Women's Studies International Quarterly 2 (1979): 195-201.
- Geertz, Clifford. Works and Lives: The Anthropologist as Author. Stanford, CT: Stanford University Press, 1988.
- Geoffrey of Vinsauf. Documentum Do Modo Et Arte Dictandi Et Versificandi. Trans. Roger P. Parr. Milwaukee, WI: Marquette University Press, 1968.
- Gergen, Kenneth and Mary M. Gergen. "Narrative Form and the Construction of Psychological Sciences," in *Narrative Psychology*, ed. T. R. Sarbin, New York: Praeger Press, 1986: 22-44.
- Gilbert, Katherine E. and Helmut Kuhn. A History of Esthetics. Bloomington, IN: University of Indiana Press, 1953.
- Gill, Ann. Rhetoric and Human Understanding. Prospect Heights, IL: Waveland, 1994.
- Gilligan, Carol. In a Different Voice. Cambridge, MA: Harvard University Press, 1982.
- Glenn, Cheryl. "Locating Aspasia on the Rhetorical Map," in *Listening to Their Voices*, ed. Molly Meijer Wertheimer, Columbia, SC: University of South Carolina Press, 1997: 19–41.
- Golden, James L. and Edward P. J. Corbett. The Rhetoric of Blair, Campbell, and Whately. New York: Holt, Rinehart and Winston, 1968.
- Goldin, Paul Rakita. "Miching Mallecho: The Zhanguo Ce and Classical Rhetoric," *Sino-Platonic Papers* 41 (1993): 1–26.
- Goodman, Anthony and Angus MacKay, eds. The Impact of Humanism on Western Europe. London: Longman, 1990.
- Gorgias. Encomium on Helen. Trans. LaRue VanHook. The Classical Weekly 6 (1913): 122-123.
- Gotoff, Harold C. Cicero's Caesarian Speeches: A Stylistic Commentary. Chapel Hill, NC: University of North Carolina Press, 1993.
- Govier, Trudy. Selected Issues in Logic and Communication. Belmont, CA: Wadsworth, 1988.
- Grassi, Ernesto. Rhetoric as Philosophy. University Park, PA: Pennsylvania State University Press, 1980.
- Gregg, Richard B. Symbolic Inducement and Knowing: A Study in the Foundations of Rhetoric. Columbia, SC: University of South Carolina Press, 1984.
- Gries, Laurie. Still Life with Rhetoric: A New Materialist Approach for Visual Rhetorics. Logan, UT: Utah, 2015.
- Grimaldi, Michael. "The Aristotelian Topics," *Traditio* 14 (1958): 1–16.
- Grimaldi, Michael. Aristotle, Rhetoric I: A Commentary. New York: Fordham University Press, 1980.

Gronbeck, Bruce E. "Gorgias on Rhetoric and Poetic: A Rehabilitation," Southern Speech Communication Journal 38 (1972): 27-38.

Gross, Alan. The Rhetoric of Science. Cambridge: Harvard University Press, 1990.

Gross, Alan. "A Theory of the Rhetorical Audience: Reflections on Chaim Perelman," *Quarterly Journal of Speech* 85, (1999): 202–211.

Gross, Alan, Joseph Harmon, and Michael Reidy. Communicating Science: The Scientific Article from the 17th Century to the Present. Oxford: Oxford University Press, 2002.

Grube, G. Introduction to Longinus, on the Sublime. Indianapolis, IN: Library of Liberal Arts, 1957.

Grube, G. Plato's Thought. 1935. Boston, MA: Beacon Press, 1958.

Grube, G. M. A. The Greek and Roman Critics. Toronto: University of Toronto Press, 1965.

Guthrie, W. K. C. The Sophists. Cambridge: Cambridge University Press, 1971.

Gutting, Gary, ed. Michel Foucault's Archaeology of Scientific Reason. New York: Cambridge University Press, 1989.

Gutting, Gary, ed. The Cambridge Companion to Foucault. Cambridge: Cambridge University Press, 1994.

Habermas, Jürgen. Toward a Rational Society. Trans. Jeremy J. Shapiro. Boston, MA: Beacon Press, 1970.

Habermas, Jürgen. Knowledge and Human Interests. Trans. Jeremy J. Shapiro. Boston, MA: Beacon Press, 1971.

Habermas, Jürgen. Theory and Practice. Trans. John Viertel. Boston, MA: Beacon Press, 1973.

Habermas, Jürgen. Communication and the Evolution of Society. Trans. Thomas McCarthy. Boston, MA: Beacon Press, 1979.

Habermas, Jürgen. The Philosophical Discourse of Modernity: Twelve Lectures. Ed. Thomas McCarthy. Trans. Frederick Lawrence. Cambridge, MA: MIT Press, 1987.

Habermas, Jürgen. The Lure of Technocracy. Cambridge: Polity Press, 2016.

Habicht, Christian. Cicero the Politician. Baltimore, MD: Johns Hopkins University Press, 1990.

Halperin, David. Saint Foucault: Toward a Gay Hagiography. New York: Oxford University Press, 1995.

Haraway, Donna J. Staying with the Trouble: Making Kin in the Chthulucene. Durham, NC: Duke University Press, 2016.

Hare, R. M. Plato. Oxford: Oxford University Press, 1982.

Haskins, Ekaterina. Logos and Power in Isocrates and Aristotle. Columbia, SC: University of South Carolina Press, 2004.

Hauser, Gerard. "The Example in Aristotle's *Rhetoric*: Bifurcation or Contradiction?" *Philosophy and Rhetoric* 1 (1968): 78–90.

Hauser, Gerard. "Empiricism, Description and the New Rhetoric," *Philosophy and Rhetoric* 5 (1972): 24-44.

Hazzard, Paul. La Pensée Européenne au XVIIIe Siècle de Montesquieu À Lessing. Paris: Arthone Fayard, 1963.

Herman, Cohen. "Belles-lettres," in Encyclopedia of Composition and Rhetoric: Communication from Ancient Times to the Information Age, ed. Theresa Enos, New York: Routledge, 2004: 71–72.

Hill, Forbes I. "The Rhetoric of Aristotle," in A Synoptic History of Classical Rhetoric. Davis, CA: Hermagoras Press, 1983: 19-76.

Hogan, Michael J., ed. Rhetoric and Community: Studies in Unity and Fragmentation. Columbia, SC: University of South Carolina Press, 1998.

Holquist, Michael. Dialogism: Bakhtin and His World. London: Routledge, 1990.

Home, Henry Lord Kames. The Elements of Criticism. 1761. New York: Barnes and Burr, 1865.

Horner, Winifred Bryan, ed. The Present State of Scholarship in Historical and Contemporary Rhetoric. Columbia, MO: University of Missouri Press, 1990.

Howard, Dick. "A Politics in Search of the Political," Theory and Society 1 (1974): 271-306.

Howell, Wilbur Samuel. Logic and Rhetoric in England: 1500-1700. New York: Russell and Russell, 1961.

Howell, Wilbur Samuel. "John Locke and the New Rhetoric," *Quarterly Journal of Speech* 53 (1967): 319–321.

Howell, Wilbur Samuel. Eighteenth-Century British Logic and Rhetoric. Princeton, NJ: Princeton University Press, 1971.

Hoy, David Couzens. Foucault: A Critical Reader. Oxford: Basil Blackwell, 1986.

- Hubbard, B. A. F. and E. S. Karnofsky. *Plato's Protagoras: A Socratic Commentary*. Chicago, IL: University of Chicago Press, 1982.
- Jarratt, Susan and Rory Ong, "Aspasia: Rhetoric, Gender, and Colonial Ideology," in *Reclaiming Rhetorica:* Women in the Rhetorical Tradition, ed. Andrea Lunsford, Pittsburgh, PA: University of Pittsburgh Press, 1995; 9–24.
- Jarratt, Susan. Rereading the Sophists: Classical Rhetoric Refigured. Carbondale, IL: Southern Illinois University Press, 1991.
- Jennings, Margaret C. S. J. "The Ars Compendi Sermones of Ranulph Higden," in Medieval Eloquence: Studies in the Theory and Practice of Medieval Rhetoric, ed. James J. Murphy, Berkeley, CA: University of California Press, 1978: 112–126.
- Johannesen, Richard L., Rennard Strickland, and Ralph T. Eubanks, eds. Language Is Sermonic: Richard M. Weaver on the Nature of Rhetoric. Baton Rouge, LA: Louisiana State University Press, 1970.
- Johnson, W. R. "Isocrates Flowering: The Rhetoric of Augustine," *Philosophy and Rhetoric* 9 (1976): 217–231.
- Jonsen, Albert R. and Stephen Toulmin. *The Abuse of Casuistry*. Berkeley, CA: University of California Press, 1988.
- Kahn, Victoria. Rhetoric, Prudence, and Skepticism in the Renaissance. Ithaca, NY: Cornell University Press, 1985.
- Kahn, Victoria. *Machiavelian Rhetoric: From the Counter-Reformation to Milton*. Princeton, NJ: Princeton University Press, 1994.
- Katula, Richard and James J. Murphy, eds. A Synoptic History of Classical Rhetoric. Davis, CA: Hermagoras Press, 1995.
- Katz, Stephen B. "The Ethic of Expediency: Classical Rhetoric, Technology, and the Holocaust," *College English* 54 (1992): 255–275.
- Kelley, Donald R. Renaissance Humanism. Boston, MA: Twayne Publishers, 1991.
- Kennedy, George, trans. The Art of Persuasion in Ancient Greece. Princeton, NJ: Princeton University Press, 1963.
- Kennedy, George, trans. Classical Rhetoric and Its Secular and Christian Tradition. Chapel Hill, NC: University of North Carolina Press, 1980.
- Kennedy, George, trans. Aristotle on Rhetoric: A Theory of Civic Discourse. New York: Oxford University Press, 1991.
- Kennedy, George, trans. Comparative Rhetoric: An Historical and Cross-Cultural Introduction. New York: Oxford University Press, 1998.
- Kennedy, George, trans. Classical Rhetoric and Its Secular and Christian Tradition, 2nd ed. Chapel Hill, NC: University of North Carolina Press, 1999.
- Kenny, Anthony. Aristotle on the Perfect Life. Oxford: Clarendon Press, 1992.
- Kerferd, G. B. The Sophistic Movement. Cambridge: Cambridge University Press, 1981.
- Kersey, Shirley. "Medieval Education of Girls and Women," Educational Horizons 58(4) (1980): 188-192.
- Kinney, Arthur F. Humanist Poetics: Thought, Rhetoric, and Fiction in Sixteenth-Century England. Amherst, MA: University of Massachusetts Press, 1986.
- Kinsman, Robert, ed. The Darker Vision of the Renaissance: Beyond the Fields of Reason. Berkeley, CA: University of California Press, 1974.
- Kirsch, Gesa E. and Jacqueline J. Royster. "Feminist Rhetorical Practices: In Search of Excellence," *College Composition and Communication* 61(4) (2010): 640–672.
- Kitto, H. D. F. The Greeks. Baltimore, MD: Penguin Books, 1968.
- Kraye, Jill, ed. The Cambridge Companion to Renaissance Humanism. Cambridge: Cambridge University Press, 1996.
- Kristeller, Paul Oskar. Renaissance Thought: The Classic, Scholastic, and Humanist Strains. New York: Harper and Row, 1961.
- Kuhn, Thomas. The Structure of Scientific Revolutions, 2nd ed. Chicago, IL: University of Chicago Press, 1970.
- Labalme, Patricia H. Beyond Their Sex: Learned Women of the European Past. New York: New York University Press, 1984.

- Landau, Misia. "Paradise Lost: The Theme of Terrestriality in Human Evolution," in *The Rhetoric of the Human Sciences*, eds. John Nelson, Allen Megill, and D. N. McCloskey, Madison, WI: University of Wisconsin Press, 1987: 111–124.
- Lanham, Richard. A Handbook of Rhetorical Terms, 2nd ed. Berkeley, CA: University of California Press, 1991.
- Lanham, Richard. The Electronic Word: Democracy, Technology and the Arts. Chicago, IL: University of Chicago Press, 1993.
- Lanham, Richard. The Economics of Attention: Style and Substance in the Age of Information. Chicago, IL: University of Chicago Press, 2007.
- LaRusso, Dominic. "Root or Branch? A Reexamination of Campbell's 'Rhetoric'," Western Speech 32 (1968): 85–91.
- Latour, Bruno. We Have Never Been Modern. Cambridge MA: Harvard University Press, 1991.
- Leff, Michael. "The Logician's Rhetoric: Boethius' De Differentiis Topicis, Book IV," in In Medieval Eloquence: Studies in the Theory and Practice of Medieval Rhetoric, ed. James J. Murphy, Berkeley, CA: University of California Press, 1978: 3–24.
- Leff, Michael. "The Topics of Argumentative Invention in Latin Rhetorical Theory from Cicero to Boethius," *Rhetorica* 1 (1983): 23–44.
- Leff, Michael. Rethinking Rhetorical Theory, Criticism, and Pedagogy: The Living Art of Michael C. Leff. eds. A. de Velasco, J. A. Campbell, and D. Henry. East Lansing, MI: Michigan State University Press, 2016.
- Lemert, Charles C. and Garth Gillan, *Michel Foucault: Social Theory as Transgression*. New York: Columbia University Press, 1982.
- Levi-Strauss, Claude. "The Structural Study of Myth," *The Journal of American Folklore* 68(270) (1955): 428-444.
- Lincoln, Abraham. "Second Inaugural Address," in *The World's Great Speeches*, ed. Lewis Copeland, New York: Dover, 1958: 316–317.
- Lloyd, G. E. R. Aristotle: The Growth and Structure of His Thought. Cambridge: Cambridge University Press, 1968.
- Lloyd, Keith. "Rethinking Rhetoric from an Indian Perspective: Implications in the 'Nyaya Sutra'," *Rhetoric Review* 26(4) (2007): 365–384.
- Locke, John. Essay on Human Understanding. London: Edward Mory, 1690.
- Lodge, Rupert C. Plato's Theory of Art. 1953. New York: Russell and Russell, 1975.
- Longinus. On the Sublime. Ed. and Trans. G. M. A. Grube. Indianapolis, IN: Library of Liberal Arts, 1957.
- Lu, Xing. Rhetoric in Ancient China, Fifth to Third Century, B.C.E. Columbia, SC: University of South Carolina Press, 1998.
- Lunsford, Andrea A. Reclaiming Rhetorica: Women in the Rhetorical Tradition. Pittsburgh, PA: University of Pittsburgh Press, 1995.
- Lyotard, Jean-Francois. *The Postmodern Condition: A Report on Knowledge*. Trans. Geoff Bennington and Brian Massumi. Minneapolis, MN: University of Minnesota Press, 1988.
- Mack, Peter. Renaissance Argument: Valla and Agricola in the Tradition of Rhetoric and Dialectic. Leiden, Holland: E. J. Brill, 1993.
- Mailloux, Steven. "One Size Doesn't Fit All: The Contingent Universality of Rhetoric," in Sizing up Rhetoric, eds. David Zarefsky and Elizabeth Benacka, Long Grove, IL: Waveland Press, 2008: 7–19.
- Mali, Joseph. The Rehabilitation of Myth: Vico's "New Science." Cambridge: Cambridge University Press, 1992.
- Mann, Nicholas. "The Origins of Humanism," in *The Cambridge Companion to Renaissance Humanism*, ed. Jill Kraye, Cambridge: Cambridge University Press, 1996: 1–19.
- Manson, Richard. The Theory of Knowledge of Giambattista Vico. Hamden, CT: Archon Books, 1969.
- Mao, LuMing and Bo Wang, "Manifesting a Future for Comparative Rhetoric," Rhetoric Review 34(3) (2015): 239-274.
- Margolis, Nadia. "'The Cry of the Chameleon': Evolving Voices in the Epistles of Chris Tine De Pisan," Post and Utz Disputatio 1(37–70) (1996).
- Martin, Bill. Matrix and Line. Albany, NY: State University of New York Press, 1992.
- Martines, Lauro. Power and Imagination: City States in Renaissance Italy. New York: Knopf, 1979.

- May, James M. Trials of Character: The Eloquence of Ciceronian Ethos. Chapel Hill, NC: University of North Carolina Press, 1988.
- McCarthy, Thomas. The Critical Theory of Jürgen Habermas. Cambridge, MA: MIT Press, 1978.
- McCarthy, Thomas, ed. The Philosophical Discourse of Modernity: Twelve Lectures. Trans. Frederick Lawrence. Cambridge, MA: MIT Press, 1987.
- McCloskey, D. N. The Rhetoric of Economics. Madison, WI: University of Wisconsin Press, 1985.
- McCloskey, D. N. "The Neglected Economics of Talk," Planning for Higher Education 22 (1994): 11-16.
- McCloskey, D. N. "The Rhetoric of the Economy and the Polity, whether or Not 'In Crisis'," *Prudentia* (December, 2010) www.deirdremccloskey.com/articles/polity.php
- McCloskey, D. N. Bourgeois Dignity: Why Economics Can't Explain the Modern World. Chicago, IL: University of Chicago Press, 2011.
- McConnell-Ginet, Sally, Ruth Borker, and Nelly Furman, Women and Language in Literature and Society. New York: Praeger, 1980.
- McDermott, Douglas. "George Campbell and the Classical Tradition," Quarterly Journal of Speech 49 (1963): 403-409.
- McHoul, Alec and Wendy Grace, A Foucault Primer: Discourse, Power and the Subject. New York: New York University Press, 1997.
- McKeon, Richard. Rhetoric: Essays in Invention and Discovery, ed. Mark Backman, Woodbridge, CT: Ox Bow Press, 1987.
- McKerrow, Ray E. "Campbell and Whately on the Utility of Syllogistic Logic," Western Speech Communication Journal 40 (1976): 3-13.
- Meador, Prentice A. "Quintilian and the Institutio Oratoria," in A Synoptic History of Classical Rhetoric, ed. James J. Murphy, Davis, CA: Hermagoras Press, 1983: 151-176.
- Megill, Allan. "Recounting the Past: Description, Explanation, and Narrative in Historiography," American Historical Review 94 (1989): 627–653.
- Mendelson, Sarah Heller. The Mental World of Stuart Women: Three Studies. Amherst, MA: The University of Massachusetts Press, 1987.
- Mercier, Hugo and Dan Sperber, "How Do Humans Reason? Arguments for an Argumentative Theory," Behavioral and Brain Sciences 34 (2011): 57-111.
- Midgley, Mary. Evolution as a Religion. London: Routledge, 1985: 30.
- Midgley, Mary. Science as Salvation: A Modern Myth and Its Meaning. London: Routledge, 1992.
- Midgley, Mary. The Myths We Live By. New York: Routledge, 2004.
- Miller, Diane Helene. "The Future of Feminist Rhetorical Criticism," in Listening to Their Voices: The Rhetorical Activities of Historical Women, ed. Molly Meijer Wertheimer, Columbia, SC: University of South Carolina Press, 1997: 359–380.
- Miller, J., M. Prosser, and T. Benson, eds. *Readings in Medieval Rhetoric*. Bloomington, IN: Indiana University Press, 1974.
- Mohrmann, G. P. "The Civile Conversation: Communication in the Renaissance," Speech Monographs 39 (1972): 193-204.
- Moline, Jon. Plato's Theory of Understanding. Madison, WI: University of Wisconsin Press, 1981.
- Mooney, Michael. Renaissance Thought and Its Sources. New York: Columbia University Press, 1979.
- Monfasani, John. *George of Trebizond: A Biography and A Study of His Rhetoric and Logic.* Leiden, Holland: Brill Publishers, 1976.
- Mooney, Michael. Vico in the Tradition of Rhetoric. Princeton, NJ: Princeton University Press, 1985.
- Morris, Meaghan and Paul Patton, eds. *Michel Foucault: Power, Truth, Strategy.* Sydney: Feral Publications, 1979.
- Moscorici, Claudia. Double Dialectics: Between Universalism and Relativism in Enlightened and Postmodern Thought. Lanham, MD: Rowman and Littlefield, 2002.
- Moss, Jean Deitz, ed. Rhetoric and Praxis: The Contribution of Classical Rhetoric to Practical Reasoning. Washington, DC: Catholic University of America Press, 1986.
- Murphy, James J. "Saint Augustine and the Debate about a Christian Rhetoric," *Quarterly Journal of Speech* 46 (1960): 400–410.
- Murphy, James J. Three Medieval Rhetorical Arts. Berkeley, CA: University of California Press, 1971.

Murphy, James J., ed. Medieval Eloquence: Studies in the Theory and Practice of Medieval Rhetoric. Berkeley, CA: University of California Press, 1978.

Murphy, James J. A Synoptic History of Classical Rhetoric. Davis, CA: Hermagoras Press, 1983.

Murphy, James J., ed. A Short History of Writing Instruction. Davis, CA: Hermagoras Press, 1990.

Murphy, James J., ed. *Peter Ramus's Attack on Cicero*. Trans. Carole Newlands. Davis, CA: Hermagoras Press, 1992.

Nash, Christopher and Martin Warner, eds. Narrative in Culture. London: Routledge, 1988.

Nauert, Charles G., Jr. Humanism and the Culture of Renaissance Europe. Cambridge: Cambridge University Press, 1995.

Nelson, John, Allen McGill, and D. N. McCloskey, eds. *The Rhetoric of the Human Sciences*. Madison, WI: University of Wisconsin Press, 1987.

Norris, Christopher. Derrida. Cambridge, MA: Harvard University Press, 1987.

Nothstine, William, Carole Blair, and Gary Copeland. *Critical Questions*. New York: St. Martin's Press, 1994.

Ober, Josiah. Mass and Elite in Democratic Athens: Rhetoric, Ideology, and the Power of the People. Princeton, NJ: Princeton University Press, 1989.

Ochs, Donovan J. "Aristotle's Concept of Formal Topics," Speech Monographs 36 (1969): 419-425.

Ochs, Donovan J. "Cicero's Topica: A Process View of Invention," in Explorations in Rhetoric: Essays in Honor of Douglas Ehninger, ed. Ray E. McKerrow, Glenview, IL: Scott, Foresman, 1982: 178–186.

Ong, Walter J. S. J. Ramus: Method and the Decay of Dialogue. Cambridge, MA: Harvard University Press, 1983.

Ong, Walter J. Orality and Literacy: The Technologizing of the Word, 2nd ed. New York: Routledge, 2002.

Perelman, Chaim. The Idea of Justice and the Problem of Argument. New York: Random House, 1963.

Perelman, Chaim. Justice. New York: Random House, 1967.

Perelman, Chaim. The New Rhetoric and the Humanities. Trans. William Kluback. Dordrecht, Holland: D. Reidel, 1979.

Perry, Linda A., M. Lynn, H. Turner, and Helen M. Sterk, eds. Constructing and Reconstructing Gender: The Links among Communication, Language, and Gender. Albany, NY: State University of New York Press, 1992.

Philostratus. Lives of the Sophists. Trans. W. C. Wright. London: Loeb Classical Library, 1965.

Pico. Oration on the Dignity of Man. Quoted in Anthony Grafton, "Humanism, Magic and Science," in The Impact of Humanism on Western Europe, eds. Anthony Goodman and Angus MacKay. London: Longman, 1990: 99–117.

Plato. Gorgias. Trans. W. C. Helmbold. Indianapolis, IN: Bobbs-Merrill, 1952.

Plato. Phaedrus. Trans. W. C. Helmbold and W. G. Rabinowitz. Indianapolis, IN: Liberal Arts Press, 1956.

Plato. Gorgias. Trans. T. Irwin. Oxford: Clarendon Press, 1979.

Plato. Sophist. Trans. William S. Cobb. Savage, MD: Rowman and Littlefield, 1990.

Plato. Protagoras. Trans. C. C. W. Taylor. Oxford: Clarendon Press, 1991.

Plochmann, George Kimball and Franklin E. Robinson. A Friendly Companion to Plato's Gorgias. Carbondale, IL: Southern Illinois University Press, 1988.

Polanyi, Michael. Personal Knowledge. Chicago, IL: University of Chicago Press, 1962.

Poulakos, John. "Toward a Sophistic Definition of Rhetoric," Philosophy and Rhetoric 16 (1983): 35-48.

Poulakos, John. "Terms for Sophistical Rhetoric," in Rethinking the History of Rhetoric: Multidisciplinary Essays on the History of Rhetoric, ed. Takis Poulakos, Boulder, CO: Westview, 1993: 53-74.

Poulakos, John. Sophistical Rhetoric in Classical Greece. Columbia, SC: University of South Carolina Press, 1995.

Prado, C. G. Starting with Foucault: An Introduction to Genealogy. Boulder, CO: Westview, 1995.

Prelli, Lawrence. A Rhetoric of Science: Inventing Scientific Discourse. Columbia, SC: University of South Carolina Press, 1989.

Prelli, Lawrence, ed. Rhetorics of Display. Columbia, SC: University of South Carolina, 2006.

Putnam, L. L. and M. E. Pacanowsky, eds. *Communication and Organizations: An Interpretive Approach*. Newbury Park, CA: Sage, 1983: 99–122.

Quandahl, Ellen. "Aristotle's Rhetoric: Reinterpreting Invention," Rhetoric Review 4 (1986): 128-137.

Quintilian. *Institutio Oratoria*. Vol. 4. Trans. H. E. Butler. Cambridge, MA: Loeb Classical Library, 1959–1963.

Quintilian. On the Early Education of the Citizen Orator. Ed. James J. Murphy. Trans. John S. Watson. Indianapolis, IN: Library of Liberal Arts, 1965.

Rabinow, Paul, ed. The Foucault Reader. New York: Pantheon Books, 1984.

Racevskis, Karlis. Michel Foucault and the Subversion of Intellect. Ithaca, NY: Cornell University Press, 1983.

Ramus, Peter. Rhetoricae Distinctiones in Quintilianum (Arguments in Rhetoric against Quintilian). Trans. Carole Newlands. 1549. DeKalb, IL: Northern University Press, 1986.

Ramus, Peter. The Questions of Brutus. 1549. Trans. Carole Newlands. Davis, CA: Hermagoras Press, 1992.

Randall, John H. Aristotle. New York: Columbia University Press, 1960.

Rankin, H. D. Sophists, Socratics and Cynics. London: Croom Helm, 1983.

Ray, John W. "Perelman's Universal Audience," Quarterly Journal of Speech 64 (1978): 361-375.

Redfern, Jenny R. "Christine de Pisan and The Treasure of the City of Ladies: A Medieval Rhetorician and Her Rhetoric," in Reclaiming Rhetorica: Women in the Rhetorical Tradition, ed. Andrea A. Lunsford, Pittsburgh, PA: University of Pittsburgh Press, 1995; 73–92.

Renfrew, Alastair. Mikhail Bakhtin. London: Routledge, 2015.

Richardson, Malcom. "Women, Commerce, and Rhetoric in Medieval England," in *Listening to Their Voices: The Rhetorical Activities of Historical Women*, ed. Molly Meijer Wertheimer, Columbia, SC: University of South Carolina Press, 1997: 133–149.

Sallis, J. Being and Logos. Atlantic Highlands, NJ: Humanities Press, 1986.

Schiappa, Edward. "Did Plato Coin Rhetorike?" American Journal of Philology 111 (1990): 457-470.

Schiappa, Edward. Protagoras and Logos. Columbia, SC: University of South Carolina Press, 1991.

Scott, Izora. Controversies over the Imitation of Cicero in the Renaissance. 1910. Davis, CA: Hermagoras Press, 1991.

Scott, Robert L. "On Viewing Rhetoric as Epistemic," Central States Speech Journal 18 (1967): 9–16.

Scott, Robert L. and Wayne Brockriede. *The Rhetoric of Black Power.* New York: Harper and Row, 1969: 132.

Scult, Allen. "Perelman's Universal Audience: One Perspective," Central States Speech Journal 27 (1976): 176–180.

Segal, Charles P. "Gorgias and the Psychology of the Logos," Harvard Studies in Classical Philology 66 (1962): 99-155.

Seigel, Jerrold E. Rhetoric and Philosophy in Renaissance Humanism. Princeton, NJ: Princeton University Press, 1968.

Shackleton-Bailey, D. R. Cicero. New York: Charles Scribner's Sons, 1971.

Sher, Richard B. Church and University in the Scottish Enlightenment. Edinburgh: Edinburgh University Press, 1985.

Sheridan, Thomas. A Course of Lectures on Elocution. London: J. Dodsley, 1762.

Sheridan, Thomas. A General Dictionary of the English Language. London: J. Dodsley, 1780.

Sheridan, Thomas. A Discourse Being Introductory to His Course of Lectures on Elocution and the English Language. 1759. Los Angeles: Augustan Reprint Society, 1969.

Shumway, David R. Michel Foucault. Charlottesville, VA: University Press of Virginia, 1989.

Simons, Herbert W., ed. The Rhetorical Turn. Chicago, IL: University of Chicago Press, 1990.

Simons, Herbert W. Sizing Up Rhetoric, eds. D. Zarefsky and E. Benacka, Long Grove, IL: Waveland Press, 2008.

Sloane, Thomas O. On the Contrary: The Protocol of Traditional Rhetoric. Washington, DC: Catholic University of America Press, 1997.

Spender, Dale. Man Made Language. Boston, MA: Routledge and Kegan Paul, 1980.

Spitzer, Adele. "The Self-Reference of the Gorgias," Philosophy and Rhetoric 8 (1975): 1–22.

Stark, Ryan John. "Margaret Cavendish and Composition Style," *Rhetoric Review* 17(2) (Spring 1999): 264–281.

Steinby, Liisa and Tintti Klapuri, eds. Bakhtin and His Others: Intersubject (ivism), Chronotype, Dialogism. London: Anthem, 2013.

- Stevens, Kevin M. "Books Fit for a Portuguese Queen: The Lost Library of Catherine of Austria and the Milan Connection (1540)," in *Documenting the Early Modern Book World*, eds. Malcolm Walsby and Natasha Constantinidou, Boston, MA: Brill, 2013: 85–116.
- Stringfellow, Frank. The Meaning of Irony: A Psychoanalytic Investigation. Albany, NY: State University of New York Press, 1994.
- Strini, Tom. "A Taut Take on Beethoven's Ninth," The Milwaukee Journal Sentinel May 13 (2006).
- Sullivan, Dale. "Kairos and the Rhetoric of Belief," Quarterly Journal of Speech 78 (1992): 317-332.
- Sutton, Jane. "The Marginalization of Sophistical Rhetoric and the Loss of History," in *Rethinking the History of Rhetoric*, ed. Takis Poulakos, Boulder, CO: Westview Press, 1993a: 75–90.
- Sutton, Jane. "The Taming of *Polos/Polis*: Rhetoric as an Achievement without Woman," *Southern Communication Journal* 57(2) (1993b): 97–119.
- Swearingen, Jan. Rhetoric and Irony: Western Literacy and Western Lies. Oxford: Oxford University Press, 1991.
- Swearingen, Jan. "A Lover's Discourse: Diotima, Logos, and Desire," in *Reclaiming Rhetorica*, ed. Andrea Lunsford, Pittsburgh, PA: University of Pittsburgh Press, 1995: 25–51.
- Tagliacozzo, G. and D. P. Verene, eds. Giambattista Vico's Science of Humanity. Baltimore, MD: Johns Hopkins University Press, 1976.
- Tannen, Deborah. Gender and Discourse. Oxford: Oxford University Press, 1994.
- Thorne, Barrie and Nancy Henley. Language and Sex: Difference and Dominance. Rowley, MA: Newberry House, 1975.
- Tompkins, Jane P., ed. Reader Response Criticism. Baltimore, MD: Johns Hopkins University Press, 1980.
- Trapp, Robert and Janice Schuetz, eds. Perspectives on Argument: Essays in Honor of Wayne Brockriede. Prospect Heights, IL: Waveland Press, 1990.
- Troup, Calvin. Temporality, Eternity, and Wisdom: The Rhetoric of Augustine's Confessions. Columbia, SC: University of South Carolina Press, 1999.
- Untersteiner, Mario. The Sophists. Oxford: Oxford University Press, 1954.
- van Eemeren, Frans H., Rob Grootendorst, and Tjark Kruiger. *Handbook of Argumentation Theory*. Dordrecht, Holland: Foris Publications, 1987.
- Van Maanen, John. Tales of the Field. Chicago, IL: University of Chicago Press, 1988.
- Vasaly, Ann. Representations: Images of the World in Ciceronian Oratory. Berkeley, CA: University of California Press, 1993.
- Vatz, Richard. "The Myth of the Rhetorical Situation," Philosophy and Rhetoric 6 (1973): 154-161.
- Verene, Donald Philip, ed. *Giambattista Vico*, On Humanistic Education: Six Inaugural Orations, 1699–1707. Trans. G. A. Pinton and A. W. Shippe. Ithaca, NY: Cornell University Press, 1993.
- Vickers, Brian, ed. Rhetoric Revalued: Papers from the International Society for the History of Rhetoric. Binghamton, NY: Center for Medieval and Early Renaissance Studies, 1982a.
- Vickers, Brian. "On the Practicalities of Renaissance Rhetoric," in *Rhetoric Revalued*, ed. Brian Vickers, Binghamton, NY: Center for Medieval and Early Renaissance Studies, 1982b: 133–141.
- Vickers, Brian. In Defense of Rhetoric. Oxford: Oxford University Press, 1988.
- Vico, Giambattista. *On Humanistic Education: Six Inaugural Orations*, 1699–1707. Ed. Donald Phillip Verene. Trans. G. A. Pinton and A. W. Shippe. Ithaca, NY: Cornell University Press, 1993.
- Vico, Giambattista. "Oratzione in Morte Donna Angela Cimmino Marchesa Petrella," in *Opere Di G. B. Vico*, vol. 7, ed. Fausto Nicolini, Rome: Bari Laterza, 1911–1914: 170.
- Vitanza, Victor, ed. Writing Histories of Rhetoric. Carbondale, IL: Southern Illinois University Press, 1994.
- Vonnegut, K. S. "Listening for Women's Voices," Communication Education 41 (1992): 26-39.
- Walker, Jeffrey. Rhetoric and Poetics in Antiquity. Oxford: Oxford University Press, 2000.
- Walsh, Lynda. "The Rhetoric of Oracles," Rhetoric Society Quarterly 33 (Summer, 2003): 55-78.
- Walzer, Arthur. "Magic and Rhetoric from Antiquity to the Renaissance: Some Ruminations," *Rhetorica* 6 (1988): 57–118.
- Walzer, Arthur. George Campbell: Rhetoric in the Age of Enlightenment. Albany NY: State University of New York Press, 2003.
- Ward, John O. "From Antiquity to the Renaissance: Glosses and Commentaries on Cicero's Rhetorica," in *Medieval Eloquence*, ed. J. J. Murphy, Berkeley, CA: University of California Press, 1978: 25–67.

- Warnick, Barbara. The Sixth Canon: Belletristic Rhetorical Theory and Its French Antecedents. Columbia, SC: University of South Carolina Press, 1993.
- Waterfield, Robin, ed. The First Philosophers: The Presocratics and the Sophists. Oxford: Oxford University Press, 2009.
- Watson, Burton. Visions of Order. Baton Rouge, LA: Louisiana State University Press, 1964.
- Watson, Burton. Basic Writings of Mo Tzu, Hsun Tzu, and Han Fei Tzu. New York: Columbia University Press, 1967.
- Welch, Kathleen E. The Contemporary Reception of Classical Rhetoric: Appropriations of Ancient Discourse. Hillsdale, NJ: Lawrence Erlbaum, 1990.
- Wells, Susan. Sweet Reason: Rhetoric and the Discourses of Modernity. Chicago, IL: University of Chicago Press, 1996.
- Wenzel, Joseph. "Three Perspectives on Argument," in *Perspectives on Argumentation: Essays in Honor of Wayne Brockriede*, eds. Robert Trapp and Janice Schuetz, Prospect Heights, IL: Waveland, 1990: 9–26.
- Whately, Richard. *Elements of Rhetoric*, ed. Douglas Ehninger. 1828. Carbondale, IL: Southern Illinois University Press, 1963.
- Willard, Charles Arthur. "Argumentation and Postmodern Critique," in *Perspectives on Argumentation:* Essays in Honor of Wayne Brockriede, eds. Robert Trapp and Janice Schuetz, Prospect Heights, IL: Waveland, 1990: 221–231.
- Wilson, Katharina, ed. Women Writers of the Renaissance and Reformation. Athens: University of Georgia Press, 1987.
- Wood, Ellen Meiksins. Peasant-Citizen and Slave: The Foundations of Athenian Democracy. London: Verso, 1988.
- Wood, Julia T. Gendered Lives. Belmont, CA: Wadsworth, 1994.
- Woods, Marjorie Currie. "The Teaching of Writing in Medieval Europe," in A Short History of Writing Instruction, ed. James J. Murphy, Davis, CA: Hermagoras Press, 1990: 77-94.